

University of Tennessee, Knoxville TRACE: Tennessee Research and Creative Exchange

Masters Theses

Graduate School

5-1991

The geometric programming problem

Yancheng Lin

Follow this and additional works at: https://trace.tennessee.edu/utk_gradthes

Recommended Citation

Lin, Yancheng, "The geometric programming problem. " Master's Thesis, University of Tennessee, 1991. https://trace.tennessee.edu/utk_gradthes/12457

This Thesis is brought to you for free and open access by the Graduate School at TRACE: Tennessee Research and Creative Exchange. It has been accepted for inclusion in Masters Theses by an authorized administrator of TRACE: Tennessee Research and Creative Exchange. For more information, please contact trace@utk.edu.

To the Graduate Council:

I am submitting herewith a thesis written by Yancheng Lin entitled "The geometric programming problem." I have examined the final electronic copy of this thesis for form and content and recommend that it be accepted in partial fulfillment of the requirements for the degree of Master of Science, with a major in Mathematics.

Yueh-er Kuo, Major Professor

We have read this thesis and recommend its acceptance:

Ben Fitzpatrick, Tadeusz Janik

Accepted for the Council: Carolyn R. Hodges

Vice Provost and Dean of the Graduate School

(Original signatures are on file with official student records.)

To the Graduate Council:

I am submitting herewith a thesis written by Yancheng Lin entitled "The Geometric Programming Problem". I have examined the final copy of this thesis for form and content and recommend that it be accepted in partial fulfillment of the requirements for the degree of Master of Science, with a major in Mathematics.

Yucher Kub Yucher Kuo, Major Professor

We have read this thesis and recommend its acceptance:

Accepted for the council:

wminkel

Vice Provost and Dean of the Graduate School

STATEMENT OF PERMISSION TO USE

In presenting this thesis in partial fulfillment of the requirements for a Master's degree at the University of Tennessee, Knoxville, I agree that the Library shall make it available to borrowers under rules of the Library. Brief quotations from this thesis are allowable without special permission, provided that accurate acknowledgement of the source is made.

Permission for extensive quotation from or reproduction of this thesis may be granted by my major professor, or in her absence, by the Head of Interlibrary Services when, in the opinion of either, the proposed use of the material is for scholarly purposes. Any copying or use of the material in this thesis for financial gain shall not be allowed without my written permission.

Signature Jancheng Lin

Date April 11, 1991

THE GEOMETRIC PROGRAMMING PROBLEM

A Thesis

Presented for the

Master of Science

Degree

The University of Tennessee, Knoxville

Yancheng Lin

May 1991

ACKNOWLEDGEMENTS

The author wishes to express his sincere gratitute to Dr. Yueher Kuo for her generous guidance and kind help in preparing this thesis.

Much appreciation is also extended to Drs. Ben Fitzpatrick and Tadeusz Janik who provided invaluable assistance in the completion of this paper.

ABSTRACT

The aim of this thesis is to investigate the theory and methods of the geometric programming problem. The stress of the presentation is placed on the methods for solving the problem as well as the proofs of the theorems, which reveal the nature of the geometric programming problem and form the bases of the methods.

The first two chapters discuss the properties of the (posynomial) geometric programming problem and the methods for solving the problem.

Chapter I gives a discussion on the dual method, which solves the geometric programming problem via an indirect approach, and its theoretic basis.

Chapter II presents the primal method, which finds the solutions of the geometric programming problem directly but approximately, and its theoretic basis.

The last two chapters discuss the properties of the generalized geometric programming problem, that is, signomial geometric programming problem, and the methods for solving the problem.

Chapter III presents the general features of the signomial geometric programming problem and the method for solving the problem by way of transforming the problem to complementary geometric programming.

Chapter IV gives a discussion about the properties of reversed geometric programming, which is a special case of signomial geometric programming, and the method for solving the problem. In fact, this method, which can be seen as a supplement for that presented in chapter III, shows another approach to solve the signomial geometric programming problem.

TABLE OF CONTENTS

CHAPTER		PAGE
Ι	THE DUAL METHOD	1
H	THE PRIMAL METHOD	26
III	THE GENERALIZED GEOMETRIC PROGRAMMING PROBLEM.	47
IV	THE REVERSED GEOMETRIC PROGRAMMING PROBLEM	57
BIBLIOGRAPHY		83
APPENDIX		85
VITA		91

CHAPTER I

THE DUAL METHOD

In this chapter, we consider the dual method for solving the (posynomial) geometric programming problem. In order to present the method clearly, we will discuss some of the related properties of the problem first. The basic results of this chapter can be found in [6], [7], [9] and [11].

Geometric programming is known as

minimize
$$g_0(t)$$
,
subject to $t \in F_P$, (1.1)

where

$$\begin{split} F_p &= \{ t \mid g_k(t) \leq 1, \text{ for } k = 1, 2, ..., p \text{ (forced constraint)}, \\ t_j &> 0, \text{ for } j = 1, 2, ..., m \text{ (natural constraint)} \}, \\ g_k(t) &= \sum_{i \in [k]}^{i} u_i(t) = \sum_{i \in [k]} c_i \prod_{i=1}^{m} t_i^{a_{ij}}, \text{ for } c_i > 0, a_{ij} \in R. \end{split}$$

Throughout this paper we use index set $[k] \subset I = \{1, 2, ..., n\}$ for k = 0, 1, ..., p to stand for numbers of the terms of $g_k(\underline{t})$. It is implied here that $g_k(\underline{t}) > 0$, in which case $g_k(\underline{t})$ is called posynomial (positive polynomial).

The associated dual problem of (1.1) can be reached via appropriate rearrangement of Lagrange multiplier (see Appendix). That is,

maximize
$$V(\underline{d}) = \prod_{i \in I} (c_i / d_i)^{d_i} \prod_{k=1}^{p} (\lambda_k)^{\lambda_k}$$
,
subject to $\underline{d} \in F_D$, (1.2)

where

 $F_D = \{ \underline{d} \mid \sum_{i \in I} a_{ij} d_i = 0, \text{ for } j = 1, 2, ..., m \text{ (orthogonality condition),} \}$

$$\begin{split} \lambda_0 &= 1 \quad (\text{normality condition}), \\ d_i &\geq 0, \quad \text{for } i = 1, \, 2, \, ..., \, n \quad (\text{positivity condition}) \; \}, \\ \lambda_k &\geq 0, \quad \text{for } k = 1, \, 2, \, ..., \, p, \\ \lambda_k &= \sum_{i \in [k]} d_i, \quad \text{for } k = 0, \, 1, \, ..., \, p. \end{split}$$

Note the orthogonality condition here indicates that the exponential matrix A = $(a_{ij})_{nxm}$ is orthogonal to the dual vector $\underline{d} = (d_1, d_2, ..., d_n)^T$ (T denotes the transpose).

In general, $g_k(\underline{t})$ is not necessarily a convex function (e.g., $t_j^{a_{ij}}$ is not convex for $a_{ij} \in (0,1)$), so problem (1.1) is not necessarily a convex program.

However, by convexity theory, it is easy to show that the problem that

minimize
$$f_0(\underline{X})$$
,
subject to $\underline{X} \in F_{P1}$, (1.3)

where

$$\begin{split} & \mathsf{F}_{\mathsf{P1}} = \{ \underline{x} \mid \mathsf{f}_{\mathsf{k}}(\underline{x}) \leq 1, \text{ for } \mathsf{k} = 1, 2, ..., p \} \\ & \mathsf{f}_{\mathsf{k}}(\underline{x}) = \mathsf{g}_{\mathsf{k}}(\underline{t}) = \sum_{i \in [k]} \mathsf{c}_{i} \exp \{ \sum_{j=1}^{m} \mathsf{a}_{ij} \, \mathsf{x}_{j} \}, \\ & \mathsf{x}_{j} = \log \mathsf{t}_{j}, \text{ for } j = 1, 2, ..., m \end{split}$$

is a convex program. So, any of its local minimum is its global minimum. In view of the fact that (1.3) is obtained through an one-to-one corresponding transformation we have the following important result.

Theorem 1.1. Any of the local minimum of (1.1) is its global minimum.

Similarly, the fact that

$$\log V(\underline{d}) = \sum_{i=1}^{n} d_{i} (\log c_{i} - \log d_{i}) + \sum_{k=1}^{p} \lambda_{k}(\underline{d}) \log \lambda_{k}(\underline{d})$$

is a concave function (see [6]) and F_D is a convex set leads to:

Theorem 1.2. Any of the local maximum of (1.2) is its global maximum.]

Theorem 1.3. For $u_i > 0$, $d_i \in (0,1)$, i = 1, 2, ..., n and $\sum_{i=1}^n d_i = 1$, it holds

that

$$\prod_{i=1}^{n} (u_i / d_i)^{d_i} \leq \sum_{i=1}^{n} u_i^{}, \qquad (1.4)$$

where "=" holds if and only if $u_i = u_j$, for $i \neq j$, i, j = 1, 2, ..., n.

Proof. Let $y_i = \log x_i$, for i = 1, 2, ..., n. Then

$$d^2y_i / dx_i^2 = (-1) / x_i^2 < 0$$
, for $x_i \in S = (0,\infty)$ (convex set).

By convexity theory, it is clear that y_i is a concave function on S and

$$\sum_{i=1}^{n} d_{i} \log x_{i} \leq \log \left(\sum_{i=1}^{n} d_{i} x_{i} \right),$$

that is,

$$\log \left(\prod_{i=1}^{n} x_{i}^{d_{i}}\right) \leq \log \left(\sum_{i=1}^{n} d_{i} x_{i}\right).$$

where "=" holds if and only if $x_i = x_j$, for $i \neq j$, i, j = 1, 2, ..., n. Thus,

 $\prod_{i=1}^{n} x_{i}^{d_{i}} \leq \sum_{i=1}^{n} d_{i} x_{i}$ (monotony of logarithmic function),

where "=" holds if and only if $x_i = x_j$, for $i \neq j$, i, j = 1, 2, ..., n.

Let
$$x_i = u_i / d_i$$
, for $i = 1, 2, ..., n$. Then

$$\prod_{i=1}^{n} (u_{i} / d_{i})^{d_{i}} \leq \sum_{i=1}^{n} u_{i}^{d_{i}},$$

where "=" holds if and only if $u_i = u_j$, for $i \neq j, i, j = 1, 2, ..., n$.

The inequality shown in (1.4) is called geometric inequality, which is the foundamental inequality of geometric programming. As we will see later, it plays a root role in the derivation of the basic properties of the geometric programming problem.

Corollary 1.1. For $\lambda = \sum_{i=1}^{n} d_i$,

$$\prod_{i=1}^{n} (u_i / d_i)^{d_i} \lambda^{\lambda} \leq \left(\sum_{i=1}^{n} u_i \right)^{\lambda}, \qquad (1.4')$$

and "=" holds if and only if

$$d_j \sum_{i=1}^{n} u_i = u_j \sum_{i=1}^{n} d_i$$
, for i, j = 1, 2, ..., n.

Proof. We know that for $e_i > 0$ and $\sum_{i=1}^{n} e_i = 1$,

$$\begin{split} &\prod_{i=1}^{n} \left(u_{i} / e_{i}\right)^{e_{i}} \leq \sum_{i=1}^{n} u_{i} \, .\\ &\text{Let } e_{i} = d_{i} / \lambda \text{ where } \lambda = \sum_{i=1}^{n} d_{i} \text{ and } \sum_{i=1}^{n} e_{i} = \sum_{i=1}^{n} d_{i} / \lambda = 1. \text{ Then}\\ &\prod_{i=1}^{n} \left(u_{i} \lambda / d_{i}\right)^{d_{i} / \lambda} \leq \sum_{i=1}^{n} u_{i} \, , \end{split}$$

that is,

$$\prod_{i=1}^{n} \left(u_{i}^{\prime} / d_{i}^{\prime} \right)^{d_{i}^{\prime}} \lambda^{\lambda} \leq \left(\sum_{i=1}^{n} u_{i}^{\prime} \right)^{\lambda}.$$

Let $v_i = u_i / e_i$. Then it is equivalent to (1.4') that

$$\prod_{i=1}^{n} (v_i)^{e_i} \leq \sum_{i=1}^{n} e_i v_i$$

where "=" holds if and only if $v_i = v_j$ (Theorem 1.3), i.e.,

$$u_i / e_i = u_j / e_j$$
, for all $i \neq j$, $i, j = 1, 2, ..., n$.

Note that $e_i = d_i / \lambda$, we have

$$(u_i / d_i) \lambda = (u_j / d_j) \lambda, \ u_i = d_i (u_j / d_j), \ \sum_{i=1}^n u_i = \sum_{i=1}^n d_i (u_j / d_j).$$

Hence, "=" holds in (1.4') if and only if

$$d_{j} \sum_{i=1}^{n} u_{i} = u_{j} \sum_{i=1}^{n} d_{i}$$
, for all $i \neq j$, $i, j = 1, 2, ..., n$

Theorem 1.4. (main lemma) If $\underline{t} \in F_P$, $\underline{d} \in F_D$, then

$$V(\underline{d}) \leq g_0(\underline{i}) \prod_{k=1}^{p} [g_k(\underline{i})]^{\lambda_k} \leq g_0(\underline{i}),$$

and $V(\underline{d}) = g_0(\underline{t})$ if and only if

$$\begin{split} & \textbf{d}_{i} = \textbf{u}_{i}(\underline{t}) \ / \ \textbf{g}_{0}(\underline{t}), \quad \text{for } i \in [0], \\ & \textbf{d}_{i} = \lambda_{k} \ \textbf{u}_{i}(\underline{t}), \quad \text{for } i \in [k], \ k = 1, 2, ..., p. \end{split}$$

Then \underline{t} and \underline{d} are the optimum for (1.1) and (1.2), respectively.

Proof. From Corollary 1.1 we know that

$$\begin{split} \prod_{i \in [k]} \left[u_{i}(\underline{t}) / d_{i} \right]^{d_{i}} \lambda_{k}^{\lambda_{k}} &\leq g_{0}(\underline{t}) \left[g_{k}(\underline{t}) \right]^{\lambda_{k}}, \quad \text{for } k = 0, 1, ..., p, \\ \text{where } u_{i}(\underline{t}) &= c_{i} \prod_{j=1}^{m} t_{j}^{a_{ij}} . \text{ That is,} \\ \prod_{i \in I} \left[u_{i}(\underline{t}) / d_{i} \right]^{d_{i}} \prod_{k=1}^{p} (\lambda_{k})^{\lambda_{k}} &= \prod_{i \in I} \left(c_{i} \prod_{j=1}^{m} t_{j}^{a_{ij}} / d_{i} \right)^{d_{i}} \prod_{k=1}^{p} (\lambda_{k})^{\lambda_{k}} \\ &\leq g_{0}(\underline{t}) \prod_{k=1}^{p} \left[g_{k}(\underline{t}) \right]^{\lambda_{k}}. \\ \text{Note that } \sum_{i \in I} a_{ij} d_{i} = 0, \ g_{k}(\underline{t}) \leq 1 \ (\underline{t} \in F_{p}, \ \underline{d} \in F_{D}), \text{ and} \\ &\prod_{i=1}^{n} \left(u_{i}(\underline{t}) / d_{i} \right)^{d_{i}} = \prod_{i=1}^{n} \left(c_{i} \prod_{j=1}^{m} t_{j}^{a_{ij}} / d_{i} \right)^{d_{i}} \end{split}$$

$$\prod_{i=1}^{n} (u_{i}(t) / d_{i})^{d_{i}} = \prod_{i=1}^{n} (c_{i} \prod_{j=1}^{n} t_{j}^{a_{ij}} / d_{i})^{d_{i}}$$

$$= \prod_{i=1}^{n} (c_{i} / d_{i})^{d_{i}} \prod_{j=1}^{m} t_{j}^{y_{j}} \quad (\text{ where } y_{j} = \sum_{i=1}^{n} a_{ij} d_{i} = 0)$$

$$= \prod_{i=1}^{n} (c_{i} / d_{i})^{d_{i}}.$$

Hence,

$$V(\underline{d}) = \prod_{j=1}^{n} (c_j / d_j) \overset{d_i}{\longrightarrow} \prod_{k=1}^{p} (\lambda_k) \overset{\lambda_k}{\longrightarrow} = \prod_{i=1}^{n} [u_i(\underline{t}) / d_i] \overset{d_i}{\longrightarrow} \prod_{k=1}^{p} (\lambda_k) \overset{\lambda_k}{\longrightarrow}$$
$$\leq g_0(\underline{t}) \prod_{k=1}^{p} [g_k(\underline{t})] \overset{\lambda_k}{\longrightarrow} \leq g_0(\underline{t}).$$

(For sufficiency) Note that for $i \in [0]$, $d_i = u_i(\underline{t}) / g_0(\underline{t})$,

$$\prod_{i \in [0]} [u_i(t) / d_i] d_i (\lambda_0)^{\lambda_0} = g_0(t) \qquad (\lambda_0 = \sum_{i \in [0]} d_i = 1)$$

For $i \in [k]$, k = 1, 2, ..., p, $d_i = \lambda_k u_i(t)$,

$$\prod_{i \in [k]} \left[u_i(\underline{t}) / d_i \right]^{d_i} (\lambda_k)^{\lambda_k} = \prod_{i \in [k]} \left[u_i(\underline{t}) / (u_i(\underline{t}) \lambda_k) \right]^{d_i} (\lambda_k)^{\lambda_k} = 1. \left(\lambda_k = \sum_{i \in [k]} d_i \right).$$

Hence,

$$V(\underline{d}) = \prod_{i \in [0]} [u_i(\underline{t}) / d_i] \overset{d_i}{\to} (\lambda_0)^{\lambda_0} \prod_{k=1}^p \prod_{i \in [k]} [u_i(\underline{t}) / d_i] \overset{d_i}{\to} (\lambda_k)^{\lambda_k}$$
$$= g_0(\underline{t}).$$

(For necessity) From the argument above and Corollary 1.1, we know

$$V(\underline{d}) = \prod_{k=0}^{p} \prod_{i \in [k]} [u_i(\underline{t}) / d_i] d_i (\lambda_k)^{\lambda_k} \leq \prod_{k=0}^{p} [g_k(\underline{t})]^{\lambda_k} = g_0(\underline{t}) \prod_{k=1}^{p} [g_k(\underline{t})]^{\lambda_k}$$

and

$$\prod_{i \in [k]} [u_i(t) / d_i]^{d_i} (\lambda_k)^{\lambda_k} \leq [\sum_{i \in [k]} u_i(t)]^{\lambda_k} \quad (\lambda_0 = 1)$$

where "=" holds if and only if

$$d_j \sum_{i \in [k]} u_i(t) = u_j(t) \sum_{i \in [k]} d_i \quad \text{ (by Corollary1.1)}.$$

So, $V(\underline{d}) = g_0(\underline{t})$ implies $g_k(\underline{t}) = 1$ and "=" holds for the inequality above. Note

$$\sum_{i \in [0]} d_i = 1, \quad \sum_{i \in [k]} d_i = \lambda_k, \quad \sum_{i \in [0]} u_i(\underline{t}) = g_0(\underline{t}), \quad \sum_{i \in [k]} u_i(\underline{t}) = g_k(\underline{t}) = 1.$$

Therefore,

$$\begin{split} & \textbf{d}_{i} = \textbf{u}_{i}(\underline{t}) \ / \ \textbf{g}_{0}(\underline{t}), \quad \text{for } i \in [0], \\ & \textbf{d}_{i} = \lambda_{k} \ \textbf{u}_{i}(\underline{t}), \quad \text{for } i \in [k], \ k = 1, 2, ..., p. \end{split}$$

The following results can be immediately reached from Theorem 1.4.

Corollary 1.2.

(i) Suppose $\underline{i}^* \in F_P$, $\underline{d}^* \in F_D$, and $V(\underline{d}^*) = g_0(\underline{i}^*)$. Then \underline{i}^* , \underline{d}^* must be the optimum for problem (1.1) and (1.2), respectively.

(ii) Suppose that F_P and F_D are nonempty. Let

$$\begin{split} \mathbf{M}_{\mathsf{P}} &= \inf \, \mathbf{g}_{\mathsf{0}}(\underline{t}), \quad \text{for all } \underline{t} \in \mathsf{F}_{\mathsf{P}}, \\ \mathbf{M}_{\mathsf{D}} &= \sup \, \mathsf{V}(\underline{d}), \quad \text{for all } \underline{d} \in \mathsf{F}_{\mathsf{D}}. \end{split}$$

Then

$$0 < M_{P} \le M_{D} \le \infty.$$
(iii) Suppose $V(\underline{d}^{*}) = \max V(\underline{d})$, for all $\underline{d} \in F_{D}$. Then $\underline{t} \in F_{P}$ such that
$$\sum_{j=1}^{m} a_{ij} \log t_{j} = \log (d_{i}^{*} V(\underline{d}^{*}) / c_{i}), \text{ for } i \in [0],$$

$$\sum_{j=1}^{m} a_{ij} \log t_{j} = \log (d_{i}^{*} / c_{i} \lambda_{k}), \text{ for } i \in [k], k = 1, 2, ..., p$$

is the optimum solution for (1.1).

Lemma 1.1. (Farkas lemma) For $\underline{\xi} = (\xi_1, \xi_2, ..., \xi_m)^T \ge \underline{0}$ and matrix A,

 $A_{\xi} = b$

if and only if it holds that for any vector \underline{x} , if

then

 $\underline{\mathbf{x}}^{\mathsf{T}}\underline{\mathbf{b}} \ge \mathbf{0}.$

Theorem 1.5. (Kuhn-Tucker theorem) Suppose that $f_k(\underline{x})$ for k = 1, 2, ..., p are convex functions with continuous partial derivatives of first order, and that for the convex program that

minimize $f_0(\underline{x})$,

subject to
$$\underline{x} \in F$$
, (1.5)

where

 $F = \{ \underline{x} \mid f_k(\underline{x}) \le 0, \text{ for } k = 1, 2, ..., p \},\$

there exists x such that

 $f_k(\underline{x}) < 0$, for k = 1, 2, ..., p (Slater condition).

Then $\underline{x}^* \in F$ is the optimum for the program (1.5) if and only if there exists Lagrange multiplier $\underline{\mu}^* = (\mu_1^*, \mu_2^*, ..., \mu_p^*)^T \ge \underline{0}$ such that

$$\nabla L (\underline{x}^*, \underline{\mu}^*) = \underline{0},$$

 $\mu_k^* f_k(\underline{x}^*) = 0, \text{ for } k = 1, 2, ..., p,$

where

$$L(\underline{x}, \underline{\mu}) = f_0(\underline{x}) + \sum_{k=1}^{p} \mu_k f_k(\underline{x})$$

is Lagrangian function.

Proof. (For necessity) Suppose that

 $f_0(\underline{x}^*) = \min f_0(\underline{x}), \text{ for all } \underline{x} \in F.$

For k such that $f_k(\underline{x}^*) < 0$, let $\mu_k^* = 0$. So, we can assume for k = 1, 2, ..., p,

$$f_k(\underline{x}^*) = 0.$$

Let \underline{x} be any point such that

$$(\underline{x} - \underline{x}^*)^{\dagger} \nabla_{\underline{x}} f_k(\underline{x}^*) \le 0$$
, for k = 1, 2, ..., p.

Since (1.5) satisfies Slater condition, that is, $\exists \underline{x}^1$ such that for k = 1, 2, ..., p, $f_k(\underline{x}^1) < 0$.

Note that $f_k(\underline{x})$ are convex functions with continuous partial derivatives of first order, then

$$(\nabla_{\underline{X}} f_{k}(\underline{x}^{*}))^{\mathsf{T}} (\underline{x}^{1} - \underline{x}^{*}) = (\nabla_{\underline{X}} f_{k}(\underline{x}^{*}))^{\mathsf{T}} (\underline{x}^{1} - \underline{x}^{*}) + f_{k}(\underline{x}^{*}) \quad (f_{k}(\underline{x}^{*}) = 0)$$

$$\leq f_{k}(\underline{x}^{1}) < 0. \quad (\text{convexity of } f_{k}(\underline{x})).$$

Let

$$\underline{y} = (1 \cdot \theta) \underline{x} + \theta \underline{x}^1$$
, for $\theta \in (0, 1)$.

Then for k = 1, 2, ..., p,

$$(\nabla_{\underline{X}} f_k(\underline{x}^*))^{\mathsf{T}} (\underline{y} - \underline{x}^*) < 0.$$

Let

$$\underline{z} = \underline{x}^* + \tau (\underline{y} - \underline{x}^*), \text{ for } \tau > 0.$$

Then for sufficiently small τ ,

$$\begin{aligned} (\nabla_{\underline{X}} f_{\underline{k}}(\underline{z}))^{\mathsf{T}} (\underline{y} - \underline{x}^{*}) < 0. \quad (\text{continuity of } \nabla_{\underline{X}} f_{\underline{k}}(\underline{x})). \\ f_{\underline{k}}(\underline{z}) = f_{\underline{k}}(\underline{z}) - f_{\underline{k}}(\underline{x}^{*}) \qquad (f_{\underline{k}}(\underline{x}^{*}) = 0) \\ &\leq (\nabla_{\underline{X}} f_{\underline{k}}(\underline{z}))^{\mathsf{T}} (\underline{y} - \underline{x}^{*}) < 0. \end{aligned}$$

That is, $\underline{z} \in F$.

For the objective function, we have

$$(\nabla_{\underline{X}} f_0(\underline{z}))^{\mathsf{T}} (\underline{y} - \underline{x}^*) \ge f_0(\underline{z}) - f_0(\underline{x}^*) \quad (\text{continuity of } f_0(\underline{x})) \\ \ge 0. \quad (f_0(\underline{x}^*) = \min f_0(\underline{x}), \text{ for all } \underline{x} \in \mathsf{F}).$$

That is, as $\tau \rightarrow 0$,

$$(\nabla_{\underline{X}} f_0(\underline{x}^*))^{\mathsf{T}} ((1-\theta) \underline{x} + \theta \underline{x}^1 - \underline{x}^*) \geq 0,$$

and as $\theta \rightarrow 0$,

$$(\nabla_{\underline{X}} f_0(\underline{x}^*))^{\mathsf{T}} (\underline{x} - \underline{x}^*) \ge 0,$$

that is,

$$(\underline{x} - \underline{x}^*)^{\mathsf{T}} [-\nabla_{\underline{X}} f_0(\underline{x}^*)] \leq 0.$$

That means for any x such that

$$(\underline{x} - \underline{x}^{*})^{\mathsf{T}} [\nabla_{\underline{X}} f_{k}(\underline{x}^{*})] \leq 0, \text{ for } k = 1, 2, ..., p,$$

it holds that

 $(\underline{x} - \underline{x}^*)^T [-\nabla_{\underline{X}} f_0(\underline{x}^*)] \le 0.$ So, by Lemma 1.1, for $\mu_L^* \ge 0$,

$$\sum_{k=1}^{p} \mu_{k}^{*} \nabla_{\underline{X}} f_{k}(\underline{x}^{*}) = - \nabla_{\underline{X}} f_{0}(\underline{x}^{*})$$

That is,

$$\begin{split} \nabla_{\underline{X}} f_{\underline{0}}(\underline{X}^{\star}) &+ \sum_{k=1}^{p} \mu_{k}^{\star} \nabla_{\underline{X}} f_{\underline{k}}(\underline{X}^{\star}) = \underline{0} , \\ \nabla \ L (\underline{X}^{\star}, \mu_{k}^{\star}) &= \underline{0} . \end{split}$$

(For sufficiency) For all $\underline{x} \in F$, we have

$$\mu_k^* f_k(\underline{x}) \le 0$$
, for $k = 1, 2, ..., p$. $(\mu_k^* \ge 0, f_k(\underline{x}) \le 0)$.

Note that in view of the convexity of $f_k(\underline{x})$,

$$f_k(\underline{x}) \ge f_k(\underline{x}^*) + (\underline{x} - \underline{x}^*)^T [\nabla_{\underline{x}} f_k(\underline{x}^*)], \text{ for } k = 0, 1, ..., p.$$

Therefore,

$$\begin{split} f_{0}(\underline{x}) &\geq f_{0}(\underline{x}) + \sum_{k=1}^{p} \mu_{k}^{*} f_{k}(\underline{x}) \quad (\mu_{k}^{*} f_{k}(\underline{x}) \leq 0, \text{ for } k = 1, 2, ..., p). \\ &\geq \left[f_{0}(\underline{x}^{*}) + (\underline{x} - \underline{x}^{*})^{T} (\nabla_{\underline{x}} f_{0}(\underline{x}^{*}))\right] + \sum_{k=1}^{p} \mu_{k}^{*} f_{k}(\underline{x}) \\ &= f_{0}(\underline{x}^{*}) + \left[(\underline{x} - \underline{x}^{*})^{T} (\nabla_{\underline{x}} f_{0}(\underline{x}^{*})) + \sum_{k=1}^{p} \mu_{k}^{*} (\underline{x} - \underline{x}^{*})^{T} \nabla_{\underline{x}} f_{k}(\underline{x}^{*})\right] \\ &- \sum_{k=1}^{p} \mu_{k}^{*} (\underline{x} - \underline{x}^{*})^{T} \nabla_{\underline{x}} f_{k}(\underline{x}^{*}) + \sum_{k=1}^{p} \mu_{k}^{*} f_{k}(\underline{x}) \\ &= f_{0}(\underline{x}^{*}) + (\underline{x} - \underline{x}^{*})^{T} \nabla L (\underline{x}^{*}, \mu_{k}^{*}) + \sum_{k=1}^{p} \mu_{k}^{*} [f_{k}(\underline{x}) - (\underline{x} - \underline{x}^{*})^{T} (\nabla_{\underline{x}} f_{k}(\underline{x}^{*})] \\ &\geq f_{0}(\underline{x}^{*}) + (\underline{x} - \underline{x}^{*})^{T} \nabla L (\underline{x}^{*}, \mu_{k}^{*}) + \sum_{k=1}^{p} \mu_{k}^{*} f_{k}(\underline{x}^{*}). \end{split}$$

Note that it is given that

$$\nabla L (\underline{x}^{\star}, \mu_{k}^{\star}) = \underline{0},$$
$$\sum_{k=1}^{p} \mu_{k}^{\star} f_{k}(\underline{x}^{\star}) = 0.$$

Therefore, $f_0(\underline{x}) \ge f_0(\underline{x}^*)$, for all $\underline{x} \in F$.

Definition 1.1.

(i) A program (primal or dual) is said to be *consistent* if there is at least one vector satisfying its constraints.

(ii) Primal program is said to be *superconsistent* if there is at least one vector <u>t</u> such that

$$g_k(t) < 1$$
, for $t_i > 0$, $j = 1, 2, ..., m$, $k = 1, 2, ..., p$.

Theorem 1.6. (duality theorem) Suppose that (1.1) is superconsistent and has an optimum solution \underline{t}^* . Then there exists $\underline{\mu}^* = (\mu_1^*, \mu_2^*, ..., \mu_p^*)^T \ge$ 0 such that

$$\begin{split} \nabla g_0(\underline{i}^*) + \sum_{k=1}^p (\mu_k^*) \ \nabla g_k(\underline{i}^*) &= \underline{0} \ , \\ \mu_k^* \left[\ g_k(\underline{i}^*) - 1 \ \right] &= 0, \quad \text{for } k = 1, 2, ..., p. \end{split}$$
 And $\underline{d}^* = (\ d_1^{**}, \ d_2^{**}, \ ..., \ d_n^{**})^T$ such that $d_i^* = u_i(\underline{i}^*) \ / \ g_0(\underline{i}^*), \quad \text{for } i \in [0], \end{split}$

 $d_i^* = (\mu_k^*) u_i(\underline{t}^*) / g_0(\underline{t}^*), \text{ for } i \in [k], k = 1, 2, ..., p$ is the optimum solution for (1.2) and satisfies $V(\underline{d}^*) = g_0(\underline{t}^*).$

Proof. Since (1.1) is superconsistent and \underline{t}^* is the optimum for (1.1), (1.3) is convex program satisfying Slater condition and \underline{x}^* is the optimum solution for (1.3). Then by Theorem 1.5, we have

$$\nabla f_0(\underline{x}^*) + \sum_{k=1}^{p} \mu_k^* \nabla f_k(\underline{x}^*) = \underline{0}, \qquad (1,6)$$

$$\mu_{k}^{*} [f_{k}(\underline{x}^{*}) - 1] = 0, \text{ for } k = 1, 2, ..., p.$$
(1.7)

Note that $g_k(\underline{t}^*) = f_k(\underline{x}^*)$, so (1.7) implies that

 $\mu_k^* [g_k(\underline{t}^*) - 1] = 0, \text{ for } k = 1, 2, ..., p.$

Assume that $\mu_0^* = 1$, then (1.6) becomes

$$\begin{split} \sum_{k=0}^{p} (\mu_{k}^{*}) & \nabla f_{k}(\underline{x}^{*}) = \underline{0} .\\ \text{Since } x_{j}^{*} &= \log t_{j}^{*} \text{ and }\\ & \frac{\partial}{\partial x_{j}} f_{k}(\underline{x}) = \sum_{i \in [k]} c_{i} a_{ij} \exp \{ \sum_{j=1}^{m} a_{ij} x_{j} \}, \text{ for } j = 1, 2, ..., m, \end{split}$$

we have

$$\sum_{k=0}^{p} \mu_{k}^{*} \sum_{i \in [k]} c_{i} a_{ij} \prod_{j=1}^{m} (t_{j}^{*})^{a_{ij}} = 0, \text{ for } j = 1, 2, ..., m.$$

Then for $\phi = 1, 2, ..., m, t_{\phi}^* > 0$,

$$\frac{\partial}{\partial t_{\varphi}} \left[g_{0}(\mathbf{i}^{*}) + \sum_{k=1}^{p} (\mu_{k}^{*}) g_{k}(\mathbf{i}^{*}) \right] = (1 / t_{\varphi}^{*}) \left[\sum_{k=0}^{p} \mu_{k}^{*} \sum_{i \in [k]} c_{i} a_{ij} \prod_{j=1}^{m} (t_{j}^{*})^{a_{ij}} \right] = 0.$$

That is,

$$\nabla g_0(\mathbf{i}^*) + \sum_{k=1}^p \mu_k^* \nabla g_k(\mathbf{i}^*) = \mathbf{0}$$

For <u>d</u>* such that

$$\begin{split} & d_i^* = u_i(\underline{t}^*) / g_0(\underline{t}^*), & \text{for } i \in [0], \\ & d_i^* = (\mu_k^*) u_i(\underline{t}^*) / g_0(\underline{t}^*), & \text{for } i \in [k], \ k = 1, 2, ..., p \end{split}$$

it is clear to see that

$$\begin{split} \underline{d}^{*} &\geq 0 \quad (\text{positivity}) \quad (\text{since } \mu_{k}^{*}, \, u_{i}(\underline{t}^{*}) \geq 0), \\ \lambda_{0} &= \sum_{i \in [0]} d_{i}^{*} = 1 \quad (\text{normality}), \\ \sum_{i \in I} a_{ij} d_{i}^{*} &= [1 / g_{0}(\underline{t}^{*})] [\sum_{k=0}^{p} \mu_{k}^{*} \sum_{i \in [k]} c_{i} a_{ij} \prod_{j=1}^{m} (t_{j}^{*})^{a_{ij}}] = 0 \quad (\text{orthogonality}). \end{split}$$

That means $\underline{d}^* \in \mathbf{F}_{\mathbf{D}}$.

Note that

$$\lambda_{k}^{*} = (\mu_{k}^{*}) \left[\sum_{i \in [k]} u_{i}(\underline{t}^{*}) \right] / g_{0}(\underline{t}^{*}) = \mu_{k}^{*} / g_{0}(\underline{t}^{*}) \quad (\mu_{k}^{*} \left[g_{k}(\underline{t}^{*}) - 1 \right] = 0).$$

Note $\mu_k^* = \lambda_k^* g_0(t^*)$, so,

 $d_{i}^{*} = [\lambda_{k}^{*} g_{0}(\underline{t}^{*})] u_{i}(\underline{t}^{*}) / g_{0}(\underline{t}^{*}) = \lambda_{k}^{*} u_{i}(\underline{t}^{*}).$

Therefore, by Theorem 1.4, $V(\underline{d}^*) = g_0(\underline{t}^*)$, and \underline{d}^* is the optimum for (1.2).

From Theorem 1.6 we know that for all $i \in [0]$, $u_i(\underline{1}^*) > 0$ implies $d_i^* > 0$. For $i \in [k]$, k = 1, 2, ..., p, $d_i^* = 0$ if and only if $\mu_k^* = 0$. That $g_k(\underline{1}^*) < 1$ implies $\mu_k^* = 0$ for $1 \le k \le p$ since $(\mu_k^*)[g_k(\underline{1}^*) - 1] = 0$, so it holds that $d_i^* = 0$ for all $i \in [k]$, k = 1, 2, ..., p. That $d_q^* = 0$ for some $q \in [k]$ also implies $\mu_k^* = 0$, so $d_i^* = 0$ for all $i \in [k]$, k = 1, 2, ..., p. As a result, we reach the following important fact.

Corollary 1.3. Suppose that

 $g_0(\underline{i}^*) = \min g_0(\underline{i}), \text{ for all } \underline{i} \in F_P,$ $V(\underline{d}^*) = \max V(\underline{d}), \text{ for all } \underline{d} \in F_D.$

Then for all $i \in [k]$, k = 1, 2, ..., p,

$$d_i^* > 0$$
, if $g_k(t^*) = 1$,
 $d_i^* = 0$, if $g_k(t^*) < 1$.

From Corollary 1.3, we know that for k = 1, 2, ..., p, in the case that $d_i^* = 0$ for some $i \in [k], k = 1, 2, ..., p, g_k(\underline{i})$ must be inactive constraint in (1.1) ($g_k(\underline{i}^*) < 1$) and can be revoked, so can d_i for $i \in [k]$, k = 1, 2, ..., p. Clearly, this simplifies solving (1.2).

Note that the existence of an optimum solution for (1.1) is a necessary condition for Theorem 1.6. The following theorem gives a sufficient condition on which (1.1) has an optimum solution.

Theorem 1.7. If (1.1) is consistent and there exists $d_i > 0$ for some i, then (1.1) has an optimum solution.

Proof. Recall that

$$f_k(\underline{x}) = g_k(\underline{t}) = \sum_{i \in [k]} c_i \exp \{ \sum_{j=1}^m a_{ij} x_j \}, \text{ for } k = 0, 1, ..., p.$$

Let

$$y_i^n = \sum_{j=1}^m a_{ij} x_j^n$$
, for i =1, 2, ..., n.

Suppose rank A = r, where A = $(a_{ij})_{nxm}$, and the first r columns of A are linearly independent. Then $\exists \alpha_t = (\alpha_{1t}, \alpha_{2t}, ..., \alpha_{rt})^T$, for t = r+1, r+2, ..., m, such that

$$a_{it} = \sum_{s=1}^{r} \alpha_{st} a_{is}$$
.

So,

$$y_{i}^{n} = \sum_{j=1}^{m} a_{ij} x_{j}^{n} = \sum_{s=1}^{r} a_{is} x_{s}^{n} + \sum_{t=r+1}^{m} x_{t}^{n} \left(\sum_{s=1}^{r} \alpha_{st} a_{is}\right)$$
$$= \sum_{s=1}^{r} a_{is} x_{s}^{n} + \sum_{s=1}^{r} a_{is} \left(\sum_{t=r+1}^{m} \alpha_{st} x_{t}^{n}\right)$$
$$= \sum_{s=1}^{r} a_{is} \left(x_{s}^{n} + \sum_{t=r+1}^{m} \alpha_{st} x_{t}^{n}\right).$$

Let

$$z_s^n = x_s^n + \sum_{t=r+1}^m \alpha_{st} x_t^n.$$

Then

$$y_i^n = \sum_{s=1}^r a_{is} z_s^n$$
, for $i \in [k], k = 0, 1, ..., p$.

That (1.1) is consistent implies (1.3) is also consistent. So, $\exists \{ \underline{x}^n \}$ where $\underline{x}^n = (x_1^n, x_2^n, ..., x_m^n)^T$, n = 1, 2, ..., such that

$$f_k(\underline{x}^n) \le 1$$
, for $k = 1, 2, ..., p$,

and

$$f_0(\underline{x}^n) \to M_p \text{ as } n \to \infty.$$

In view of the fact that

$$\begin{split} f_{k}(\underline{x}^{n}) &= \sum_{i \in [k]} c_{i} \exp \{ y_{i}^{n} \} \leq 1, \\ \sum_{i=1}^{n} d_{i} y_{i}^{n} &= \sum_{i=1}^{n} d_{i} \left(\sum_{j=1}^{m} a_{ij} x_{j}^{n} \right) \\ &= \sum_{j=1}^{m} x_{j}^{n} \left(\sum_{i=1}^{n} a_{ij} d_{i} \right) = 0 \quad \left(\sum_{i=1}^{n} a_{ij} d_{i} = 0, \exists d_{i} > 0 \text{ for some } i \right), \end{split}$$

we know that y_i^n must have upper and lower bounds. Note that $A_{.s}$ (the s-th column of matrix A), s = 1, 2, ..., r is independent of $A_{.t}$, $s, t = 1, 2, ..., r, s \neq t$, so z_s^n must also be bounded. Hence, $\exists z_s^*$, for s = 1, 2, ..., r, such that $z_s^n \rightarrow z_s^*$ as $n \rightarrow \infty$.

Let

$$x_j^* = z_j^*$$
, for j =1, 2, ..., r,
 $x_j^* = 0$, for j = r+1, r+2, ..., m.
n $\exists y_i^*$ corresponding to z_i^* such

Then $\exists y_i^*$ corresponding to z_i^* such that

$$y_{i}^{*} = \sum_{s=1}^{i} a_{is} z_{s}^{*} = \sum_{j=1}^{in} a_{ij} x_{j}^{*}.$$

Now that

$$\begin{split} f_0(\underline{x}^*) &= \sum_{i \in [0]} c_i \exp \{ y_i^* \} = M_p , \\ f_k(\underline{x}^*) &= \sum_{i \in [k]} c_i \exp \{ y_i^* \} \le 1, \text{ for } k = 1, 2, ..., p, \end{split}$$

thus, $t_j^* = \exp\{x_j^*\}$, for j = 1, 2, ..., m are the optimum for the problem (1.1).

As a matter of fact, Corollary 1.2 (iii) claims if (1.2) has the optimum solution \underline{d}^* , then <u>i</u> that appears in Corollary 1.2 (iii) must be the the optimum solution for (1.1). These provide an indirect way for solving (1.1), that is, the dual method, in which the solution of (1.1) is found via solving (1.2). That is nice, because in general (1.2) is of only linear constraints and the solution for (1.2) is far easier to find than that of (1.1), which is highly nonlinear.

Definition 1.2. $\delta = n - (m + 1)$ is called degree of difficulty for (1.1) and (1.2).

Normality and orthogonality conditions of (1.2) form a system of linear equations including n variables and (m+1) equations. So, if $\delta = 0$ and rank A = m + 1, where A is the coefficient matrix of the system, the system has unique solution \underline{d}^* . If $d_i^* < 0$ for some i $(1 \le i \le n)$, which means d_i^* does not satisfy positivity condition, then $F_D = \phi$, and (1.1) has no constrained minimum solution. If $d_i^* \ge 0$ for i = 1, 2, ..., n, then $\underline{d}^* \in F_D \neq \phi$ is also the optimum solution of (1.2), and so the constrained minimum solution of (1.1), $\underline{1}^*$, can be found by Corollary 1.2 (iii). Conversely, what can be found by Corollary 1.2 (iii) is $\underline{1}^*$. Thus, for $\delta = 0$, there is no difficulty in solving (1.2).

However, for $\delta > 0$, the solution is not so easy to find. The system and (1.2) may have infinite many groups of feasible solutions, from which it is difficult to find the optimum. And generally, the larger δ is, the more difficult it is. That is why δ is called degree of difficulty.

For the case that $\delta > 0$, the general solution to the dual constraints can be written as

$$\underline{d}(\underline{y}) = \underline{\beta}^{(0)} + \sum_{r=1}^{\delta} \underline{\beta}^{(r)} y_r \ge \underline{0} ,$$

or in component form,

$$d_i(\underline{y}) = \beta_i^{(0)} + \sum_{r=1}^{\delta} \beta_i^{(r)} y_r \ge 0$$
, for $i = 1, 2, ..., n_i$

where $\underline{y} = (y_1, y_2, ..., y_{\delta})^T$, and y_r for $r = 1, 2, ..., \delta$ are basic variables. Vectors $\underline{\beta}^{(r)}$ for $r = 0, 1, ..., \delta$ are basic vectors, of which $\underline{\beta}^{(0)}$ is called normality vector, and $\underline{\beta}^{(r)}$ for $r = 1, 2, ..., \delta$ are called nullity vectors. So, in tems of basic variables and vectors the dual function has the form that

$$V(\underline{d}) = V(\underline{y}) = \prod_{i \in I} (c_i / d_i)^{d_i} \prod_{k=1}^{p} (\lambda_k)^{\lambda_k}$$
$$= \prod_{i=1}^{n} c_i^{\beta_i^{(0)}} \prod_{r=1}^{\delta} (\prod_{i=1}^{n} c_i^{\beta_i^{(r)}})^{y_r} \prod_{i=1}^{n} (d_i)^{-d_i} \prod_{k=1}^{p} [\lambda_k(\underline{d})]^{\lambda_k(\underline{d})}$$

Let

$$C_0 = \prod_{i=1}^n c_i \beta_i^{(0)},$$
$$C_r = \prod_{i=1}^n c_i \beta_i^{(r)},$$

where $\mathbf{C}_{\mathbf{r}}$ is called basic constant. Then

$$V(\underline{y}) = C_0 \prod_{r=1}^{\delta} C_r y_r \prod_{i=1}^{n} [d_i(\underline{y})]^{-d_i(\underline{y})} \prod_{k=1}^{p} [\lambda_k(\underline{y})]^{\lambda_k(\underline{y})}$$

or in logarithmic form,

$$\log V(\underline{y}) = \log C_0 + \sum_{r=1}^{\delta} y_r \log C_r - \sum_{i=1}^{n} (d_i(\underline{y}) \log d_i(\underline{y})) + \sum_{k=1}^{p} \lambda_k(\underline{y}) (\log \lambda_k(\underline{y}))$$

Note that $\max V(\underline{y}) = V(\underline{y}^*)$ if and only if $\max \log V(\underline{y}) = \log V(\underline{y}^*)$, so the dual program becomes a concave program:

maximize
$$\log V(\underline{y}) = \log V \left(\underline{\beta}^{(0)} + \sum_{r=1}^{\delta} \underline{\beta}^{(r)} y_r \right),$$

subject to $\underline{y} \in F_{D1},$ (1.8)

where

$$\mathsf{F}_{\mathsf{D}1} = \{ \, \underline{y} \mid \underline{d} = \underline{\beta}^{(0)} + \sum_{r=1}^{\delta} \underline{\beta}^{(r)} \, y_r \geq \underline{0} \, \}.$$

Theorem 1.8. (maximizing equations) Suppose $\underline{d} \in F_D$, $d_i \ge 0$, for i = 1, 2, ..., n. Then $\underline{d}(\underline{y}^*)$ is a maximizing point for (1.2) if and only if

$$C_r = \prod_{i=1}^n d_i (\underline{y}^*)^{\beta_i(r)} \prod_{k=1}^p [\lambda_k(\underline{y}^*)]^{-\lambda_k(r)}$$

where

$$C_r = \prod_{i=1}^n c_i^{\beta_i(r)}, r = 1, 2, ..., \delta.$$

Moreover,

$$V(\underline{y}^{\star}) = C_0 \prod_{i=1}^{n} d_i(\underline{y}^{\star}) d_i^{-\beta_i(0)} \prod_{k=1}^{p} [\lambda_k(\underline{y}^{\star})]^{\lambda_k(0)}$$

Proof. Note the facts that

$$\begin{split} & d_{i}(\underline{y}) = \beta_{i}^{(0)} + \sum_{r=1}^{\delta} \beta_{i}^{(r)} y_{r}, \\ & \frac{\partial}{\partial y_{r}} d_{i}(\underline{y}) = \beta_{i}^{(r)}, \\ & \frac{\partial}{\partial y_{r}} [\log d_{i}(\underline{y})] = \beta_{i}^{(r)} / d_{i}(\underline{y}); \\ & \lambda_{k} = \sum_{i \in [k]} d_{i}(\underline{y}) = \lambda_{k}^{(0)} + \sum_{r=1}^{\delta} \lambda_{k}^{(r)} y_{r}, \text{ where } \lambda_{k}^{(r)} = \sum_{i \in [k]} \beta_{i}^{(r)}, \\ & \frac{\partial}{\partial y_{r}} d_{i}(\underline{y}) = \sum_{i \in [k]} \beta_{i}^{(r)}, \\ & \frac{\partial}{\partial y_{r}} \log \lambda_{k}(\underline{y})] = [\sum_{i \in [k]} \beta_{i}^{(r)}] / \lambda_{k}(\underline{y}). \end{split}$$

So,

$$\frac{\partial}{\partial y_{r}} [\log V(\underline{y})] = \log C_{r} - \frac{\partial}{\partial y_{r}} [\sum_{i=1}^{n} d_{i}(\underline{y}) \log d_{i}(\underline{y})] + \frac{\partial}{\partial y_{r}} [\sum_{k=1}^{p} \lambda_{k}(\underline{y}) \log \lambda_{k}(\underline{y})],$$

where

$$\begin{split} \frac{\partial}{\partial y_{r}} \{ \sum_{i=1}^{n} (d_{i}(\underline{y}) \log [d_{i}(\underline{y})] \} &= \sum_{i=1}^{n} \{ \frac{\partial}{\partial y_{r}} [d_{i}(\underline{y})] \log d_{i}(\underline{y}) + d_{i}(\underline{y}) \frac{\partial}{\partial y_{r}} [\log d_{i}(\underline{y})] \} \\ &= \sum_{i=1}^{n} \beta_{i}^{(r)} + \log [\prod_{i=1}^{n} d_{i}(\underline{y})^{\beta_{i}^{(r)}}], \\ \frac{\partial}{\partial y_{r}} [\sum_{k=1}^{p} \lambda_{k}(\underline{y}) \log \lambda_{k}(\underline{y})] &= \sum_{k=1}^{p} \{ \frac{\partial}{\partial y_{r}} [\lambda_{k}(\underline{y})] \log \lambda_{k}(\underline{y}) + \lambda_{k}(\underline{y}) \frac{\partial}{\partial y_{r}} [\log \lambda_{k}(\underline{y})] \} \\ &= \sum_{i=1}^{n} \beta_{i}^{(r)} + \log [\prod_{k=1}^{p} d_{i}(\underline{y})^{\lambda_{k}^{(r)}}]. \end{split}$$

That is,

$$\frac{\partial}{\partial y_{r}} [\log V(\underline{y})] = \log C_{r} - \log [\prod_{i=1}^{n} d_{i}(\underline{y})^{\beta_{i}^{(r)}} \prod_{i=1}^{n} \lambda_{k}(\underline{y})^{-\lambda_{k}^{(r)}}].$$

Suppose that $d_i(\underline{y}) > 0$ and $\frac{\partial}{\partial y_r} [\log V(\underline{y}^*)] = 0$. Then \underline{y}^* is the stationary point for log $[V(\underline{y})]$, and the maximum solution because of the concavity of (1.7). That is,

$$C_{r} = \prod_{i=1}^{n} d_{i}(\underline{y}^{*})^{\beta_{i}^{(r)}} \prod_{k=1}^{p} [\lambda_{k}(\underline{y}^{*})]^{-\lambda_{k}^{(r)}}$$

where

$$C_r = \prod_{i=1}^{n} c_i^{\beta_i^{(r)}}, \text{ for } r = 1, 2, ..., \delta$$

Furthermore,

$$V(\underline{y}^{*}) = C_{0} \prod_{r=1}^{\delta} \left[\prod_{i=1}^{n} d_{i}(\underline{y}^{*}) \beta_{i}^{(r)} \prod_{k=1}^{p} \left[\lambda_{k}(\underline{y}^{*}) \right]^{-\lambda_{k}^{(r)}} \right]^{y_{r}^{*}}$$
$$\prod_{i=1}^{n} \left[d_{i}(\underline{y}^{*}) \right]^{-d_{i}(y^{*})} \prod_{k=1}^{p} \left[\lambda_{k}(\underline{y}^{*}) \right]^{\lambda_{k}^{(\underline{y}^{*})}}$$

$$= C_0 \prod_{i=1}^{n} [d_i(\underline{y}^*)]^{-d_i(\underline{y}^*) + S_i} \prod_{k=1}^{p} [\lambda_k(\underline{y}^*)]^{\lambda_k(\underline{y}^*) - T_k}$$

where $S_i = \sum_{r=1}^{\delta} \beta_i^{(r)} y_r^{}$, $T_k = \sum_{r=1}^{\delta} \lambda_k^{(r)} y_r^{}$.

Note that

$$\begin{split} d_{i}(\underline{y}^{\star}) &= \beta_{i}^{(0)} + \sum_{r=1}^{\delta} \beta_{i}^{(r)} \, y_{r} = \beta_{i}^{(0)} + S_{i} \, , \\ \lambda_{k}(\underline{y}^{\star}) &= \lambda_{k}^{(0)} + \sum_{r=1}^{\delta} \lambda_{k}^{(r)} \, y_{r} = \lambda_{k}^{(0)} - T_{k} \, . \end{split}$$

So

$$\begin{aligned} - d_i(\underline{y}^*) + S_i &= -\beta_i^{(0)}, \\ \lambda_k(\underline{y}^*) - T_k &= \lambda_k^{(0)}. \end{aligned}$$

Therefore,

$$V(\underline{y}^{*}) = C_{0} \prod_{i=1}^{n} [d_{i}(\underline{y}^{*})^{-\beta_{i}^{(0)}} \prod_{k=1}^{p} [\lambda_{k}(\underline{y}^{*})]^{\lambda_{k}^{(0)}}.$$

In order to find basic vector $\underline{\beta}^{(r)}$, for $r = 0, 1, ..., \delta$, consider the dual constraints that

$$\lambda_0 = \sum_{i \in [0]} d_i = 1$$

and

$$\sum_{i \in I} a_{ij} d_i = 0,$$

where

$$d_i(\underline{y}) = \beta_i^{(0)} + \sum_{r=1}^{\delta} \beta_i^{(r)} y_r^{}.$$

That is,

$$\lambda_{0} = \sum_{i \in [0]} \beta_{i}^{(0)} + \sum_{r=1}^{\delta} \left[\sum_{i \in [0]} \beta_{i}^{(r)} \right] y_{r}$$
$$= \lambda_{0}^{(0)} + \sum_{r=1}^{\delta} \lambda_{0}^{(r)} y_{r} = 1$$

and

$$\sum_{i \in I} a_{ij} \left[\beta_i^{(0)} + \sum_{r=1}^{\delta} \beta_i^{(r)} y_r \right] = 0.$$

So, we need

$$\begin{split} &\sum_{i \in [0]} \beta_i^{(0)} = 1, \\ &\sum_{i \in [0]} \beta_i^{(r)} = 0, \ \text{ for } r = 1, \, 2, \, ..., \, \delta, \end{split}$$

and

$$\sum_{i \in I} a_{ij} \beta_i^{(r)} = 0, \text{ for } r = 1, 2, ..., \delta, j = 1, 2, ..., m.$$

Let P_{kl} denote the elementary matrix got by exchanging row k and row l with $l = (e_{ij}) = 1$ if i = j, or = 0 if $i \neq j$, and P_e denote elementary matrix with proper row transformation with l such that

$$\mathsf{P}_{\mathsf{e}} \ \mathsf{A}^{\mathsf{T}} \mathsf{P}_{\mathsf{k}\mathsf{l}} = \mathsf{A}_1^{\mathsf{T}} = (\mathsf{I} \ \mathsf{A}_2^{\mathsf{T}}).$$

Let

$$\mathsf{B}_1 = \begin{pmatrix} -\mathsf{A}_2^\mathsf{T} \\ \mathsf{I} \end{pmatrix}.$$

Then

$$\mathbf{A_1^T B_1} = (\mathbf{I} \quad \mathbf{A_2^T}) \ \begin{pmatrix} \mathbf{-A_2^T} \\ \mathbf{I} \end{pmatrix} = \mathbf{0}.$$

Let

$$B = P_{kl} B_1$$

Then

$$(P_{e} A^{T} P_{kl}) (P_{kl} B) = 0 (P_{kl}^{2} = I),$$

$$A^T B = 0$$

So, if let

$$\mathsf{B} = (\mathsf{b}_{ii}) = (\underline{\mathsf{B}}_{.1}, \underline{\mathsf{B}}_{.2}, ..., \underline{\mathsf{B}}_{.\delta}),$$

then

$$A^{T}B_{,r} = 0$$
, for r = 1, 2, ..., δ ,

i.e., that $\sum_{i=1}^{n} a_{ij} b_{ir} = 0$ is satisfied. If let

$$\begin{split} & k_{r} = \sum_{i \in [0]} b_{ir}, \quad k_{1} = \sum_{i \in [0]} b_{i1} \quad (\text{for } r = 1), \\ & \underline{\beta}^{(0)} = \underline{B}_{.1} / k_{1}, \quad \text{or } \beta_{i}^{(0)} = b_{i1} / k_{1}, \\ & \underline{\beta}^{(r)} = \underline{B}_{.(r+1)} - k_{r+1} \, \underline{\beta}^{[0]}, \quad \text{or } \beta_{i}^{(r)} = b_{i,(r+1)} - k_{r+1} \, \beta_{i}^{(0)}, \end{split}$$

then $\sum_{i \in [0]} \beta_i^{(0)} = 1$, $\sum_{i \in [0]} \beta_i^{(r)} = 0$, and $\sum_{i=1}^n a_{ij} \beta_i^{(r)} = 0$, for $r = 1, 2, ..., \delta, j = 1, 2, ..., \delta$

m are satisfied.

The following three illustrative examples are given to show how the dual method is used to solve the geometric programming problem.

Example 1.1. Find the solution of the problem that

$$\begin{array}{ll} \text{minimize} & g_0(\underline{t}) = (1 \ / \ 2) \ t_1 \ t_2^2 \ t_3 + 3 \ (t_1)^{-1} \ t_2 \ t_3^2 \ , \\ \text{subject to} & g_1(\underline{t}) = 3 \ t_1 \ (t_3)^{-2} + 2 \ (t_2)^{-2} \ (t_3)^{-1} \ \leq 1 \ , \\ & t_j > 0 \ , \ \ \text{for} \ j = 1 \ , 2 \ , 3 \ . \end{array}$$

Solution. The degree of difficulty of the problem is $\delta = 4 - (3 + 1) = 0$. The dual function is:

$$V(\underline{d}) = (\frac{1/2}{d_1})^{d_1} (\frac{3}{d_2})^{d_2} (\frac{3}{d_3})^{d_3} (\frac{2}{d_4})^{d_4} (d_3 + d_4)^{d_3 + d_4}$$

and the dual constraints are:

$$\begin{aligned} d_1 + d_2 &= 1, \\ d_1 - d_2 + d_3 &= 0, \\ 2d_1 + d_2 &- 2d_4 = 0, \\ d_1 + 2d_2 - 2d_3 - d_4 &= 0. \\ d_3 + d_4 &= \lambda_1. \end{aligned}$$

So, $d_1^* = 1 / 5, \ d_2^* = 4 / 5, \ d_3^* &= 3 / 5, \ d_4^* &= 3 / 5, \ \lambda_1 &= 6 / 5, \text{ and} \\ g_0(t^*) &= V(\underline{d}^*) \\ &= (\frac{1/2}{1/5})^{1/5} (\frac{3}{4/5})^{4/5} (\frac{3}{3/5})^{3/5} (\frac{2}{3/5})^{3/5} (6/5)^{6/5} = 5 (3^{7/5}). \end{aligned}$

By Corollary 1.3 (iii), we know that

$$\begin{split} &u_i(\underline{t}) = d_i^* \ V(\underline{d}^*), \ \text{for } i \in [0], \\ &u_i(\underline{t}) = d_i^* / \lambda_k^*, \ \text{for } i \in [k]. \end{split}$$

That is,

$$\begin{aligned} u_{i}(t) &= (1/2) t_{1} t_{2}^{2} t_{3}^{2} = (1/5) 5 (3^{7/5}) = 3^{7/5} ,\\ u_{i}(t) &= 3 (t_{1})^{-1} t_{2} t_{3}^{2} = (4/5) (3^{7/5}) = 4 (3^{7/5}) ,\\ u_{i}(t) &= 3 t_{1} (t_{3})^{-2} = \frac{3/5}{6/5} = 1/2 ,\\ u_{i}(t) &= 2 (t_{2})^{-2} (t_{3})^{-1} = \frac{3/5}{6/5} = 1/2 . \end{aligned}$$

Thus, the solution for the problem is $t_1^* = (1/2)(3^{7/5}), t_2^* = 2 (3^{-3/5}), t_3^* = 3^{6/5}.$

Example 1.2. Solve the problem that minimize $g_0(t) = 6 (t_1)^{-2} (t_2)^3 (t_3)^{-1} + (1/2) t_1 (t_3)^{1/2} + 4 (t_1)^{-1/2} (t_2)^{-1} (t_3)^{-1/4}$, subject to $g_1(t) = 3 (t_1)^{-1} (t_2)^2 + 2 (t_2)^{1/2} (t_3)^{-1} \le 1$,

$$t_i > 0$$
, for $j = 1, 2, 3$

Solution. $\delta = 5 - (3 + 1) = 1$.

$$V(\underline{d}) = \left(\frac{6}{d_1}\right)^{d_1} \left(\frac{1/2}{d_2}\right)^{d_2} \left(\frac{4}{d_3}\right)^{d_3} \left(\frac{3}{d_4}\right)^{d_4} \left(\frac{2}{d_5}\right)^{d_5} \left(d_4 + d_5\right)^{d_4 + d_5}.$$

$$A^{T} = \left(\begin{array}{cccc} -2 & 1 & -1/2 & -1 & 0\\ 3 & 0 & -1 & 2 & 1/2\\ -1 & 1/2 & -1/4 & 0 & -1 \end{array}\right) \implies \left(\begin{array}{cccc} 1 & 0 & -1/3 & 3/4 & 0\\ 0 & 1 & -7/6 & 1/2 & 0\\ 0 & 0 & 0 & -1/2 & 1 \end{array}\right) = (1 \quad A_2^{T}) = A \cdot B_1 = \left(\begin{array}{cccc} -A_2^{T}\\ I \end{array}\right) = \left(\begin{array}{cccc} -A_2^{T}\\ 1 & 0\\ 1 & 0\\ 0 & 1 \end{array}\right), \qquad B = \left(\begin{array}{cccc} -3/4 & 1/3\\ -1/2 & 7/6\\ 0 & 1\\ 1 & 0\\ 1/2 & 0 \end{array}\right),$$

 $k_1 = -5/4, \quad k_2 = 5/2.$

$$\underline{\beta}^{(0)} = B_{\cdot 1} / k_1 = (-4/5) \begin{pmatrix} -3/4 \\ -1/2 \\ 0 \\ 1 \\ 1/2 \end{pmatrix} = \begin{pmatrix} 3/5 \\ 2/5 \\ 0 \\ -4/5 \\ -2/3 \end{pmatrix},$$

$$\underline{\beta}^{(1)} = B_{\cdot 2} - k_2 \underline{\beta}^{(0)} = \begin{pmatrix} 1/3 \\ 7/6 \\ 1 \\ 0 \\ 0 \end{pmatrix} - (5/2) \begin{pmatrix} 3/5 \\ 2/5 \\ 0 \\ -4/5 \\ -2/3 \end{pmatrix} = \begin{pmatrix} -7/6 \\ 1/6 \\ 1 \\ 2 \\ 1 \end{pmatrix},$$

$$\underline{d}(\mathbf{y}) = \underline{\beta}^{(0)} + \underline{\beta}^{(1)} \mathbf{y} = \begin{pmatrix} 3/5 - (7/6)\mathbf{y} \\ 2/5 - (1/6)\mathbf{y} \\ \mathbf{y} \\ -4/5 + 2\mathbf{y} \\ -2/6 + \mathbf{y} \end{pmatrix}.$$

For $\underline{d} \ge \underline{0}$, we have $y \in [2/5, 18/35]$.

The dual problem becomes:

maximize V(y) =
$$\left(\frac{6}{3/5-7y/6}\right)^{3/5-7y/6} \left(\frac{1/2}{2/5+y/6}\right)^{2/5+y/6} \left(\frac{4}{y}\right)^{y}$$

 $\left(\frac{3}{-4/5+2y}\right)^{-4/5+2y} \left(\frac{2}{-2/5+y}\right)^{-2/5+y} \left(-6/5+3y\right)^{-6/5+3y},$

subject to y ∈ [2/5, 18/35].

Trial. Let

$$y_1 = 2/5 = 0.4, \quad \underline{d}(y_1) = (2/15, 7/15, 2/5, 0, 0)^T,$$

 $y_2 = 17/35 = 0.486, \quad \underline{d}(y_2) = (1/30, 101/210, 17/35, 6/35, 3/35)^T,$
 $y_3 = 1/2 = 0.5, \quad \underline{d}(y_3) = (1/60, 29/60, 1/2, 1/5, 1/10)^T,$
 $y_4 = 18/35 = 0.514, \quad \underline{d}(y_4) = (0, 17/35, 18/35, 8/35, 4/35)^T.$

Then

$$V(y_1) = V (\underline{d}(y_1)) = 4.308,$$

$$V(y_2) = V (\underline{d}(y_2)) = 5.091,$$

$$V(y_3) = V (\underline{d}(y_3)) = 5.123 \text{ (max)},$$

$$V(y_3) = V (\underline{d}(y_3)) = 5.041.$$

Let $t_1 = t_3 = 4$. Then

$$g_{1}(t_{2}) = (3/4) t_{2}^{2} + (1/2) t_{2}^{1/2} < 1 \text{ if } t_{3} \in (0, 0.845],$$

$$g_{0}(t_{2}) = (0.094) t_{2}^{3} + 4 + (1.4142) t_{2}^{-1}.$$

Take $t_2 = 0.5$, 0.6, 0.825, 0.83, 0.84, 0.845. Then $g_0(t_2) = 6.84$, 6.377, 5.767, 5.758, 5.739 (min), 5.829, respectively. Note that by Theorem 1.4 we know for all the feasible solution <u>t</u> and <u>d</u>, it holds that

 $V(\underline{d}) \leq g_0(\underline{t}^*) \leq g_0(\underline{t}).$

Therefore, the trial above gives the upper and lower bounds for $g_0(t^*)$, that is,

 $5.123 \le g_0(t^*) \le 5.739$,

or,

 $g_0(t^*) \in [5.431-0.308, 5.431+0.308].$

Example 1.3. Consider the solution of the problem that

$$\begin{array}{ll} \text{minimize} & g_0(t) = 12 \ t_1 \ t_2 + 5 \ t_1 \ (t_2)^{2/3} \ (t_3)^3 + 8 \ t_1 \ (t_2)^{2/3} \ (t_3)^3 \ (t_4), \\ \text{subject to} & g_1(t) = 75 \ (t_1)^{-1} \ (t_2)^{-1} \ (t_3)^{-1} \ (t_4)^{-1} + 2 \ (t_2)^{-1/3} \ (t_3)^2 \leq 1, \\ & g_2(t) = 67 \ t_2 \ t_3 \ t_4 + 3 \ t_4 \leq 1, \\ & t_i > 0, \quad \text{for } j = 1, \, 2, \, 3, \, 4. \end{array}$$

Solution. $\delta = 7 - (4 + 1) = 2$.

$$A^{\mathsf{T}} = \begin{pmatrix} 1 & 1 & 1 & -1 & 0 & 0 & 0 \\ 1 & 2/3 & 2/3 & -1 & -1/3 & 1 & 0 \\ 0 & 3 & 3 & -1 & 2 & 1 & 0 \\ 0 & 0 & 1 & -1 & 0 & 1 & 1 \end{pmatrix} \implies \begin{pmatrix} 1 & 0 & 0 & 0 & 0 & -7 & 0 \\ 0 & 1 & 0 & 0 & 6 & -1 \\ 0 & 0 & 1 & 0 & 1 & -9 & 1 \\ 0 & 0 & 1 & 1 & -10 & 0 \end{pmatrix},$$
$$B = \begin{pmatrix} 0 & 7 & 0 \\ 0 & -6 & 1 \\ -1 & 9 & -1 \\ -1 & 10 & 0 \\ 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix}.$$

$$\begin{aligned} k_1 &= -1, \ k_2 &= 10, \ k_3 &= 0. \\ & \underline{\beta}^{(0)} &= (0, \ 0, \ 1, \ 1, \ -1, \ 0, \ 0))^T, \\ & \underline{\beta}^{(1)} &= (7, \ -6, \ -1, \ 0, \ 10, \ 1, \ 0)^T, \\ & \underline{\beta}^{(2)} &= (0, \ 1, \ -1, \ 0, \ 0, \ 0, \ 1)^T. \end{aligned}$$

$$\begin{aligned} & \underline{d}(\underline{y}) &= \begin{pmatrix} 7y_1 \\ -6y_1 + y_2 \\ 1 - y_1 - y_2 \\ 1 \\ -1 + 10y_1 \\ y_2 \end{pmatrix} \geq \underline{0}. \end{aligned}$$

The dual program becomes:

maximize
$$V(\underline{y}) = (\frac{12}{7y_1})^{7y_1} (\frac{5}{-6y_1+y_2})^{-6y_1+y_2} (\frac{8}{1-y_1-y_2})^{-y_1-y_2} (75)$$

 $(\frac{2}{-1+10y_1})^{-1+10y_1} (\frac{67}{y_1})^{y_1} (3/y_2)^{y_2} (10y_1)^{10y_1} (y_1+y_2)^{y_1+y_2}$

subject to $y_1 \in [1/10, (1/6) y_2]$,

$$(y_1 + y_2) \le 1.$$

CHAPTER II

THE PRIMAL METHOD

:5

This chapter devoted to the primal method for solving the posynomial geometric programming problem. The presentation concentrates on condensation (an effective way of transforming problem (1.1) to a monomial geometric program) and "cutting-plane" algorithm. The main content can be found in [1] and [3].

Consider the inequality that for $e_i(\underline{t}) \ge 0$, $\sum_{i \in [k]} e_i(\underline{t}) = 1$, $[k] \in I = \{1, 2, ..., n\}$,

$$\prod_{i \in [k]} (u_i(\underline{t}) / e_i(\underline{t})) \stackrel{e_i(\underline{t})}{=} \leq \sum_{i \in [k]} u_i(\underline{t}),$$

where

$$u_i(t) = c_i \prod_{j=1}^m t_j^{a_{ij}}$$
, for $t_j > 0, j = 1, 2, ..., m$.

Note that the left side of the inequality is monomial. Furthermore, we have

$$\begin{split} \prod_{i \in [k]} \left[u_{i}(t) / e_{i}(t) \right] e_{i}(t) &= \prod_{i \in [k]} \left[(C_{i} / e_{i}(t)) \prod_{j=1}^{m} t_{j}^{a_{ij}} \right] e_{i}(t) \\ &= \prod_{i \in [k]} \left[C_{i} / e_{i}(t) \right] e_{i}(t) \prod_{j=1}^{m} \prod_{i \in [k]} t_{j}^{a_{ij}} e_{i}(t) \\ &= c_{k}(t) \prod_{j=1}^{m} t_{j}^{a_{kj}}(t), \end{split}$$

where

$$\begin{aligned} c_k(t) &= \prod_{i \in [k]} [c_i / e_i(t)] e_i(t), \\ a_{kj}(t) &= \sum_{i \in [k]} a_{ij} e_i(t). \end{aligned}$$

Let

$$\begin{split} g_k(\underline{t}) &= \sum_{i \in [k]} u_i(\underline{t}), \\ g_k(\underline{t}, \underline{e}(\underline{t})) &= \prod_{i \in [k]} [u_i(\underline{t}) / e_i(\underline{t})] \stackrel{e_i(\underline{t})}{=} c_k(\underline{t}) \prod_{j=1}^m t_j^a k_j(\underline{t}), \text{ for } k = 0, 1, ..., p. \end{split}$$

Then we get a monomial geometric program related to (1.1). That is,

minimize
$$g_0(\underline{t}, \underline{e}(\underline{t}))$$
,
subject to $\underline{t} \in F_{Pe}$, (2.1)

where

$$F_{Pe} = \{ 1 \mid g_k(1, \underline{e}(1)) \le 1, \text{ for } k = 1, 2, ..., p, \\ t_j > 0, \text{ for } j = 1, 2, ..., m \},$$

which is called the condensed program of (1.1).

Let

$$\begin{aligned} x_{j} &= \log t_{j}, & \text{for } j = 1, 2, ..., m, \\ C_{k}(\underline{t}) &= \log c_{k}(\underline{t}) = \sum_{i \in [k]} e_{i}(\underline{t}) \ [\log c_{i} - \log e_{i}(\underline{t})], & \text{for } k = 0, 1, ..., p, \\ G_{k}(\underline{x}, \underline{e}(\underline{t})) &= \log g_{k}(\underline{t}, \underline{e}(\underline{t})) = C_{k}(\underline{t}) + \sum_{j=1}^{m} a_{kj}(\underline{t}) x_{j}, & \text{for } k = 0, 1, ..., p. \end{aligned}$$

Then the transformed linear program of (2.1) can be written as:

minimize
$$G_0(\underline{x}, \underline{e}(\underline{t}))$$
,
subject to $\underline{x} \in F_{PL}$, (2.2)

where

$$F_{PL} = \{ X \mid G_k(X, \underline{e}(\underline{t})) \le 0, \text{ for } k = 1, 2, ..., p \}.$$

The associated dual program of (2.1) is:

maximize
$$V(\underline{\lambda}) = \prod_{k=0}^{n} (c_k(\underline{t}))^{\lambda_k}$$
,
subject to $\underline{\lambda} \in F_{De}$, (2.3)

where

$$F_{De} = \{ \underline{\lambda} \mid \sum_{k=1}^{p} a_{kj}(\underline{t}) \lambda_{k} = 0, \lambda_{0} = 1, \lambda_{k} \ge 0, \text{ for } k = 1, 2, ..., p \},\$$

$$c_{k}(\underline{t}) = \prod_{i \in [k]} [c_{i} / e_{i}(\underline{t})] \overset{\Theta_{i}(\underline{t})}{=} \sum_{i \in [k]} a_{ij} \Theta_{i}(\underline{t}).$$

The transformed linear program of (2.3) can be obtained by taking logarithmic transformation of the objective function. That is,

maximize
$$\log V(\underline{\lambda}) = \sum_{k=0}^{l} \lambda_k \log c_k(\underline{1}),$$

subject to $\lambda_k \in F_{DL},$ (2.4)

where

$$\begin{aligned} \mathsf{F}_{\mathsf{DL}} &= \{ \underline{\lambda} \mid \sum_{k=1}^{n} a_{kj}(\underline{t}) \ \lambda_k = 0, \ \lambda_0 = 1, \ \lambda_k \ge 0, \ \text{ for } k = 1, 2, \ ..., p \}, \\ c_k(\underline{t}) &= \prod_{i \in [k]} [c_i / e_i(\underline{t})]^{\Theta_i(\underline{t})}, \\ a_{kj}(\underline{t}) &= \sum_{i \in [k]} a_{ij} \Theta_i(\underline{t}). \end{aligned}$$

Let

$$\begin{split} & \mathsf{F}_{\mathsf{P}} = \{ \, \underline{t} \mid \underline{g}_{k}(\underline{t}) \leq 1, \quad \text{for } \underline{t}_{j} > 0, \ \underline{j} = 1, \, 2, \, ..., \, m \, \}, \\ & \mathsf{F}_{\mathsf{P}_{\mathsf{P}}} = \{ \, \underline{t} \mid \underline{g}_{k}^{-}(\underline{t}, \, \underline{e}(\underline{t})) \leq 1, \quad \text{for } \underline{t}_{j} > 0, \ \underline{j} = 1, \, 2, \, ..., \, m \, \}, \\ & \mathsf{M}_{\mathsf{P}} = \inf \, \underline{g}_{0}^{-}(\underline{t}), \quad \text{for all } \underline{t} \in \mathsf{F}_{\mathsf{P}} \, , \\ & \mathsf{M}_{\mathsf{P}_{\mathsf{P}}} = \inf \, \underline{g}_{0}^{-}(\underline{t}, \, \underline{e}(\underline{t})) \, , \quad \text{for all } \underline{t} \in \mathsf{F}_{\mathsf{P}_{\mathsf{P}}} \, . \end{split}$$

Then we have the following theorems that reveal some useful properties for using (2.1) to find the solution of (1.1).

Theorem 2.1. $F_P \subset F_{Pe}$.

Proof. By geometric inequality, for all $\underline{t} \ge \underline{0}$,

$$g_k(\underline{t}, \underline{e}(\underline{t})) \leq g_k(\underline{t}), \text{ for } k = 1, 2, ..., p,$$

that is,

$$\prod_{i \in [k]} [u_i(\underline{t}) / \Theta_i(\underline{t})] \stackrel{\Theta_i(\underline{t})}{=} \sum_{i \in [k]} u_i(\underline{t}).$$

Clearly,

 $g_k(\underline{t}, \underline{e}(\underline{t})) \le 1$, if $g_k(\underline{t}) \le 1$,

and the converse is not necessarily true. That is, $\underline{1} \in F_{Pe}$ for all $\underline{1} \in F_P$ and it is possible that $\exists \underline{1}^1 \in F_{Pe}$ such that $\underline{1}^1 \notin F_P$. Hence, $F_P \subset F_{Pe}$.

Theorem 2.2. $M_{Pe} \leq M_{P}$.

proof. Suppose that $M_p < M_{pe}$, and there exists $\underline{i}^* \in F_P$ such that $g_0(\underline{i}^*) = \inf g_0(\underline{i}) = M_P$ for all $\underline{i} \in F_P$. Then $g_0(\underline{i}^*) < \inf g_0(\underline{i}, \underline{e}(\underline{i})) = M_{Pe}$ for all $\underline{i} \in F_{Pe}$. Note that $\underline{i}^* \in F_{Pe}$. Since $F_P \subset F_{Pe}$, we get

 $g_0(\underline{t}^*) < g_0(\underline{t}, \underline{e}(\underline{t})), \text{ for } \forall \underline{t} \in F_{Pe}.$

That is,

$$\sum_{i \in [0]} u_i(\underline{t}^*) < \prod_{i \in [0]} [u_i(\underline{t}) / e_i(\underline{t})] \stackrel{e_i(\underline{t})}{=}, \text{ for } \forall \underline{t} \in F_{\mathsf{Pe}}.$$

That contradicts Theorem 1.3.

Theorem 2.3. For $e_i(\underline{t}') = u_i(\underline{t}') / g_k(\underline{t}')$, $i \in k, k = 1, 2, ..., p$, it holds that

,

 $g_k(\underline{t}, \underline{e}(\underline{t}')) = g_k(\underline{t}'), \text{ for } \underline{t} = \underline{t}'.$

Proof. Now that

$$\begin{split} \mathbf{e}_{i}(\mathbf{t}') &= \mathbf{u}_{i}(\mathbf{t}') / \mathbf{g}_{k}(\mathbf{t}'), \\ \sum_{i \in [k]} \mathbf{e}_{i}(\mathbf{t}') &= \sum_{i \in [k]} \mathbf{u}_{i}(\mathbf{t}') / \mathbf{g}_{k}(\mathbf{t}') = 1 \end{split}$$

SO,

$$g_{k}(\mathbf{i}', \underline{\mathbf{e}}(\mathbf{i}')) = g_{k}(\mathbf{i}, \underline{\mathbf{e}}(\mathbf{i}'))|_{\mathbf{1}=\mathbf{i}'}$$

$$= \prod_{i \in [k]} [u_{i}(\mathbf{i}) / e_{i}(\mathbf{i}')]^{\Theta_{i}(\mathbf{i}')}|_{\mathbf{1}=\mathbf{i}'}$$

$$= \prod_{i \in [k]} [u_{i}(\mathbf{i}) g_{k}(\mathbf{i}') / u_{i}(\mathbf{i}')]^{\Theta(\mathbf{i}')}|_{\mathbf{1}=\mathbf{i}'} \qquad (\Theta_{i}(\mathbf{i}') = u_{i}(\mathbf{i}') / g_{k}(\mathbf{i}'))$$

$$= \prod_{i \in [k]} [u_{i}(\mathbf{i}') g_{k}(\mathbf{i}') / u_{i}(\mathbf{i}')]^{\Theta(\mathbf{i}')}$$

1

=
$$[g_k(\underline{i}')]^E$$
 (where $E = \sum_{i \in [k]} e_i(\underline{i}) = 1$)
= $g_k(\underline{i}')$

Lemma 2.1. Suppose that

$$e_{i}(\mathbf{t}') = u_{i}(\mathbf{t}') / g_{k}(\mathbf{t}') = u_{i}(\mathbf{t}') / \sum_{i \in [k]} u_{i}(\mathbf{t}'), \text{ for } i \in [k],$$

$$g_{k}(\mathbf{t}, \underline{e}(\mathbf{t}')) = \prod_{i \in [k]} [u_{i}(\mathbf{t}) / e_{i}(\mathbf{t}')] e_{i}(\mathbf{t}'),$$

$$f(\mathbf{t}, \underline{e}(\mathbf{t}')) = \prod_{i \in [k]} [u_{i}(\mathbf{t}) / e_{i}(\mathbf{t}')] e_{i}(\mathbf{t}'),$$

then for $j = 1, 2, ..., m, t = (t_1, t_2, ..., t_m)^{T}$,

$$\frac{\partial}{\partial t_{i}}[g_{k}(t, \underline{e}(t'))] = \sum_{i \in [k]} \{ [e_{i}(t') / u_{i}(t)] [g_{k}(t, \underline{e}(t'))] \frac{\partial u_{i}(t)}{\partial t_{j}} \}.$$

Proof. For $[k] = \{1\}$,

$$\begin{aligned} \frac{\partial}{\partial t_{j}} \left[g_{k} \left(\mathbf{t}, \mathbf{e}(\mathbf{t}') \right) \right] &= \frac{\partial}{\partial t_{j}} \left[u_{i}(\mathbf{t}) / \mathbf{e}_{i}(\mathbf{t}') \right]^{\mathbf{e}_{i}}(\mathbf{t}') \\ &= \left[\mathbf{e}_{i}(\mathbf{t}') \right]^{-\mathbf{e}_{i}}(\mathbf{t}') \frac{\partial}{\partial t_{j}} \left[u_{i}(\mathbf{t}) \right]^{\mathbf{e}_{i}}(\mathbf{t}') \\ &= \left[\mathbf{e}_{i}(\mathbf{t}') \right]^{1-\mathbf{e}_{i}}(\mathbf{t}') \left[u_{i}(\mathbf{t}) \right]^{\mathbf{e}_{i}}(\mathbf{t}') - 1 \frac{\partial u_{i}(\mathbf{t})}{\partial t_{j}} \\ &= \left[\mathbf{e}_{i}(\mathbf{t}') / u_{i}(\mathbf{t}) \right] \left[u_{i}(\mathbf{t}) / \mathbf{e}_{i}(\mathbf{t}') \right]^{\mathbf{e}_{i}}(\mathbf{t}') \frac{\partial u_{i}(\mathbf{t})}{\partial t_{j}}. \end{aligned}$$

For [k] = {1, 2},

$$\begin{aligned} \frac{\partial}{\partial t_{j}} \left[g_{k} \left(\mathbf{t}, \mathbf{e}(\mathbf{t}') \right) \right] &= \frac{\partial}{\partial t_{j}} \left[\prod_{i=1}^{2} \left(u_{i}(\mathbf{t}) / \mathbf{e}_{i}(\mathbf{t}') \right)^{\mathbf{e}_{i}}(\mathbf{t}') \right] \\ &= \left[u_{2}(\mathbf{t}) / \mathbf{e}_{2}(\mathbf{t}') \right]^{\mathbf{e}_{2}} \left[\left(u_{1}(\mathbf{t}) / \mathbf{e}_{1}(\mathbf{t}') \right)^{\mathbf{e}_{1}}(\mathbf{t}') \right] \\ &+ \left[u_{1}(\mathbf{t}) / \mathbf{e}_{1}(\mathbf{t}') \right]^{\mathbf{e}_{1}} \left[\left(u_{2}(\mathbf{t}) / \mathbf{e}_{2}(\mathbf{t}') \right)^{\mathbf{e}_{2}}(\mathbf{t}') \right] \end{aligned}$$

|

$$= \sum_{i=1}^{2} \{ [e_{i}(t') / u_{i}(t)] [u_{i}(t) / e_{i}(t')] e_{i}(t') \frac{\partial u_{i}(t)}{\partial t_{j}} \}$$

$$[u_{\xi}(t) / e_{\xi}(t')] e_{\xi}(t') \quad (\text{for } \xi \in [k], \xi \neq i)$$

$$= \sum_{i=1}^{2} \{ [e_{i}(t') / u_{i}(t)] [\prod_{i=1}^{2} (u_{i}(t) / e_{i}(t')) e_{i}(t')] \frac{\partial u_{i}(t)}{\partial t_{j}} \}.$$

Assume that for $[k] = \{1, 2, ..., n - 1\}$, it is true that

$$\frac{\partial}{\partial t_j} \left[g_k(t, \underline{e}(t')) \right] = \sum_{i=1}^{n-1} \left\{ \left[e_i(t') / u_i(t) \right] \left[\prod_{i=1}^{n-1} (u_i(t) / e_i(t')) e_i(t') \right] \frac{\partial u_i(t)}{\partial t_j} \right\}$$

Then for $[k] = \{1, 2, ..., n-1, n\},\$

$$\begin{split} \frac{\partial}{\partial t_{j}} \left[g_{k}\left(t, \underline{e}(t')\right) \right] &= \frac{\partial}{\partial t_{j}} \left[\prod_{i=1}^{n} \left(u_{i}(t) / e_{i}(t')\right)^{e_{i}(t')} \right] \\ &= \left(u_{n}(t) / e_{n}(t')\right)^{e_{n}(t')} \frac{\partial}{\partial t_{j}} \left[\prod_{i=1}^{n-1} \left(u_{i}(t) / e_{i}(t')\right)^{e_{i}(t')} \right] \\ &+ \prod_{i=1}^{n-1} \left(u_{i}(t) / e_{i}(t')\right)^{e_{i}(t')} \frac{\partial}{\partial t_{j}} \left[\left(u_{n}(t) / e_{n}(t')\right)^{e_{n}(t')} \right]^{e_{n}(t')} \right] \\ &= \sum_{i=1}^{n-1} \left\{ \left[e_{i}(t') / u_{i}(t) \right] \left[\prod_{i=1}^{n} \left(u_{i}(t) / e_{i}(t')\right)^{e_{i}(t')} \right] \frac{\partial u_{i}(t)}{\partial t_{j}} \right\} \\ &+ \left\{ \left[e_{n}(t') / u_{n}(t) \right] \left[\prod_{i=1}^{n} \left(u_{i}(t) / e_{i}(t')\right)^{e_{i}(t')} \right] \frac{\partial u_{i}(t)}{\partial t_{j}} \right\} \\ &= \sum_{i=1}^{n} \left\{ \left[e_{i}(t') / u_{i}(t) \right] \left[\prod_{i=1}^{n} \left(u_{i}(t) / e_{i}(t')\right)^{e_{i}(t')} \right] \frac{\partial u_{i}(t)}{\partial t_{j}} \right\}. \end{split}$$

Therefore, we obtain that for j = 1, 2, ..., m,

$$\frac{\partial}{\partial t_{j}} [g_{k}(\underline{t}, \underline{e}(\underline{t}'))] = \sum_{i \in [k]} \{ [e_{i}(\underline{t}') / u_{i}(\underline{t})] g_{k}(\underline{t}, \underline{e}(\underline{t}')) \frac{\partial u_{i}(\underline{t})}{\partial t_{j}} \}.$$

Theorem 2.4. $\nabla_{\underline{i}} [g_k(\underline{i}', \underline{e}(\underline{i}'))] = \nabla_{\underline{i}} g_k(\underline{i}').$

Proof. Note that $e_i(\underline{i}) = u_i(\underline{i}) / g_k(\underline{i})$, for $i \in [k]$,

$$\begin{bmatrix} \mathbf{e}_{i}(\mathbf{t}') / \mathbf{u}_{i}(\mathbf{t}) \end{bmatrix} \begin{bmatrix} \prod_{i \in [k]} (\mathbf{u}_{i}(\mathbf{t}) / \mathbf{e}_{i}(\mathbf{t}')) \\ \mathbf{e}_{i}(\mathbf{t}') \end{bmatrix} \begin{bmatrix} \frac{\partial \mathbf{u}_{i}(\mathbf{t})}{\partial \mathbf{t}_{j}} \end{bmatrix}_{\mathbf{t} = \mathbf{t}'} =$$

$$= \begin{bmatrix} \mathbf{e}_{i}(\mathbf{t}') / \mathbf{u}_{i}(\mathbf{t}') \end{bmatrix} \begin{bmatrix} \mathbf{g}_{k}(\mathbf{t}') \end{bmatrix} \begin{bmatrix} \mathbf{e}_{i}(\mathbf{t}') \\ \frac{\partial \mathbf{u}_{i}(\mathbf{t}')}{\partial \mathbf{t}_{j}} \qquad \text{(where } \mathbf{E}' = \sum_{i \in [k]} \mathbf{e}_{i}(\mathbf{t}') = 1 \text{)}$$

$$= \frac{\partial \mathbf{u}_{i}(\mathbf{t}')}{\partial \mathbf{t}_{j}} \qquad (\mathbf{e}_{i}(\mathbf{t}') / \mathbf{u}_{i}(\mathbf{t}') = 1 / \mathbf{g}_{k}(\mathbf{t}')).$$

That is, by Lemma 2.1, for all j = 1, 2, ..., m,

$$\frac{\partial}{\partial t_j} [g_k(\mathbf{t}', \underline{\mathbf{e}}(\mathbf{t}'))] = \sum_{i \in [k]} [\frac{\partial}{\partial t_j} u_i(\mathbf{t}')] = \frac{\partial}{\partial t_j} [\sum_{i \in [k]} u_i(\mathbf{t}')]$$
$$= \frac{\partial}{\partial t_j} [g_k(\mathbf{t}')].$$

Therefore, it holds that

 $\nabla_{\mathbf{i}} \left[g_k(\mathbf{i}', \underline{\mathbf{e}}(\mathbf{i}')) \right] = \nabla_{\mathbf{i}} g_k(\mathbf{i}').$

The next theorem gives the relation between the optimum solution of (2.1) and that of (1.2).

Theorem 2.5. Suppose F_{Pe} is nonempty, then $M_{Pe} > 0$ if and only if there exists $\underline{d}^* \in F_D$ such that for $e_i \in (0,1)$,

$$\begin{split} e_i &= d_i^* / \sum_{i \in [k]} d_i^*, \\ M_{\mathsf{Pe}} &= \mathsf{V}(\underline{d}^*). \\ \textit{Proof.} \quad (\textit{For necessity}) \quad \textit{Suppose that } \mathsf{F}_{\mathsf{Pe}} \neq \phi, \textit{ and } \mathsf{M}_{\mathsf{Pe}} > 0, \textit{ then} \\ M_{\mathsf{PL}} &= \inf G_o(\underline{x}, \underline{e}(\underline{t})) \quad (\textit{for all } \underline{x} \in \mathsf{F}_{\mathsf{PL}}) \\ &= \inf \log g_0(\underline{t}) \quad (\textit{for all } \underline{t} \in \mathsf{F}_{\mathsf{Pe}}) \\ &= \log \inf g_0(\underline{t}) \quad (\textit{for all } \underline{t} \in \mathsf{F}_{\mathsf{Pe}}) \end{split}$$

$$= \log M_{Pe.}$$
$$M_{DL} = \sup \log V(\underline{\lambda}) \quad (\text{for all } \underline{\lambda} \in F_{DL})$$
$$= \log M_{De}.$$

By the duality theorem for linear programming, we know $M_{DL} = M_{PL}$, So,

$$M_{De} = \sup V(\underline{\lambda}) = M_{Pe}$$
 (for all $\underline{\lambda} \in F_{De}$)

Since

$$V (\underline{\lambda}^{\star}) = \sup V(\underline{\lambda}) \quad (\text{for all } \underline{\lambda} \in \mathsf{F}_{\mathsf{De}}) \\ = \prod_{\mathsf{k}=0}^{\mathsf{p}} (c_k(\underline{t}))^{\lambda_{\mathsf{k}}^{\star}},$$

we have

$$\begin{split} \mathsf{M}_{\mathsf{P}\mathbf{e}} &= \prod_{\mathsf{k}=0}^{\mathsf{p}} \left(c_{\mathsf{k}} \left(\underline{\mathbf{t}} \right) \right) \, \lambda_{\mathsf{k}}^{\star} \\ &= \prod_{\mathsf{k}=0}^{\mathsf{p}} \prod_{\mathsf{i} \in [\mathsf{k}]} \left(c_{\mathsf{i}} / e_{\mathsf{i}} \right) \, e_{\mathsf{i}} \lambda_{\mathsf{k}}^{\star}, \end{split}$$

where $\lambda_{k}^{\star}\,$ is implied to be consistent optimum solution for (2.3) and (2.4).

Define
$$\underline{d}^*$$
 as
 $d_i^* = e_i \lambda_k^*$, for $i \in [k]$, $k = 0, 1, ..., p$,
 $\sum_{i \in [k]} d_i^* = \sum_{i \in [k]} e_i \lambda_k^* = \lambda_k^*$ ($\sum_{i \in [k]} e_i = 1$ for $k = 0, 1, ..., p$).

That is,

$$d_i^* = e_i \sum_{i \in [k]} d_i^*$$
, i.e., $e_i = d_i^* / \sum_{i \in [k]} d_i^*$.

Hence,

$$M_{Pe} = \prod_{k=0}^{p} \prod_{i \in [k]} (c_i / e_i)^{d_i^*}$$
$$= \prod_{k=0}^{p} [\prod_{i \in [k]} (c_i \sum_{i \in [k]} d_i^* / d_i^*)^{d_i^*}$$
$$= \prod_{k=0}^{p} \prod_{i \in [k]} (c_i / d_i^*)^{d_i^*} \prod_{k=0}^{p} [\prod_{i \in [k]} (\sum_{i \in [k]} d_i^*)^{d_i^*}]$$

$$= \prod_{k=0}^{p} \prod_{i \in [k]} (c_i / d_i^*)^{d_i^*} \prod_{k=0}^{p} (\lambda_k^*)^{\lambda_k^*}$$
$$= V(\underline{d}^*) .$$

Since $e_i \ge 0$, $\lambda_k^* \ge 0$, $\lambda_0^* = 1$, we have

$$\begin{aligned} d_{i}^{*} &\geq 0, \quad \sum_{i \in [0]} d_{i}^{*} = 1, \\ &\sum_{i=1}^{n} a_{ij} d_{i}^{*} = \sum_{k=1}^{p} \left(\sum_{i \in [k]} a_{ij} e_{i} \right) \lambda_{k}^{*} \quad (d_{i}^{*} = e_{i} \lambda_{k}^{*}) \\ &= \sum_{k=1}^{p} a_{kj}(\underline{i}) \lambda_{k}^{*} = 0, \text{ for } \underline{j} = 1, 2, ..., m. \end{aligned}$$

Hence, $\underline{d}^* \in F_D$.

(For sufficiency) Suppose that $\underline{d}^* \in F_D$, $d_i^* = e_i \sum_{i \in [k]} d_i^*$, for $i \in [k]$, k = 0, 1, ..., p, and $M_{Pe} = V(\underline{d}^*)$. Let $\lambda_k^* = \sum_{i \in [k]} d_i^*$, for k = 0, 1, ..., p.

Then
$$\lambda_0^* = 1$$
, $\lambda_k^* \ge 0$, $\sum_{i=1}^n a_{ij}(t) d_i^* = 0$. So,

$$\sum_{k=0}^{p} a_{kj}(\underline{i}) \lambda_{k}^{*} = \sum_{k=0}^{p} \left(\sum_{i \in [k]} a_{ij} e_{i} \right) \lambda_{k}^{*}$$
$$= \sum_{k=0}^{p} \left(\sum_{i \in [k]} a_{ij} d_{i}^{*} \right) \quad (d_{i}^{*} = e_{i} \lambda_{k}^{*} \text{ for } i \in [k])$$
$$= 0.$$

Thus, $\lambda_k^* \in F_{De}$. Since F_{Pe} is nonempty, (2.1) and (2.3) are consistent. Therefore, by Corollary 1.2 (ii), $M_{Pe} > 0$.

The theorem below provides the relation between (1.1) and (2.1), which forms the main basis for the primal algorithm that will be presented later in this chapter.

Theorem 2.6. Suppose that (1.1) is superconsistent and that

$$\begin{split} g_{0}(\underline{i}^{*}) &= \min g_{0}(\underline{i}), \text{ for all } \underline{i} \in F_{P}, \\ e_{i}(\underline{i}^{*}) &= u_{i}(\underline{i}^{*}) / g_{k}(\underline{i}^{*}), \text{ for } i \in [k], k = 0, 1, ..., p. \end{split}$$

Then it holds that for all $\underline{t} \in F_{Pe}$,

$$g_o(t^*, \underline{e}(t^*)) = \min g_o(t, \underline{e}(t^*)),$$

$$M_{Pe} = M_P$$
.

Proof. By theorem 2.2, we know that $M_{Pe} \le M_{P}$. Now suppose $M_{Pe} < M_{P}$, then $\exists t_1^1 \in F_{Pe}$ such that

$$\begin{split} g_k(\underline{\mathfrak{l}}^1, \underline{e}(\underline{\mathfrak{l}})) &\leq 1, \quad \text{for } k = 1, 2, ..., p, \\ g_0(\underline{\mathfrak{l}}^1, \underline{e}(\underline{\mathfrak{l}})) &< \mathsf{M}_{\mathsf{P}}. \end{split}$$

Assume that $\exists f^2 \in F_P \subset F_{P_P}$, such that

$$g_k(t^2) < 1$$
, for k = 1, 2, ..., p.

Then

$$g_k(\underline{t}^2, \underline{e}(\underline{t})) \leq g_k(\underline{t}^2) < 1.$$
 ("="holds $\Leftrightarrow \underline{t} = \underline{t}^2$)

Let

$$\underline{\hat{t}} = (\underline{t}^1)^{1-\theta} (\underline{t}^2)^{\theta}, \text{ for } \theta \in (0,1),$$

i.e.

$$\hat{t}_{j} = (t_{j}^{1})^{1-\theta} (t_{j}^{2})^{\theta}$$
, for $j = 1, 2, ..., m, \theta \in (0,1)$.

Then for k = 1, 2, ..., p,

$$g_{k}(\hat{t}, \underline{e}(t)) = c_{k}(t) \prod_{j=1}^{m} \hat{t}_{j}^{a_{kj}}(t)$$

$$= c_{k}(t) \prod_{j=1}^{m} [(t_{j}^{1})^{1-\theta} (t_{j}^{2})^{\theta}]^{a_{kj}}(t)$$

$$= [(c_{k}(t))^{1-\theta} \prod_{j=1}^{m} (t_{j}^{1})^{(1-\theta)} a_{kj}(t)] [(c_{k}(t))^{\theta} \prod_{j=1}^{m} (t_{j}^{2})^{\theta}]^{a_{kj}}(t)$$

$$= [g_{k}(t^{1}, \underline{e}(t))]^{1-\theta} [g_{k}(t^{2}, \underline{e}(t))]^{\theta}.$$

Note that $g_k(\underline{t}, \underline{e}(\underline{t}))$ are monomials, and $g_k(\underline{t}^1, \underline{e}(\underline{t})) \le 1$, $g_k(\underline{t}^2, \underline{e}(\underline{t})) < 1$, so,

 $g_k(\hat{\underline{t}}, \underline{e}(\underline{t})) < 1$, for k = 1, 2, ..., p, and note that $g_o(\underline{t}^1, \underline{e}(\underline{t})) < M_P$, so,

$$g_{0}(\underline{\hat{t}}, \underline{e}(\underline{t})) < M_{P}, \text{ as } \theta \rightarrow 0 \quad (\text{i.e. as } \underline{\hat{t}} \in N(\underline{t}^{1}, \theta)).$$

For $\xi \in (0, 1)$, let
$$\underline{t} = (\underline{\hat{t}})^{\xi} (\underline{t}^{*})^{1-\xi}.$$

Then as $\xi \to 0$ (i.e., $t \to t^*$, $t \in N$ (t^* , ξ), ξ is sufficiently small))

$$\begin{split} g_{k}(t) &= \sum_{i \in [k]} u_{i}(t) = \sum_{i \in [k]} [c_{i} \prod_{j=1}^{m} t_{j}^{a_{ij}}] \\ &= \sum_{i \in [k]} \{c_{i} \prod_{j=1}^{m} [(\hat{t}_{j})^{\xi} (t_{j}^{*})^{1-\xi}]^{a_{ij}}\} \\ &= \sum_{i \in [k]} \{[c_{i} \prod_{j=1}^{m} (t_{j}^{*})^{a_{ij}}] [\prod_{j=1}^{m} (\hat{t}_{j}/t_{j}^{*})^{a_{ij}}]^{\xi}\} \\ &= \sum_{i \in [k]} \{u_{i}(t^{*}) [\prod_{j=1}^{m} (\hat{t}_{j}/t_{j}^{*})^{a_{ij}}]^{\xi}\}. \end{split}$$

So, as $\xi \to 0$ (i.e., $t \to t^*$, $t \in N$ (t^* , ξ), ξ is sufficiently small))

$$\begin{aligned} \frac{d}{d\xi} \left[g_{k}(t) \right] |_{\xi=0} &= \frac{d}{d\xi} \left[\sum_{i \in [k]} u_{i}(t^{*}) \prod_{j=1}^{m} (\hat{t}_{j} / t_{j}^{*})^{a_{ij}} \xi \right] |_{\xi=0} \\ &= \sum_{i \in [k]} \left\{ u_{i}(t^{*}) \frac{d}{d\xi} \left[\prod_{j=1}^{m} (\hat{t}_{j} / t_{j}^{*})^{a_{ij}} \right] \xi \right\} |_{\xi=0} \\ &= \sum_{i \in [k]} \left\{ u_{i}(t^{*}) \prod_{j=1}^{m} (\hat{t}_{j} / t_{j}^{*})^{a_{ij}} \xi |_{\xi=0} \log \left[\prod_{j=1}^{m} (\hat{t}_{j} / t_{j}^{*})^{a_{ij}} \right] \right\} \\ &= \sum_{i \in [k]} \left\{ e_{i}(t^{*}) \left[\sum_{i \in [k]} u_{i}(t^{*}) \right] \sum_{j=1}^{m} a_{ij} \log (\hat{t}_{j} / t_{j}^{*}) \right\} \end{aligned}$$

$$(u_{i}(\mathbf{1}^{*}) = e_{i}(\mathbf{1}^{*}) \sum_{i \in [k]} u_{i}(\mathbf{1}^{*}) = e_{i}(\mathbf{1}^{*}) g_{k}(\mathbf{1}^{*}))$$

$$= g_{k}(\mathbf{1}^{*}) \sum_{i \in [k]} [e_{i}(\mathbf{1}^{*}) \sum_{j=1}^{m} a_{ij} \log (\hat{t}_{j} / t_{j}^{*})]$$

$$= g_{k}(\mathbf{1}^{*}) \sum_{j=1}^{m} \{\sum_{i \in [k]} [e_{i}(\mathbf{1}^{*}) a_{ij}] \log (\hat{t}_{j} / t_{j}^{*})\}$$

$$= g_{k}(\mathbf{1}^{*}) \sum_{j=1}^{m} a_{kj}(\mathbf{1}^{*}) \log (\hat{t}_{j} / t_{j}^{*}) (a_{kj}(\mathbf{1}^{*}) = e_{i}(\mathbf{1}^{*}) a_{ij}).$$

Note that

$$g_{k}(\underline{\mathbf{t}}, \underline{\mathbf{e}}(\underline{\mathbf{t}}^{*})) = c_{k}(\underline{\mathbf{t}}^{*}) \sum_{j=1}^{m} t_{j}^{a_{kj}}(\underline{\mathbf{t}}^{*}),$$
$$\log g_{k}(\underline{\mathbf{t}}, \underline{\mathbf{e}}(\underline{\mathbf{t}}^{*})) = \log c_{k}(\underline{\mathbf{t}}^{*}) + \log \left[\sum_{j=1}^{m} t_{j}^{a_{kj}}(\underline{\mathbf{t}}^{*})\right].$$

Subsequently,

$$\begin{aligned} \frac{d}{d\xi} \left[g_{k}(t) \right] |_{\xi=0} &= g_{k}(t^{*}) \left[\sum_{j=1}^{m} \log (\hat{t}_{j})^{a_{kj}} (t^{*}) - \sum_{j=1}^{m} \log (t_{j}^{*})^{a_{kj}} (t^{*}) \right] \\ &= g_{k}(t^{*}) \left[\log \left(\prod_{j=1}^{m} (\hat{t}_{j})^{a_{kj}} (t^{*}) \right) - \log \left(\prod_{j=1}^{m} (t_{j}^{*})^{a_{kj}} (t^{*}) \right) \right] \\ &= g_{k}(t^{*}) \left\{ \left[\log c_{k}(t^{*}) + \log \left(\prod_{j=1}^{m} (\hat{t}_{j})^{a_{kj}} (t^{*}) \right) \right] \right. \\ &- \left[\log c_{k}(t^{*}) + \log \left(\prod_{j=1}^{m} (t_{j}^{*})^{a_{kj}} (t^{*}) \right) \right] \right\} \\ &= g_{k}(t^{*}) \left[\log g_{k}(\hat{t}, \underline{e}(t^{*})) - \log g_{k}(t^{*}, \underline{e}(t^{*})) \right] \end{aligned}$$

By Theorem 2.3, we know that

37

$$g_k(\hat{\underline{t}}, \underline{e}(\underline{i}^*)) < g_k(\underline{i}^*) = g_k(\underline{i}^*, \underline{e}(\underline{i}^*)), \text{ for } \hat{\underline{t}} \neq \underline{i}^*,$$

then we obtain

$$\frac{d}{d\xi} [g_k(t)]|_{\xi=0} < 0, \text{ for } k = 0, 1, ..., p.$$

That implies that for k = 1, 2, ..., p,

 $g_k(t) < g_k(t^*) \le 1$, for $t \in F_P$ and $t \in N(t^*, \xi)$,

and for k = 0,

$$g_0(t) < g_0(t^*) = M_P$$
.

This contradicts the assumption that

 $g_0(\underline{1}^*) = \min g_0(\underline{1}), \text{ for all } \underline{1} \in F_P.$ Hence, $M_{Pe} = M_P.$

Based on the properties stated above we know that the optimum for (1.1) must be the optimum for (2.1) and $F_P \subset F_{Pe}$. Therefore, we can find such an optimum point sequence $\{\mathbf{1}^{(n)}\}$ that $\{\mathbf{1}^{(n)}\} \in F_{Pe}$ for (2.1), and $\{\mathbf{1}^{(n)}\} \in F_P$ as n $\rightarrow \infty$, that is, $\{\mathbf{1}^{(n)}\}$ remains within a region condensed progressively from F_{Pe} to F_P . Thus, this primal method can be seen as a sort of exterior point method. Before stating the algorithm, let us consider transforming problem (1.1) to a program that is of the form facilitating operation.

Theorem 2.7. The problem that

minimize t_0 , subject to $t \in F_{P1}$, (2.5)

where

$$F_{P1} = \{ \underline{i} \mid t_0^{-1} g_0(\underline{i}) \le 1, \\ g_k(\underline{i}) \le 1, \text{ for } k = 1, 2, ..., p, \}$$

I

$$t_0^{}$$
, $t_i^{} > 0$, for $j = 1, 2, ..., m$ }

is equivalent to (1.1).

Proof. Let $(t_0^*, \underline{1}^*)$ be the minimum of (2.5). Then for $\forall \underline{1} \in F_{P1}$,

 $(t_0^*)^{-1} g_0(t^*) \le 1,$

where

 $t_0^* = \min(t_0) \le t_0^*$.

Now suppose \underline{t}^* is not the minimum of (1.1), then we can find a point $\underline{\hat{t}} \in F_P$ such that $g_0(\underline{\hat{t}}) < g_0(\underline{t}^*) \le t_0^*$. Let

 $\hat{t}_0 = g_0(\hat{t})$ (the upper bound of $g_0(t)$).

Then

 $t_0^* > \hat{t}_0$.

That contradicts the assumption that $t_0^* \le t_0$. Thus, t^* must be the minimum of (1.1).

Similarly, let \underline{t}^* be the minimum of (1.1), that is, $\underline{g}_0(\underline{t}^*) \leq \underline{g}_0(\underline{t})$ for $\forall \underline{t} \in F_{P_1}$

And let

$$\begin{split} t_0^* &= g_0(\underline{t}^*) \leq g_0(\underline{t}), \\ t_0^* &\geq g_0(\underline{t}) \geq t_0^*. \quad (\text{upper bound of } g_0(\underline{t})). \end{split}$$

Suppose that (t_0, \underline{i}^*) is not the minimum of (2.5). Then $\exists (\hat{t}_0, \underline{\hat{t}}), \hat{t}_0 \ge g_0(\underline{\hat{t}})$ such that

$$\hat{t}_0 < t_0^* = g_0(t^*).$$

That is,

 $\boldsymbol{g}_0\left(\underline{\hat{t}}\right) \leq \boldsymbol{\hat{t}}_0 < \boldsymbol{t}_0^* = \boldsymbol{g}_0(\underline{\boldsymbol{i}}^*)$

This contradicts the assumption that $g_0(t^*) \le g_0(t)$.

Since the range of the optimum solution can usually be estimated in practice, it is reasonable to set lower and upper bounds for \underline{t} in advance. So, we have the program that

minimize
$$t_0^{-}$$
,
subject to $\underline{t} \in F_b^{-}$, (2.6)

where

$$\begin{split} F_b &= \{ t \mid g_k(t) \leq 1, \ \text{ for } k = 0, \, 1, \, ..., \, p, \\ &\quad 0 < t_j^{\, l} \leq \, t_j \, \leq \, t_j^{\, u}, \ \text{ for } j = 1, \, 2, \, ..., \, m \, \}. \end{split}$$

And (2.1) can be rewritten as the program that for the operation vector t^1 ,

minimize
$$t_0$$
,
subject to $\underline{t} \in F_{b1}$, (2.7)

where

$$F_{b1} = \{ \underline{t} \mid g_k(\underline{t}, \underline{e}(\underline{t}^1)) \le 1, \text{ for } k = 0, 1, ..., p, \\ 0 < t_j^1 \le t_j \le t_j^u, \text{ for } j = 0, 1, ..., m \}, \\ g_k(\underline{t}, \underline{e}(\underline{t}^1)) = c_k(\underline{t}^1) \prod_{i=0}^m t_j^a A_{kj}(\underline{t}^1).$$

The logarithmic transformation of (2.7) leads to the problem that

minimize log t_o,

subject to
$$\underline{t} \in F_{bl}$$
, (2.8)

where

$$F_{bl} = \{ \underline{1} \mid \log g_k(\underline{t}, \underline{e}(\underline{t}^1)) \le 0, \text{ for } k = 0, 1, ..., p, \\ \log t_j^1 \le \log t_j \le \log t_j^u, \text{ for } j = 0, 1, ..., m \}, \\ \log g_k(\underline{t}, \underline{e}(\underline{t}^1)) = \log c_k(\underline{t}^1) + \sum_{j=0}^m a_{kj}(\underline{t}^1) \log t_j.$$

In order to use simplex method for which positive variables are needed,

define

$$\begin{split} x_{j} &= \log t_{j} - \log t_{j}^{I} \geq 0, \\ x_{j}^{u} &= \log t_{j}^{u} - \log t_{j}^{I} \\ &= \log (t_{j}^{u} / t_{j}^{I}), \text{ for } j = 0, 1, ..., m. \end{split}$$

Then we obtain such an approximating linear program that

minimize
$$x_0 + \log t_0^{-1}$$
,
subject to $\underline{x} \in F_{LP}$, (2.9')

where

$$\begin{aligned} \mathsf{F}_{\mathsf{LP}} &= \{ \underbrace{\mathbf{x}} \mid \log g_k(\underbrace{\mathbf{x}}, \underbrace{\mathbf{e}}(\underline{\mathbf{t}}^1)) \leq 0, \ 0 \leq \mathbf{x}_j \leq \mathbf{x}_j^{\mathsf{u}}, \ \text{for } \mathbf{j} = 0, 1, ..., m \}, \\ \log g_k(\underbrace{\mathbf{x}}, \underbrace{\mathbf{e}}(\underline{\mathbf{t}}^1)) &= \log c_k(\underline{\mathbf{t}}^1) + \sum_{j=0}^{\mathsf{m}} a_{kj}(\underline{\mathbf{t}}^1) \log t_j^{\mathsf{l}} + \sum_{j=0}^{\mathsf{m}} a_{kj}(\underline{\mathbf{t}}^1) \mathbf{x}_j \\ &= \log g_k(\underline{\mathbf{t}}^{\mathsf{l}}, \underline{\mathbf{e}}(\underline{\mathbf{t}}^1)) + \sum_{j=0}^{\mathsf{m}} a_{kj}(\underline{\mathbf{t}}^1) \mathbf{x}_j. \end{aligned}$$

That is,

minimize x_0 ,

subject to
$$\underline{X} \in F_{L^{p}}$$
, (2.9)

where

$$F_{LP} = \{ \underline{x} \mid \sum_{j=0}^{m} a_{kj} \ (\underline{t}^{1}) \ \underline{x}_{j} \le -\log g_{k} \ (\underline{t}^{i}, \underline{e}(\underline{t}^{1})), \text{ for } k = 0, 1, ..., p, \\ 0 \le x_{j} \le x_{j}^{u}, \text{ for } j = 0, 1, ..., m \}.$$

Now let us consider the following "cutting-plane" algorithm.

Step 1. Arbitrarily take a starting operation point $\underline{t}^0 > \underline{0}$, and let $F^0 = \{ \underline{t} \mid g_k(\underline{t}, \underline{e}(\underline{t}^0)) \le 1, \text{ for } k = 1, 2, ..., p, \underline{t}^1 \le \underline{t} \le \underline{t}^u \}.$ Then $F_P \subset F^0$. Form program (2.9) for $\underline{t}^1 = \underline{t}^0$. Set n = 1. Step 2. Solve (2.9) for $\underline{t}^1 = \underline{t}^{n-1}$. Find the solution \underline{x}^n and compute \underline{t}^n . Note that \underline{t}^n is the optimum for (2.9) and not necessarily feasible for (2.6). If (2.9) has no solution, then there is no solution for (2.6).

Step 3. For program (2.6), let

 $g_{k(a)}(\underline{t}^{n}) = \max g_{k}(\underline{t}^{n}) \le 1, \text{ for all } 1 \le k \le p.$ (2.10)

Then t^n is the optimum for (2.6), and stop. Otherwise, define

$$g_{k(n)}(t^n) = \max \{ g_k(t^n) | g_k(t^n) > 1 \}, \text{ for all } 1 \le k \le p,$$

and turn to step 4.

Step 4. Condense $g_{k(n)}(\underline{t})$ at \underline{t}^n (new operation point) to obtain a linear constraint for (2.9) (new one added to the current (2.9)) that

 $\log g_k(\underline{x}, \underline{e}(\underline{t}^n)) \le 0,$

and define

 $C^{n} = \{ \underline{t} \mid \log g_{k}(\underline{x}, \underline{e}(\underline{t}^{n})) \leq 0 \text{ (known as cuts)} \},\$ $F^{n} = F^{n-1} \cap C^{n}.$

Let n increase by 1. Turn to step 2.

At each iteration Step 4 produces a new constraint adding to (2.9) and the feasible region of (2.9) reduces, in other words, part of which is cut off. It is worth noting that since $g_k(\underline{t}, \underline{e}(\underline{t}^n)) \leq g_k(\underline{t})$, $g_k(\underline{t}) \leq 1$ implies that $g_k(\underline{t}, \underline{e}(\underline{t}^n)) \leq 1$ for $\forall \underline{t} \in F_b$, that is, $\underline{t} \in C^n$, and such $\underline{t} \in F_b$ must also satisfy the constraint added. Consequently, any part of the feasible region of (2.6) can be reserved (not cut off).

(2.10) gives a criterion for judging the optimum solution for (2.6). The following theorem provides the convergence property of the algorithm.

Theorem 2.8. If the algorithm stop within finite steps, that is, (2.10) holds, then \underline{t}^n must be the optimum solution for (2.6). Otherwise, any limit point of sequence { \underline{t}^n } is the optimum solution for (2.6).

Proof. From the argument above we know that for $\forall n, F_b \subset C^n$. It is also true that $F_b \subset F^0$. So, $F_b \subset F^n$, where $F^n = F^{n-1} \cap C^n$, n = 1, 2, ...

If (2.10) holds, then $\underline{t}^n \in F_b$ is the optimum for (2.6) in F^{n-1} ($\supset F^n$) (by Theorem 2.6) and is the optimum for (2.6) in F_b .

Suppose { t^n } for n = 1, 2, ... is an infinite point sequence, and note that $\underline{t} \in [t^l, t^u], t^l > 0$ means \underline{t}^n is bounded. So, for a subsequence of { \underline{t}^n }, { \underline{t}^{n_k} }, there exists a limit point \underline{t}^* such that $\underline{t}^{n_k} \rightarrow \underline{t}^*$ as $k \rightarrow \infty$. So, $t_0^{n_k} \leq \min t_0$ for all $\underline{t} \in F_b$, and $t_0^* \leq \min t_0$ for all $\underline{t} \in F_b$.

In fact, we have

$$g_{k(n_{k})}\left(\underline{t}^{n_{k}}\right) = g_{k(n_{k})}\left[\underline{t}^{n_{k}}, \underline{e}(\underline{t}^{n_{k}})\right] > 1,$$

SO,

For $n_{n+1} > n_k$,

 $\underline{t}^{n_{k+1}} \in F^{n_{k+1}-1} \subset F^{n_{k+1}-2} \subset ... \subset F^{n_k} \subset C^{n_k}.$

Let d (t^{n_k} , C^{n_k}) denote the distance from t^{n_k} to C^{n_k} . Then by the definition,

 $d(t^{n_k}, C^{n_k}) = \min d(t^{n_k}, t), \text{ for all } t \in C^{n_k}.$

Note that $\underline{t}^{n_{k+1}} \in C^{n_k}$, so, d $(\underline{t}^{n_k}, C^{n_k}) \leq d(\underline{t}^{n_k}, \underline{t}^{n_{k+1}})$, and

 $d \ (\ t^{n_k}, \ t^{n_{k+1}}) \to 0 \quad \text{as} \ k \to \infty \quad (\text{since} \ t^{n_k} \to t^\star \quad \text{as} \ k \to \infty).$

Thus,

d $(\underline{1}^{n_k}, \underline{C}^{n_k}) \rightarrow 0$ as $k \rightarrow \infty$.

Now suppose that $\underline{t}^* \notin F_b$. Then $\exists k (1 \le k \le p)$ such that $g_k(\underline{t}^*) > 1$. Let

 $\boldsymbol{\varepsilon} = (1/2) [g_k(t^*) - 1] > 0.$

Then $\exists N > 0$ such that for $n_k > N$,

$$g_{k(n_k)}(\underline{t}^{n_k}, \underline{e}(\underline{t}^{n_k})) = g_{k(n_k)}(\underline{t}^{n_k}) > 1 + \varepsilon \quad (\varepsilon > 0, \underline{t}^{n_k} \in N(\underline{t}^*, \varepsilon))$$
(2.11)

On the other hand, $g_k(\underline{t}, \underline{e}(\underline{t}))$ is continuous uniformly (continuous on $[\underline{t}^l, \underline{t}^u]$).

Hence, for $\underline{t}, \underline{\tau} \in [\underline{t}^{l}, \underline{t}^{u}], \exists \delta > 0$ such that for $\varepsilon > 0$,

$$|g_{k}(\underline{t}', \underline{e}(\underline{\tau}') - g_{k}(\underline{t}'', \underline{e}(\underline{\tau}'')| < \varepsilon$$

if $||\underline{t}' - \underline{t}''|| + ||\underline{\tau}' - \underline{\tau}''|| < \delta$. But for $\forall \underline{t} \in C^{n_{k}}$, we have
 $g_{k(n_{k})}(\underline{t}, \underline{e}(\underline{t}^{n_{k}})) \leq 1$, (2.12)

so, for $\forall \mathbf{j} \in \mathbf{C}^{n_{k}}$, $|g_{k}|_{n_{k}}, (\mathbf{i}^{n_{k}})$

$$g_{k(n_{k})}(\underline{t}^{n_{k}}, \underline{e}(\underline{t}^{n_{k}})) - g_{k(n_{k})}(\underline{t}, \underline{e}(\underline{t}^{n_{k}})) \mid \geq |1 + \varepsilon - 1| = \varepsilon.$$

Thus, it must be true that

 $||\underline{t} - \underline{t}^{n_k}|| > \delta, \text{ for } \forall \underline{t} \in C^{n_k},$

that is,

 $\lim_{k \to \infty} d(\underline{t}^{n_k}, C^{n_k}) \neq 0.$

That contradicts the consequence that

 $d (\underline{t}^{n_k}, \mathbb{C}^{n_k}) \to 0 \text{ as } k \to \infty.$

Therefore, $\underline{t}^* = \lim_{k \to \infty} \underline{t}^n$, $\underline{t}^* \in F_b$ is the optimum for program (2.6).

The following illustrative example shows the application of the cutting plane algorithm.

Example 2.1. Find the solution of the problem that

$$\begin{array}{ll} \mbox{minimize} & t_1 \ , \\ \mbox{subject to} & g_1(t) = 0.574 \ t_1^{\ 0.388} \ t_2^{\ -0.755} + 1.148 \ t_1^{\ -0.612} \ t_2^{\ 0.225} \ \leq 1, \\ & g_2(t) = 0.083 \ t_1^{\ 1.529} \ t_2^{\ -0.529} + \ 0.083 \ t_1^{\ -0.471} \ t_2^{\ 1.471} \\ & + 1.169 \ t_1^{\ -0.471} \ t_2^{\ -0.529} \ \leq 1, \\ & 1.0 \ \leq \ t_1 \ \leq 5.5 \ , \ 1.0 \ \leq \ t_2 \ \leq 5.5. \\ \hline \mbox{Solution. Set } t^{(0)} = (t_1^{\ (0)}, \ t_2^{\ (0)})^T = (4, \ 4.5)^T. \end{array}$$

Suppose the problem is linearized at the initial operation point $\underline{t}^0 = (4, 4.5)^T$. The condensed constraints are

$$g_{1}(t, \underline{e}(t^{(0)})) = 1.720 t_{1}^{-0.305} t_{2}^{-0.083} \le 1,$$

$$g_{2}(t, \underline{e}(t^{(0)})) = 0.516 t_{1}^{-0.166} t_{2}^{-0.277} \le 1.$$

$$x_{1} = \log t_{1} - \log t_{1}^{-1} = \log t_{1}, \quad (\log t_{1}^{-1} = \log 1 = 0),$$

$$x_{2} = \log t_{2} - \log t_{2}^{-1} = \log t_{2}.$$

Consider solving the problem that

$$\begin{array}{ll} \text{minimize} & x_1, \\ \text{subject to} & -0.305 \, x_1 - 0.082 \, x_2 \leq - \ 0.524, \\ & 0.166 \, x_1 + 0.277 \, x_2 \leq 0.661, \\ & 0 \, \leq x_1 \leq \log 5.5, \\ & 0 \leq x_2 \leq \log 5.5. \end{array}$$

The solution of problem (2.13) is:

$$\begin{aligned} x_1^{(1)} &= 1.350, \quad x_2^{(1)} = 1.579, \\ t_1^{(1)} &= 3.858, \quad t_2^{(1)} = 4.852, \\ g_1(t^{(1)}) &= (0.574) \; (3.858)^{0.388} \; (4.852)^{-0.755} \\ &+ (1.148) \; (3.858)^{-0.612} \; (4.852)^{0.225} \\ &= 0.717, \\ g_2(t^{(1)}) &= (0.083)(3.858)^{1.529}(4.852)^{-0.529} \\ &+ (0.083)(3.858)^{-0.471}(4.852)^{1.472} \\ &+ (1.169)(3.858)^{-0.471}(4.852)^{-0.529} \\ &= 1.001. \end{aligned}$$

The second constraint is violated, so linearizing $g_2(t)$ at

$$t^{(1)} = (3.858, 4.852)^T$$
 (new operation point)

leads to

$$g_2(\underline{t}, \underline{\theta}(\underline{t}^{(1)})) = (0.498) t_1^{0.097} t_2^{0.369} \le 1.$$

Hence,

$$g_2(\underline{x}, \underline{e}(\underline{t}^{(1)})) = 0.097 x_1 + 0.369 x_2 \le 0.697.$$
 (2.14)

Adding (2.14) to the problem (2.13) leads to the program that

minimize
$$x_1$$
,
subject to $-0.305 x_1 - 0.082 x_2 \le -0.524$,
 $0.166 x_1 + 0.277 x_2 \le 0.661$,
 $0.097 x_1 + 0.369 x_2 \le 0.697$,
 $0 \le x_1 \le \log 5.5$,
 $0 \le x_2 \le \log 5.5$.
(2.15)

The solution of (2.15) is that

$$x_1^{(2)} = 1.354, \quad x_1^{(2)} = 1.566.$$

Hence,

$$t_1^{(2)} = 3.872, \quad t_1^{(2)} = 4.786.$$

 $g_1(t_1^{(2)}) < 1, \quad g_2(t_1^{(2)}) < 1.$

However, conducting more iterations can produce a better computation accuracy. That is,

$$t^{(0)} = (4.0, 4.5), \quad t^{(1)} = (3.872, 4.786),$$

 $t^{(2)} = (3.824, 4.824), \quad t^{(3)} = (3.823, 4.823),$

where $t^{(3)}$ is sufficiently close to t^* in the iteration.

1

CHAPTER III

THE GENERALIZED GEOMETRIC PROGRAMMING PROBLEM

In this chapter, we discuss the general features of the signomial geometric programming problem (SGP) and the method for solving the problem via transforming the problem to the complementary geometric programming problem (CGP). The basic results can be found in [2], [4], [8] and [12].

The problem that
minimize
$$g_0(t)$$
,
subject to $t \in F_{PS}$, (3.1)

where

$$F_{PS} = \{ \underline{t} \mid \underline{g}_{k}(\underline{t}) \leq \sigma_{k}, \text{ for } \sigma_{k} = 1 \text{ or } -1, k = 1, 2, ..., p \},\$$

$$g_{k}(\underline{t}) = \sum_{i \in [k]} s_{i} c_{i} \prod_{j=1}^{m} t_{j}^{a_{ij}}, \text{ for } s_{i} = 1 \text{ or } -1, c_{i}, t_{j} > 0, k = 0, 1, ..., p \text{ (signomial)}$$

is known as signomial geometric programming.

The associated dual program of (3.1) can seemingly be written as

maximize
$$V(\underline{d})$$
,
subject to $\underline{d} \in F_{DS}$, (3.2)

where

$$\begin{split} \mathsf{F}_{\mathsf{DS}} &= \{ \underline{d} \mid \sum_{i \in [0]} s_i d_i = \sigma_0, \ \sigma_0 = 1 \text{ or } -1, \\ \lambda_k &= \sigma_k \sum_{i \in [k]} s_i d_i \geq 0, \ \text{ for } k = 1, 2, ..., p, \ \lambda_0 = 1, \\ \sum_{i=1}^n s_i a_{ij} d_i &= 0, \ \text{ for } j = 1, 2, ..., m, \\ d_i \geq 0, \ \text{ for } i = 1, 2, ..., n \, \}, \end{split}$$

$$V(\underline{d}) = \sigma_0 \left[\prod_{k=0}^{p} \prod_{i \in [k]} (c_i \lambda_k / d_i)^{S_i d_i} \right] \sigma_0, \text{ for } \sigma_0 = 1 \text{ or } -1, \lambda_0 = 1.$$

For (3.1) and (3.2), we can set up the duality theory similar to that of posynomial geometric programming, and the conclusion is weaker and the proof (omitted) is more complicated. Before presenting the consequence of the theory we define the signum function such that

$$\sigma_0 = 1$$
, if $g_0(\underline{i}^*) \ge 0$,
 $\sigma_0 = -1$, if $g_0(\underline{i}^*) < 0$,

where t^* is the minimum for (3.1).

Theorem 3.1. For a local minimum point $\underline{t}^* \in F_{PS}$, there exists $\underline{d}^* \in F_{DS}$ such that $V(\underline{d}^*) = g_0(\underline{t}^*)$. Furtherfore, there is the relation between \underline{d}^* and \underline{t}^* that

$$\begin{aligned} &u_{i}(\underline{t}^{*}) = d_{i}^{*} \sigma_{0} g_{0}(\underline{t}^{*}), & \text{for } i \in [0], \\ &u_{i}(\underline{t}^{*}) = d_{i}^{*} / \lambda_{k}, & \text{for } i \in [k], \ k = 1, 2, ..., p, \quad (u_{i}(\underline{t}) = c_{i} \prod_{i=1}^{m} t_{j}^{a_{ij}}) \end{aligned}$$
(3.3)

and

$$d_i^* [g_k(\underline{t}^*) - 1] = 0, \text{ for } k = 1, 2, ..., p,$$
 (3.4)

or by taking the logarithmic form, (3.3) becomes

$$\sum_{j=1}^{m} a_{ij} \log t_j = \log \left(d_i^* \sigma_0 g_0(\underline{t}^*) / c_i \right), \text{ for } i \in [0],$$

$$\sum_{j=1}^{m} a_{ij} \log t_j = \log \left(d_i^* / c_i \lambda_k \right), \text{ for } i \in [k], k = 1, 2, ..., p.$$

It is important to note that the fact revealed by Theorem 3.1 is the relation between an arbitrary minimum point \underline{t}^* of (3.1) and only a stationary point \underline{d}^* of (3.2). In general, (3.1) is not a convex program, so, its minimum point is not necessarily unique and \underline{t}^* is not necessarily a global minimum point. Similarly, (3.2) is generally not a concave program, therefore, the dual vector \underline{d}^* that satisfies $V(\underline{d}^*) = g_0(\underline{t}^*)$ is not necessarily the maximum of (3.2) and is only a stationary point. In other words, since (3.1) does not have convexity property, it lacks the duality property that posynomial geometric programming possesses. In order to facilitate the discussion for (3.1), the concepts of quasiextreme point and quasiduality relation are introduced.

Definition 3.1. For (3.1), $\underline{t}^* \in F_{PS}$ is called *quasiminimum point* if \underline{t}^* is the Kuhn-Tucker point that satisfies the necessary condition for the local minimum, and the value of $g_0(\underline{t}^*)$ is called *quasiminimum value*, written as $g_0(\underline{t}^*) = quasimin g_0(\underline{t})$ for all $\underline{t} \in F_{PS}$. Accordingly, $\underline{d}^* \in F_{DS}$ is called *quasimaximum point*, and V(\underline{d}^*) is called *quasimaximum value* for (3.2), written as V(\underline{d}^*) = quasimax V(\underline{d}) for all $\underline{d} \in F_{DS}$.

Since Kuhn-Tucker point is not necessarily the optimum, by Definition 3.1, quasiminimum / quasimaximum point is not necessarily the local minimum / maximum point (only the candidate).

Theorem 3.2. Suppose that \underline{t}^* is the quasiminimum point of (3.1). Then there exists a quasimaximum point \underline{d}^* such that for all $\underline{t} \in F_{PS}$ and $\underline{d} \in F_{DS}$,

quasimin $g_{0}(\underline{t}) = quasimax V(\underline{d}),$

that is,

$$g_{0}(\underline{t}^{\star}) = V(\underline{d}^{\star}).$$

The duality relation concerning quasiminimum / quasimaximum is called quasiduality relation. Based on the relations shown by Theorem 3.1 and 3.2, we can use the duality method in solving (3.1), that is, solve (3.1) via finding the stationary points of (3.2).

Assume that F_{PS} is nonempty. Then for $\delta = 0$, the unique solution exists. It is a stationary point and needed to check whether it is an extremum point. For $\delta > 0$, the optimum solution is harder to find because δ independent variables are involved. The following example shows how the quasiduality relation can be used to solve the signomial geometric programming problem.

Example 3.1. Solve the problem that minimize $g_0(t) = -3 t_1^{-1} t_2^{-3} t_3 + (1/2) t_1^{-2} t_2^{-2} t_3$, subject to $g_1(t) = (1/4) t_1 t_2^{-2} t_3^{-1} - (1/3) t_1^{-3/4} t_2^{-5/2} t_3^{-1} \le -1$, $t_j > 0$, for j = 1, 2, 3.

Solution. The degree of difficulty of the problem is: $\delta = 4 - (3+1) = 0$. Note that $s_1 = -1$, $s_2 = 1$, $s_3 = 1$, $s_4 = -1$, $\sigma_1 = -1$, $\lambda_0 = 1$. Let $\sigma_0 = -1$, so,

$$\begin{aligned} -d_1 + d_2 &= -1, \\ d_1 - 2d_2 + d_3 - (3/4) & d_4 &= 0, \\ 3d_1 - 2d_2 + 2d_3 - (5/2) & d_4 &= 0, \\ -d_1 + d_2 - d_3 + d_4 &= 0. \end{aligned}$$

That is,

$$d_{1} - (1/2) d_{4} = 0,$$

$$d_{3} - (3/4) d_{4} = 0,$$

$$d_{2} - (1/4) d_{4} = 0,$$

$$d_{1} + d_{2} = -1.$$

So, we have $\underline{d}^* = (d_1^*, d_2^*, d_3^*, d_4^*)^T = (2, 1, 3, 4)^T$, and

$$\lambda_1 = s_1 \sum_{i=3}^{4} s_i d_i = (-1) [3 + (-1)(4)] = 1 > 0.$$

Thus, for stationary point \underline{t}^* and \underline{d}^* , the objective function value is:

$$g_0(\underline{t}^*) = V(\underline{d}^*) = (-1) \left[(3/2)^{-2} \left(\frac{1/2}{1} \right) \left(\frac{1/4}{3} \right)^3 \left(\frac{1/3}{4} \right)^{-4} \right]^{-1}$$
$$= (-1) \left(3^{2+3-4} 2^{-2+1+6-8} \right) = -3/8.$$

From (3.3) we know that

 $3 t_1^{-1} t_2^{-3} t_3 = (2) (-1) (-3/8) = 3/4,$

$$(1/2) t_1^{-2} t_2^{-2} t_3 = (1) (-1) (-3/8) = 3/8,$$

$$(1/4) t_1 t_2^{-2} t_3^{-1} = 3,$$

$$(1/3) t_1^{-3/4} t_2^{-5/2} t_3^{-1} = 4,$$

That is,

$$t_1^{-1} t_2^{-3} t_3 = 1/4,$$

$$t_1^{-2} t_2^{-2} t_3 = 3/4,$$

$$t_1 t_2^{2} t_3^{-1} = 12,$$

$$t_1^{3/4} t_2^{5/2} t_3^{-1} = 12.$$

The solution for this equation system is: $\mathbf{t}^* = (t_1^*, t_2^*, t_3^*)^T = (3^{-1}, 3^{-2}, 3^{-5}/4)^T$.

That $\delta = 0$ implies <u>t</u>^{*} is the only stationary point. The Hessian matrix of $g_0(t)$ is:

$$\mathsf{H}(\mathbf{i}^*) = \mathsf{H}(\mathbf{i}) \mid_{\mathbf{i}=\mathbf{i}^*} = (\mathsf{h}_{ij}(\mathbf{i})) \mid_{\mathbf{i}=\mathbf{i}^*} = (\frac{\partial^2 \mathsf{g}_0(\mathbf{i})}{\partial \mathsf{t}_i \partial \mathsf{t}_j}) \mid_{\mathbf{i}=\mathbf{i}^*},$$

where

$$\begin{split} h_{11}(\underline{i}) &= (-6) t_1^{-3} t_2^{-3} t_3 + 3 t_1^{-4} t_2^{-2} t_3, \\ h_{12}(\underline{i}) &= h_{21}(\underline{i}) = (-9) t_1^{-2} t_2^{-4} t_3 + 2 t_1^{-3} t_2^{-3} t_3, \\ h_{13}(\underline{i}) &= h_{31}(\underline{i}) = 3 t_1^{-2} t_2^{-3} - t_1^{-3} t_2^{-2}, \\ h_{22}(\underline{i}) &= (-36) t_1^{-1} t_2^{-5} t_3 + 3 t_1^{-2} t_2^{-4} t_3, \\ h_{23}(\underline{i}) &= h_{32}(\underline{i}) = 9 t_1^{-1} t_2^{-4} - t_1^{-2} t_2^{-3}, \\ h_{33}(\underline{i}) &= 0. \end{split}$$

Thus,

$$H(\underline{i}^{*}) = H(\underline{i})|_{\underline{i}=\underline{i}^{*}} = \begin{pmatrix} -3^{-5}/4 & -3^{-4}/4 & 0\\ -3^{-4}/4 & -3^{-5}/4 & 2(3^{7})\\ 0 & 2(3^{7}) & 0 \end{pmatrix}.$$

It is clear that $H(\underline{t}^*)$ is not positive definite. Therefore, \underline{t}^* is not a minimum point and is only a saddle point.

Definition 3.2. The problem

minimize t_o,

subject to
$$\underline{t} \in F_{PC}$$
, (3.3)

where

$$\begin{split} \mathsf{F}_{\mathsf{PC}} &= \{ \underline{t} \mid \mathsf{Q}_{\mathsf{k}}(\underline{t}) / \mathsf{P}_{\mathsf{k}}(\underline{t}) \leq 1, \text{ for } \underline{k} = 1, 2, ..., p \}, \\ \mathsf{Q}_{\mathsf{k}}(\underline{t}) &= \sum_{i \in [k]} c_{i} \prod_{j=0}^{m} t_{j}^{a_{ij}} \text{ and } \mathsf{P}_{\mathsf{k}}(\underline{t}) = \sum_{i \in [k]} d_{i} \prod_{j=0}^{m} t_{j}^{b_{ij}} \text{ are posynomials,} \end{split}$$

is called complementary geometric programming (CGP).

Theorem 3.3. (3.1) can be transformed to (3.3) equivalently.

Proof. By theorem 2.6, (3.1) is equivalent to the following problem

minimize
$$t_0$$
,
subject to $t \in F_{PS2}$, (3.4)

where

$$\begin{split} \mathsf{F}_{\mathsf{PS2}} &= \{ \underline{t} \ | \ t_0^{-1} \, g_0(\underline{t}) \leq 1, \\ & g_k(\underline{t}) \leq \sigma_k, \ \text{ for } \sigma_k = 1 \text{ or } -1, \ k = 1, 2, ..., p, \\ & t_j > 0, \ \text{ for } j = 0, 1, ..., m \}, \\ g_k(\underline{t}) &= \sum_{i \in [k]} s_i \, c_i \, \prod_{j=1}^m t_j^{a_{ij}}, \ \text{ for } s_i = 1 \text{ or } -1, \ c_j > 0, \ k = 0, 1, ..., p, \end{split}$$

In standard form, (3.4) can be written as:

```
minimize t_0,
subject to t \in F_{PS1}, (3.5)
```

where

$$\begin{split} F_{PS1} &= \{ \underline{1} \mid h_k(\underline{t}) \leq \sigma_k, \text{ for } \sigma_k = 1 \text{ or } -1, \ \sigma_0 = 1, \ k = 0, 1, ..., p, \\ t_j > 0, \text{ for } j = 0, 1, ..., m \}, \\ h_k(\underline{t}) &= \sum_{i \in [k]} s_i c_i \prod_{j=0}^m t_j^{a_{ij}}, \text{ for } s_i = 1 \text{ or } -1, c_i > 0, \ k = 0, 1, ..., p \text{ (signomial)}. \end{split}$$

Note that $h_k(t)$ can be expressed as:

 $h_k(\underline{t}) = h_k^{+}(\underline{t}) - h_k^{-}(\underline{t}) \leq \sigma_k^{-},$

where h_k^+ (t), h_k^- (t) are posynomials. So, for $\sigma_k = 1$, we have

 $h_{k}^{+}(t) - h_{k}^{-}(t) \le 1,$

that is,

 $h_k^+(\underline{t}) / [1 + h_k^-(\underline{t})] \le 1.$ Let $Q_k(\underline{t}) = h_k^+(\underline{t})$ and $P_k(\underline{t}) = 1 + h_k^-(\underline{t})$. Then we reach $Q_k(\underline{t}) / P_k(\underline{t}) \le 1$, for k = 0, 1, ..., p.

For $\sigma_k = -1$, we have

 $h_{k}^{+}(\underline{i}) - h_{k}^{-}(\underline{i}) \leq -1,$

that is,

 $(1 + h_{k}^{+}(\underline{t})) / h_{k}^{-}(\underline{t}) \le 1.$

Let $Q_k(\underline{i}) = 1 + h_k^+(\underline{i})$ and $P_k(\underline{i}) = h_k^+(\underline{i})$. Then we also reach

 $Q_k(t) / P_k(t) \le 1$, for k = 0, 1, ..., p.

So, via condensing $P_k(t)$, we obtain the approximating problem of (3.5) that

where

$$\begin{split} & \mathsf{F}_{\mathsf{PA}} = \{ \underline{t} \mid \mathsf{Q}_{\mathsf{k}}(\underline{t}) / \mathcal{P}_{k}(\underline{t}, \underline{e}(\underline{t})) \leq 1, \quad \text{for } \mathbf{k} = 0, 1, ..., p \} \} \\ & \mathsf{Q}_{\mathsf{k}}(\underline{t}) = \sum_{i \in [k]} c_{i} \quad \prod_{j=1}^{m} t_{j}^{a_{ij}} , \\ & \mathcal{P}_{k}(\underline{t}, \underline{e}(\underline{t})) = \prod_{i \in [k]} \{ \left[d_{i} / e_{i}(\underline{t}) \right] \quad \prod_{j=0}^{m} t_{j}^{b_{ij}} \} \stackrel{e_{i}(\underline{t})}{e_{i}(\underline{t})} , \\ & e_{i}(\underline{t}) \in (0, 1), \quad \text{for } i = 1, 2, ..., n, \quad \sum_{i \in [k]} e_{i}(\underline{t}) = 1, \end{split}$$

which is a posynomial geometric program with smaller degree of difficulty than that of (3.1) since the number of terms of $Q_k(t)$ equals that of $(1+h_k^+(t))$.

Theorem 3.4. If \underline{t}^* is the optimum solution of (3.6), then \underline{t}^* must be a feasible solution of (3.3).

Proof. By Theorem 2.1 we know that

 $P_k(\underline{t}, \underline{e}(\underline{t})) \leq P_k(\underline{t}).$

So,

$$\mathsf{Q}_{\mathsf{k}}(\underline{\mathsf{t}}) / \mathsf{P}_{\mathsf{k}}(\underline{\mathsf{t}}) \leq \mathsf{Q}_{\mathsf{k}}(\underline{\mathsf{t}}) / \mathsf{P}_{\mathsf{k}}(\underline{\mathsf{t}}, \underline{e}(\underline{\mathsf{t}})).$$

For all t > 0 such that

 $Q_k(\underline{t}) / P_k(\underline{t}, \underline{e}(\underline{t})) \le 1,$

it must hold that

 $Q_k(\underline{t}) / P_k(\underline{t}) \le 1$,

and the converse is not necessarily true. Hence, if $\underline{t} \in F_{PA}$, then $\underline{t} \in F_{PC}$, that is, $F_{PA} \subset F_{PC}$. Therefore, $\underline{t}^* \in F_{PA}$ implies $\underline{t}^* \in F_{PC}$.

Theorem 3.5. Suppose that $\underline{i}^* \in F_{PC}$ is the optimum solution of (3.6) with $P_k(\underline{i}, \underline{e}(\underline{i}^*))$ (\underline{i}^* acts as the operation point here). Then \underline{i}^* is the optimum solution of (3.3).

Proof. Suppose that \underline{i}^* is not the optimum solution of (3.3) and that $\underline{i}^1 \neq \underline{i}^*$ is the optimum solution of (3.3). Then $\underline{i}^1 \in F_{PC}$ and $\underline{i}^1 \notin F_{PA}$. By Theorem 2.3 we have

 $P_k(\underline{t}^1, \underline{e}(\underline{t}^1)) = P_k(\underline{t}^1).$

Note that

 $F_{PA} \subset F_{PC}$.

So, $\underline{t}^1 \in F_{PA}$ and \underline{t}^1 is the optimum solution of (3.6). This contracts $\underline{t}^1 \notin F_{PA}$. Thus, $\underline{t}^1 = \underline{t}^*$.

Based on Theorem 3.4 and 3.5, we can constructe an algorithm for solving (3.1) as follows.

Step 1. Transform (3.1) to (3.5). Fix an initial feasible solution $\underline{t}^0 \in F_{PC}$, and give $\varepsilon > 0$ as tolerant error for an acceptable computation accuracy. Set n = 1.

Step 2. Condense $P_k(\underline{t})$ at \underline{t}^{n-1} to form (3.6) with $P_k(\underline{t}, \underline{e}(\underline{t}^{n-1}))$.

Step 3. Solve (3.6) that is a posynomial geometric program and find its optimum solution t^* .

Step 4. Check whether $|| \underline{t}^{n-1} - \underline{t}^* || < \varepsilon$. If so, \underline{t}^* is the optimum for (3.5) and for (3.1). Otherwise, increase n by 1 and let $\underline{t}^{n-1} = \underline{t}^*$. Turn to Step 2.

The following example shows the application of the algorithm to solving SGP problem.

Example 3.2. Solve the problem that

 $\begin{array}{ll} \text{minimize} & g_0(\underline{t}) = 2 \, t_1 \, t_2^{1/2} + t_2 \, t_3^{-1} \, t_4^{\,2} + t_1^{-2} \, t_2^{-1} \, t_3^{\,2} \, , \\ \text{subject to} & g_1(\underline{t}) = t_1 \, t_2^{1/2} \, t_3^{\,-1} \, t_2^{-1} \, t_4^{\,2} \leq 1 \, , \\ & g_2(\underline{t}) = t_1 \, t_2 \, t_3^{\,2} \geq 5 \, , \\ & t_j > 0 \, , \quad \text{for } j = 1 \, , \, 2 \, , \, 3 \, , \, 4 \, . \end{array}$

Solution. The problem can be transformed to:

minimize t_o,

subject to
$$2 t_0^{-1} t_1 t_2^{1/2} + t_0^{-1} t_2 t_3^{-1} t_4^2 + t_0^{-1} t_1^{-2} t_2^{-1} t_3^2 \le 1$$
,
 $(t_1 t_2^{1/2} t_3) / (1 + t_2^{-1} t_4^2) \le 1$,
 $5 t_1^{-1} t_2^{-1} t_3^{-1} \le 1$,
 $t_j > 0$, for $j = 0, 1, 2, 3, 4$.
 $P_2(t, \underline{e}(t^0)) = [1 + (t_2^0)^{-1} (t_4^0)^2] (\frac{t_2}{t_2^0}) (-1)(t_2^0)^{-1} (t_4^0)^2 / [1 + (t_2^0)^{-1} (t_4^0)^2]$

$$(\frac{t_4}{t_4^0})^{2(t_2^0)^{-1}(t_4^0)^2/[1+(t_2^0)^{-1}(t_4^0)^2]}$$

Let $\underline{t}^0 = (10, 0.1, 5, 2)^T$, and $\underline{t}_0 = 8.905$.

Then,

$$t^{0} \in F = \{ t \downarrow t_{1} t_{2}^{1/2} t_{3}^{-1} t_{2}^{-1} t_{4}^{2} \leq 1, \\ t_{1} t_{2} t_{3}^{2} \geq 5, \\ t_{j}^{-0, \text{ for } j} = 1, 2, 3, 4 \}.$$

$$P_{2}(t, \underline{e}(t^{0})) = (1.12) t_{2}^{-0.9756} t_{4}^{1.95}.$$

The corresponding problem of (3.6) is that

$$\begin{array}{ll} \text{minimize} & t_0^{-1}, \\ \text{subject to} & 2 t_0^{-1} t_1^{-1} t_2^{-1/2} + t_0^{-1} t_2^{-1} t_3^{-1} t_4^{-1} + t_0^{-1} t_1^{-2} t_2^{-1} t_3^{-2} \leq 1, \\ & (0.89) t_1^{-1} t_2^{-1.4756} t_3^{-1.95} \leq 1, \\ & 5 t_1^{-1} t_2^{-1} t_3^{-1} \leq 1, \\ & t_j > 0, \quad \text{for } j = 0, \, 1, \, 2, \, 3, \, 4. \end{array}$$

The degree of difficulty of the problem is $\delta = 6 - (5 + 1) = 0$. We can find the only solution for the problem and the value of the objective function, that is,

$$t_1^* = 3.27, t_2^* = 0.54, t_3^* = 2.82, t_4^* = 1.9,$$

 $t_0^* = g_0(t^*) = 6.8852.$

Since $||t^* - t^2|| > \varepsilon$, more iterations are needed. Let

$$\mathbf{t}^1 = \mathbf{t}^* = (3.27, 0.54, 2.82, 1.9, 6.8852)^T$$
.

Then computation shows that

 $i \cdot \$ 1$, $t^2 = (3.066, 0.593, 2.75, 6.7836)^T$. $t^3 = (3.047, 0.598, 2.744, 1.805, 6.7786)^T$. $t^4 = (3.051, 0.597, 2.745, 1.808, 6.7815)^T$.

And $||t^* - t^4|| < \epsilon$. Thus, t^4 is the optimum solution for the problem.

CHAPTER IV

THE REVERSED GEOMETRIC PROGRAMMING PROBLEM

In this chapter, we consider the reversed geometric programming problem (RGP), which is a special case of the signomial geometric programming problem (SGP). Since the SGP problem can be transformed to the RGP problem, this chapter can be considered as an extension of the preceding one in providing an alternative approach to solve the SGP problem other than that presented in the previous chapter. The main content can be found in [5] and [11].

The reversed geometric programming problem is of the form that

minimize
$$g_0(\underline{t})$$
,
subject to $\underline{t} \in F_{PR}$, (4.1)

where

$$\begin{aligned} \mathsf{F}_{\mathsf{PR}} &= \{1 \mid g_k(t) \leq 1, \ \text{ for } k = 1, 2, ..., l, \\ g_k(t) \geq 1, \ \text{ for } k = l+1, \ l+2, ..., p, \\ t_j > 0, \ \text{ for } j = 1, 2, ..., m \}, \end{aligned}$$

$$\begin{split} g_{k}(\underline{t}) &= \sum_{i \in [k]} u_{i}(\underline{t}), & \text{for } k = 0, 1, ..., p, \\ u_{i}(\underline{t}) &= c_{i} \prod_{j=1}^{m} t_{j}^{a_{ij}}, & \text{for } i \in [k], \ k = 0, 1, ..., l, \ c_{i} \geq 0, \ a_{ij} \in R, \\ u_{i}(\underline{t}) &= c_{i} \prod_{j=1}^{m} t_{j}^{-a_{ij}}, & \text{for } i \in [k], \ k = l+1, l+2, ..., p, \ c_{i} \geq 0, \ a_{ij} \in R. \end{split}$$

Note the exponents in $u_i(t)$ for $i \in [k]$, k = l+1, l+2, ..., p, are written as $(-a_{ij})$ here to facilitate the discussion later.

The associated dual problem of (4.1) can be expressed as:

maximize
$$V(\underline{d})$$
,
subject to $\underline{d} \in F_{DR}$, (4.2)

where

$$\begin{split} F_{DR} &= \{ \ \underline{d} \ | \ \lambda_0 = 1, \\ \lambda_k &= \sum_{i \in [k]} d_i \ , \quad \text{for } k = 0, \ 1, \ ..., \ p, \\ \sum_{i=1}^n \ a_{ij} \, d_i = 0 \ , \quad \text{for } j = 1, \ 2, \ ..., \ m, \quad (\ i.e., \ A^T \underline{d} = \underline{0} \), \\ d_i &\geq 0 \ , \quad \text{for } i = 1, \ 2, \ ..., \ n, \\ V(\underline{d}) &= \prod_{k=0}^l \ \prod_{i \in [k]} (c_i / \ d_i) \ \overset{d_i}{\underset{k=l+1}{\overset{p}{\prod_{i \in [k]}}} \prod_{i \in [k]} (c_i / \ d_i) \ \overset{-d_i}{\underset{k=l+1}{\overset{l}{\prod_{i \in [k]}}} (\lambda_k)^{\lambda_k} \ \underset{k=l+1}{\overset{p}{\prod_{i \in [k]}} (\lambda_k)^{-\lambda_k} \ . \end{split}$$

As shown in the following way, problem (3.1) can be transformed to problem (4.1).

Let $M_{PS} = \inf g_0(\underline{t})$ for all $\underline{t} \in F_{PS}$. Then for $M_{PS} \ge 0$, (3.1) is equivalent to the problem that

minimize
$$t_0$$
,
subject to $\underline{t} \in F_{PR1}$, (4.3)

where

$$\begin{split} \mathsf{F}_{\mathsf{PR1}} &= \{ \ t \mid (t_0)^{-1} \ g_0(t) \leq 1, \\ &\quad \mathsf{g}_k(t) \leq \sigma_k \ , \ \text{ for } \sigma_k = 1 \text{ or } (-1), \ k = 1, 2, ..., p, \\ &\quad t_j > 0 \ , \ \text{ for } j = 1, 2, ..., m \ \}. \end{split}$$

Otherwise, for $M_{PS} < 0$, equivalent to:

minimize
$$(t_0)^{-1}$$
,
subject to $t \in F_{PR2}$, (4.4)

where

$$\begin{split} & \mathsf{F}_{\mathsf{PR2}} = \{ t \mid (t_0)^{-1} \, g_0(t) \leq -1, \\ & \mathsf{g}_k(t) \leq \sigma_k, \quad \text{for } \sigma_k = 1 \text{ or } (-1), \ k = 1, 2, ..., p, \\ & \mathsf{t}_j > 0 \ , \quad \text{for } j = 1, 2, ..., m \ \}. \end{split}$$

If it is hard to judge the sign of M_{PS} in advance, we need to solve (4.3) and (4.4).

In general, $g_k(t)$ (the constraint of (3.1)) can be expressed as:

 $g_{k}(t) = h_{1}(t) - h_{2}(t),$

where $h_1(\underline{t})$, $h_2(\underline{t})$ are posynomials. So, for $g_k(\underline{t}) \leq \sigma_k$, if $\sigma_k = -1$, then

 $h_1(\underline{t}) - h_2(\underline{t}) \le -1,$

By introducing $t_{m+1} > 0$ such that

$$h_1(t) + 1 \le t_{m+1} \le h_2(t),$$

we get

$$(t_{m+1})^{-1} (h_1(\underline{t}) + 1) \le 1,$$

 $(t_{m+1})^{-1} h_2(\underline{t}) \ge 1.$

If $\sigma_k = 1$, then

$$h_1(\underline{t}) \leq t_{m+1} \leq h_2(\underline{t}) + 1,$$

we get

$$(t_{m+1})^{-1} h_1(\underline{t}) \le 1,$$

 $(t_{m+1})^{-1} (h_2(\underline{t}) + 1) \ge 1.$

As an illustrative example, consider the transformation of the problem that

$$\begin{array}{l} \mbox{minimize} & \left[\begin{array}{c} \displaystyle \frac{(f_1(\underline{t}))^{1/2} + f_2(\underline{t})}{(f_3(\underline{t}))^{1/2} + f_4(\underline{t})} \end{array} \right]^{1/2}, \\ \mbox{subject to } t_j > 0, \ \mbox{for } j = 1, 2, ..., m \\ \mbox{where } f_i(\underline{t}) > 0 \ \mbox{for } i = 1, 3, \ \ (f_1(\underline{t}))^{1/2} + f_2(\underline{t}) > 0, \ \ (f_3(\underline{t}))^{1/2} + f_4(\underline{t}) > 0. \end{array}$$

Clearly, (4.5) is equivalent to the problem that

$$\begin{array}{l} \text{minimize} \quad (t_0)^{1/2}, \\ \text{subject to} \quad f_1(\underline{t}) > 0, \quad f_3(\underline{t}) > 0, \\ \quad (f_1(\underline{t}))^{1/2} + f_2(\underline{t}) > 0, \\ \quad (f_3(\underline{t}))^{1/2} + f_4(\underline{t}) > 0, \\ \quad (f_1(\underline{t}))^{1/2} + f_2(\underline{t}) \leq t_0 \left(f_3(\underline{t})\right)^{1/2} + t_0 f_4(\underline{t}), \\ \quad t_j > 0, \quad \text{for } j = 0, 1, ..., m. \end{array}$$

That is a SGP problem.

By introducing t_{m+1} , t_{m+2} , $t_{m+3} > 0$ as bound variables, (4.6) can be transformed to the problem that

$$\begin{array}{ll} \text{minimize} & (t_0)^{1/2}, \\ \text{subject to} & t_{m+1} \leq f_1(\underline{t}) \leq t_{m+2}, \\ & t_{m+3} \leq f_3(\underline{t}), \\ & (t_{m+1})^{1/2} + f_2(\underline{t}) \geq 0, \\ & (t_{m+3})^{1/2} + f_4(\underline{t}) \geq 0, \\ & - (t_{m+2})^{1/2} - f_2(\underline{t}) + t_0(t_{m+3})^{1/2} + t_0 f_4(\underline{t}) \geq 0, \\ & t_j \geq 0, \quad \text{for } j = 0, 1, ..., m, m+1, m+2, m+3. \end{array}$$

Suppose that

$$f_{2}(t) = h_{2}'(t) - h_{2}''(t),$$

$$f_{4}(t) = h_{4}'(t) - h_{4}''(t),$$

where $h'_i(t)$, $h''_i(t)$ for i = 2, 4 are posynomials. So,

$$(t_{m+1})^{1/2} + h_2'(\underline{t}) \ge h_2''(\underline{t}),$$

 $(t_{m+1})^{1/2} + h_2'(\underline{t}) \ge t_{m+4} \ge h_2''(\underline{t}),$

then we have

$$(t_{m+1})^{1/2} (t_{m+4})^{-1} + (t_{m+4})^{-1} h_2'(t) \ge 1,$$

$$(t_{m+4})^{-1} h_2''(t) \le 1.$$

For the same reason, we have

$$(t_{m+3})^{1/2} (t_{m+5})^{-1} + (t_{m+5})^{-1} h_4'(\underline{t}) \ge 1,$$

 $(t_{m+5})^{-1} h_4''(\underline{t}) \le 1.$

Therefore, for $t_{m+6} > 0$, we have

$$t_0 (t_{m+3})^{1/2} (t_{m+6})^{-1} + t_0 (t_{m+6})^{-1} h_4'(t) + (t_{m+6})^{-1} h_2''(t) \ge 1,$$

$$(t_{m+2})^{1/2} (t_{m+6})^{-1} + (t_{m+6})^{-1} h_2'(t) + t_0 (t_{m+6})^{-1} h_4''(t) \le 1.$$

Thus, we reach the following reversed geometric programming problem, which is equivalent to (4.5), (4.6) and (4.7). That is,

$$\begin{array}{ll} \text{minimize} & (t_0)^{1/2}, \\ \text{subject to} & (t_{m+1})^{-1} \ f_1(\underline{t}) \ \geq 1, \\ & (t_{m+2})^{-1} \ f_1(\underline{t}) \ \leq 1, \\ & (t_{m+3})^{-1} \ f_3(\underline{t}) \ \geq 1, \\ & (t_{m+4})^{1/2} \ (t_{m+4})^{-1} + (t_{m+4})^{-1} \ h_2'(\underline{t}) \ \geq 1, \\ & (t_{m+4})^{-1} \ h_2''(\underline{t}) \ \leq 1, \\ & (t_{m+3})^{1/2} \ (t_{m+5})^{-1} + (t_{m+5})^{-1} \ h_4'(\underline{t}) \ \geq 1, \\ & (t_{m+5})^{-1} \ h_4''(\underline{t}) \ \leq 1, \\ & (t_{m+5})^{-1} \ h_4''(\underline{t}) \ \leq 1, \\ & t_0 \ (t_{m+3})^{1/2} \ (t_{m+6})^{-1} + t_0 \ (t_{m+6})^{-1} \ h_4''(\underline{t}) + (t_{m+6})^{-1} \ h_2''(\underline{t}) \ \geq 1, \\ & (t_{m+2})^{1/2} \ (t_{m+6})^{-1} + (t_{m+6})^{-1} \ h_2''(\underline{t}) \ + t_0 \ (t_{m+6})^{-1} \ h_4''(\underline{t}) \ \leq 1, \\ & t_1 \ > 0, \ \text{ for } j = 1, 2, ..., m, m+1, ..., m+6. \end{array}$$

Consider the Lagrange problem of (4.1) that find $\underline{t}^* \in F_{PR}$ and Lagrange multiplier $\underline{\mu}^* = (\mu_1^*, \mu_2^*, ..., \mu_p^*)^T$ such that

where

$$L(t, \mu) = g_0(t) + \sum_{k=1}^{l} \mu_k (g_k(t) - 1) + \sum_{k=l+1}^{p} \mu_k (1 - g_k(t))$$

is Lagrange function of (4.1),

For (4.1), let

$$S = \{ k \mid g_{k}(\underline{i}^{*}) = 1, \text{ for } 1 \le k \le p \},$$

$$A_{kj} = \sum_{i \in [k]} a_{ij} u_{i}(\underline{i}^{*}), \text{ for } k \in S, j = 1, 2, ..., m,$$

$$F_{t} = \{ \underline{i} \mid \sum_{j=1}^{m} A_{kj} \ (\log t_{j} - \log t_{j}^{*}) \le 0, \text{ for } k \in S \}.$$
(4.8)

Theorem 4.1. Suppose $\underline{t}^* \in F_{PR}$. Then there exists $\underline{\mu}^*$ such that $(\underline{t}^*, \underline{\mu}^*)$ is the solution of the Lagrange problem of (4.1) if and only if for all $\underline{t} \in F_t$,

$$g_0(t^*) \leq g_0(t).$$

Proof. (For necessity) Note that for k = 0, 1, ..., p,

. . .

$$\frac{\partial g_{k}(t)}{\partial t_{j}} = (1/t_{j}) \frac{\partial}{\partial t_{j}} \left(\sum_{i \in [k]} c_{i} \prod_{i=1}^{m} t_{i} \right)^{a_{ii}} a_{ij} = (1/t_{j}) \sum_{i \in [k]} u_{i}(t) a_{ij}.$$

So, for k = 0,

$$\sum_{i \in [0]} u_i(\underline{t}) a_{ij} = t_j \frac{\partial g_0(\underline{t})}{\partial t_j}.$$

For k = 1, 2, ..., p,

$$A_{kj} = \sum_{i \in [k]} u_i(t^*) a_{ij} = t_j \frac{\partial g_k(t^*)}{\partial t_j}.$$

So

$$\frac{\partial}{\partial t_{j}} \left[\mu_{k} \left(g_{k}(t) - 1 \right) \right] = \left(\mu_{k} / t_{j} \right) A_{kj}, \text{ for } k = 1, 2, ..., l,$$

$$\frac{\partial}{\partial t_{j}} \left[\mu_{k} \left(1 - g_{k}(t) \right) \right] = \left(\mu_{k} / t_{j} \right) A_{kj}, \text{ for } k = l+1, l+2, ..., p,$$

Thus, for $k \in S$,

$$\begin{aligned} \frac{\partial}{\partial t_j} \left[L\left(\underline{t}^*, \underline{\mu}^*\right) \right] &= \frac{\partial g_0(\underline{t})}{\partial t_j} + (1/t_j^*) \sum_{k \in S} \mu_k^* A_{kj} \\ \text{Let } x_j &= \log t_j. \text{ Then for } k = 0, 1, ..., p, \text{ we know} \\ g_k(\underline{t}) &= \sum_{i \in [k]} u_i(\underline{t}) = \sum_{i \in [k]} c_i \prod_{j=1}^m t_j^{a_{ij}}, \end{aligned}$$

S0,

$$f_{k}(\underline{x}) = \sum_{i \in [k]} V_{i}(\underline{x}) = \sum_{i \in [k]} c_{i} \exp \left\{ \sum_{j=1}^{m} a_{ij} x_{j} \right\},$$

and for all $i \in [k]$, k = 0, 1, ..., p,

$$u_i(\underline{t}) = V_i(\underline{x}), \ g_k(\underline{t}) = f_k(\underline{x}).$$

Hence, for j = 1, 2, ..., m,

$$\frac{\partial f_k(\underline{x})}{\partial x_j} = \sum_{i \in [k]} c_i a_{ij} \exp \left\{ \sum_{j=1}^m a_{ij} x_j \right\} = \sum_{i \in [k]} V_i(\underline{x}) a_{ij} = \sum_{i \in [k]} u_i(\underline{t}) a_{ij}.$$

That means that

$$\nabla_{\underline{X}} f_0(\underline{X}) = \left(t_1 \frac{\partial g_0(\underline{t})}{\partial t_1}, t_2 \frac{\partial g_0(\underline{t})}{\partial t_2}, ..., t_m \frac{\partial g_0(\underline{t})}{\partial t_m} \right)^T, \text{ for } k = 0,$$

$$\nabla_{\underline{X}} f_k(\underline{X}^*) = \left(A_{k1}, A_{k2}, ..., A_{km} \right)^T, \text{ for } k = 1, 2, ..., l,$$

and

$$\nabla_{\underline{X}} f_k(\underline{X}^*) = (-A_{k1}, -A_{k2}, ..., -A_{km})^T$$
, for $k = l+1, l+2, ..., p$.

By the definition we know that $\exists \mu^*$ such that (\underline{t}^*, μ^*) forms a solution of the Lagrange problem if and only if for $k \in S$,

$$\nabla_{\underline{t}} \ \mathsf{L} \left(\underline{t}^{\star}, \, \underline{\mu}^{\star} \right) = 0.$$

That is,

$$\frac{\partial g_0(j^*)}{\partial t_j} + (1/t_j^*) \sum_{k \in S} (\mu_k^*) A_{kj} = 0, \text{ for } t_j^* > 0,$$

i.e.,

$$t_{j}^{*} \frac{\partial g_{0}(\underline{1}^{*})}{\partial t_{j}} = -\sum_{k \in S} (\mu_{k}^{*}) A_{kj}, \text{ for } \mu_{k}^{*} \ge 0, j = 1, 2, ..., m.$$

By Lemma 1.1, for $\underline{t} \in F_{\underline{t}}$, that is, for all $\underline{t} > 0$, such that for j = 1, 2, ..., m,

$$-\sum_{j=1}^{m} A_{kj} (\log t_j - \log t_j^*) \ge 0$$
, (the tangentially optimal property),

i.e.,

$$\begin{bmatrix} \nabla_{\underline{X}} f_k(\underline{x}^*) \end{bmatrix}^T (\underline{x} - \underline{x}^*) \le 0, \text{ for } k = 1, 2, ..., l, \begin{bmatrix} \nabla_{\underline{X}} f_k(\underline{x}^*) \end{bmatrix}^T (\underline{x} - \underline{x}^*) \ge 0, \text{ for } k = l+1, l+2, ..., p$$

It holds that

$$\sum_{j=1}^{m} [t_j^* \frac{\partial g_0(t^*)}{\partial t_j}] (\log t_j - \log t_j^*) \ge 0 \quad (by \; Farkas \; lemma),$$

that is,

 $\left[\ \nabla_{\underline{X}} \ f_0(\underline{x}^\star) \ \right]^T \left(\ \underline{x} - \underline{x}^\star \ \right) \geq 0.$

Note that $g_0(\underline{t})$ is posynomial and so $f_0(\underline{x})$ is convex, i.e.,

$$g_{0}(\mathfrak{t}) - g_{0}(\mathfrak{t}^{*}) = f_{0}(\mathfrak{X}) - f_{0}(\mathfrak{X}^{*}) \geq [\nabla_{\mathfrak{X}} f_{0}(\mathfrak{X}^{*})]^{\mathsf{T}} (\mathfrak{X} - \mathfrak{X}^{*}).$$

Thus, if

 $\nabla_{\underline{t}} L(\underline{t}^*, \underline{\mu}^*) = 0, \text{ for } k \in S,$

 $(\Leftrightarrow \exists \mu^* \text{ such that } (\underline{t}^*, \underline{\mu}^*) \text{ forms Lagrange problem solution}), \text{ then}$ $g_0(\underline{t}) \ge g_0(\underline{t}^*), \text{ for all } \underline{t} \in F_t.$

(For sufficiency) Conversely, suppose for all $\underline{t} \in F_t$, it holds that

 $g_0(t^*) \leq g_0(t),$

i.e.,

$$f_0(\underline{x}^*) \le f_0(\underline{x}),$$

for x such that

$$\begin{bmatrix} \nabla_{\underline{X}} f_k(\underline{X}^*) \end{bmatrix}^T (\underline{X} - \underline{X}^*) \le 0, \text{ for } k = 1, 2, ..., l, \begin{bmatrix} \nabla_{\underline{X}} f_k(\underline{X}^*) \end{bmatrix}^T (\underline{X} - \underline{X}^*) \ge 0, \text{ for } k = l+1, l+2, ..., p.$$

Then, let $\underline{y} = \underline{x}^* + \xi (\underline{x} - \underline{x}^*)$, $\xi > 0$. Then since

$$\begin{bmatrix} \nabla_{\underline{X}} f_{k}(\underline{X}^{*}) \end{bmatrix}^{\mathsf{T}} (\underline{y} - \underline{X}^{*}) = \begin{bmatrix} \nabla_{\underline{X}} f_{k}(\underline{X}^{*}) \end{bmatrix}^{\mathsf{T}} [\underline{X}^{*} + \xi (\underline{x} - \underline{X}^{*}) - \underline{X}^{*}]$$
$$= \xi \begin{bmatrix} \nabla_{\underline{X}} f_{k}(\underline{X}^{*}) \end{bmatrix}^{\mathsf{T}} (\underline{x} - \underline{X}^{*}),$$

we have

$$\begin{bmatrix} \nabla_{\underline{X}} f_k(\underline{x}^*) \end{bmatrix}^T (\underline{y} - \underline{x}^*) \le 0, \text{ for } k = 1, 2, ..., l,$$
$$\begin{bmatrix} \nabla_{\underline{X}} f_k(\underline{x}^*) \end{bmatrix}^T (\underline{y} - \underline{x}^*) \ge 0, \text{ for } k = l+1, l+2, ..., p.$$

Subsequently, $f_0(\underline{x}^*) \leq f_0(\underline{y})$ (for such a \underline{y}), and take $\xi \to 0$ ($\underline{y} \to \underline{x}^*$), then

$$\begin{split} &\lim_{\xi \to 0} \frac{f_0(\underline{y}) - f_0(\underline{x}^*)}{\underline{y} - \underline{x}^*} = \nabla_{\underline{x}} f_0(\underline{x}^*) , \\ &[\nabla_{\underline{x}} f_0(\underline{x}^*)]^T (\underline{x} - \underline{x}^*) = (1/\xi) [\nabla_{\underline{x}} f_0(\underline{x}^*)]^T (\underline{y} - \underline{x}^*) \ge 0 \end{split}$$

i.e.,

$$\sum_{j=1}^{m} \left[(t_j^*) \frac{\partial g_0(\underline{t}^*)}{\partial t_j} \right] (\log t_j - \log t_j^*) \ge 0, \text{ for all } \underline{t} \in F_t.$$

That is equivalent to that $\nabla_{\underline{1}} L(\underline{1}^*, \underline{\mu}^*) = 0$ (k \in S), and ($\underline{1}^*, \underline{\mu}^*$) forms a Lagrange solution.

Since (4.1) lacks convexity, it can not maintain the properties that (1.1) possesses, for instance, any local optimum solution is its global optimum solution, $M_D \leq M_P$, etc. For (4.1), like the solution(s) for other nonlinear programming problems, its local optimum solution(s) must be a part of its Lagrange problem solution(s), so we can find its local optimum solution(s) from within the solution(s) of its Lagrange problem. Conversely, in general (for p > l), the solution of Lagrange problem t^* does not necessarily mean the global or even local optimum solution. However, Theorem 4.1 shows that although t^* is not the local optimum solution for the problem, yet within a

certain range, that is, for all $\underline{t} \in F_t$, \underline{t}^* remains the optimum. That property is called the tangential optimality.

Definition 4.1. $\underline{t}^* \in F_{PR}$, $\underline{d}^* \in F_{DR}$ are called *primal equilibrium solution* and *dual equilibrium solution*, respectively, if it holds that

$$\begin{aligned} d_{i}^{*} g_{0}(\mathbf{i}^{*}) &= u_{i}(\mathbf{i}^{*}), & \text{for } i \in [0], \\ d_{i}^{*} &= \lambda_{k}^{*} u_{i}(\mathbf{i}^{*}), & \text{for } i \in [k], \ k = 1, 2, ..., p, \end{aligned}$$
(4.9)

where

$$\lambda_k^{\star} = \sum_{i \in [k]} d_i^{\star} .$$

Here, $g_0(t^*)$ and $V(d^*)$ are called *primal equilibrium value* and *dual equilibrium value*, respectively.

There exists the following relation between the equilibrium solution and the Lagrange problem solution.

Theorem 4.2. Suppose that $\underline{1}^*$ and $\underline{\mu}^*$ are a pair of solutions of Lagrange problem. Then $\underline{1}^*$ and \underline{d}^* defined by

$$\begin{aligned} d_{i}^{*} &= u_{i}(\underline{t}^{*}) / g_{0}(\underline{t}^{*}), & \text{for } i \in [0] , \\ d_{i}^{*} &= (\mu_{k}^{*}) u_{i}(\underline{t}^{*}) / g_{0}(\underline{t}^{*}), & \text{for } i \in [k] , k = 1, 2, ..., p, \end{aligned}$$
(4.10)

form a pair of primal and dual equilibrium solutions. Conversely, if \underline{t}^* and \underline{d}^* are a pair of primal and dual equilibrium solutions, then \underline{t}^* and $\underline{\mu}^*$ defined by

$$\mu_{k}^{*} = \lambda_{k}^{*} g_{0}(t^{*}), \text{ for } k = 1, 2, ..., p, \qquad (4.11)$$

where

$$\lambda_k^* = \sum_{i \in [k]} d_i^*,$$

is a pair of solutions of Lagrange problem.

Proof. Suppose \underline{t}^* and $\underline{\mu}^*$ are a pair of solutions of Lagrange problem, then \underline{d}^* defined by (4.10) satisfies that $\underline{d}^* \ge \underline{0}$ (positivity). By Definition 4.1, $\lambda_0^* = 1$ (normality). Let $\mu_0^* = 1$. Then

$$\sum_{k=0}^{p} \sum_{i \in [k]} a_{ij} d_{i}^{*} = \sum_{k=0}^{p} \mu_{k}^{*} \sum_{i \in [k]} a_{ij} u_{i}(\underline{t}^{*}) / g_{0}(\underline{t}^{*})$$
$$= [t_{j}^{*} / g_{0}(\underline{t}^{*})] [(1 / t_{j}^{*}) (\mu_{k}^{*}) \sum_{i \in [k]} a_{ij} u_{i}(\underline{t}^{*})]$$
$$= [t_{j}^{*} / g_{0}(\underline{t}^{*})] [\frac{\partial}{\partial t_{j}} L (\underline{t}^{*}, \underline{\mu}^{*})]$$

= 0 (orthogonality).

So, $\underline{d}^* \in F_{DR}$.

Note that $d_i^* = (\mu_k^*) u_i(\underline{i}^*) / g_0(\underline{i}^*)$ and $(\mu_k^*) [g_k(\underline{i}^*) - 1] = 0$, so, $\lambda_k^* = \sum_{i \in [k]} d_i^* = \mu_k^* / g_0(\underline{i}^*)$,

i.e.,

$$\mu_k^* = \lambda_k^* g_0(\underline{t}^*).$$

So,

$$d_{i}^{\star} = \lambda_{k}^{\star} u_{i}(\underline{t}^{\star}), \text{ for } i \in [k], \, k = 1, \, 2, \, ..., \, p.$$

Note that $d_i^* g_0(\underline{t}^*) = u_i(\underline{t}^*)$ for $i \in [0]$ and by Definition 4.1, \underline{t}^* and \underline{d}^* are a pair of primal and dual equilibrium solutions.

Suppose that <u>t</u>^{*} and <u>d</u>^{*} are a pair of primal and dual equilibrium solutions. Since <u>d</u>^{*} \geq 0, d_i^{*} = $\lambda_k^* u_i(\underline{t}^*)$ and $\lambda_k^* = \sum_{i \in [k]} d_i^* = \lambda_k^* g_k(\underline{t}^*)$, so, $g_k(\underline{t}^*) = 1$. Note

that

$$t_j^* \frac{\partial}{\partial t_j} L(\underline{t}^*, \underline{\mu}^*) = t_j^* (1/t_j^*) \sum_{k=0}^p \mu_k^* \sum_{i \in [k]} a_{ij} u_i(\underline{t}^*)$$

$$\begin{split} &= g_0(t^*) \sum_{k=0}^p \sum_{i \in [k]} a_{ij} (\lambda_k^*) u_i(t^*) \\ &= g_0(t^*) \sum_{k=0}^p \sum_{i \in [k]} a_{ij} d_i^* = 0. \end{split}$$

This leads to $\frac{\partial}{\partial t_j} L(t^*, \mu^*) = 0$ since $t_j^* > 0$. Hence, t^* and μ^* are a pair of

solutions of Lagrange problem.

From Theorem 4.1 and 4.2, we reach the following theorem.

Theorem 4.3. Suppose $\underline{t}^* \in F_{PR}$. Then \underline{t}^* is the primal equilibrium solution if and only if for all $\underline{t} > 0$, $\underline{t} \in F_t$, $g_0(\underline{t}^*) \le g_0(\underline{t})$.

Theorem 4.4. Suppose \underline{t}^* and \underline{d}^* are a pair of primal and dual equilibrium solutions. Then

(i) For all k = 0, 1, ..., p, either $d_i^* = 0$ for all $i \in [k]$, $\lambda_k^* = 0$, or $d_i^* > 0$ for all $i \in [k]$, $\lambda_k^* > 0$, especially $\lambda_0^* > 0$.

(ii) For all $\underline{\tau}$ such that $A^T \underline{\tau} = \underline{0}$, i.e., $\sum_{k=0}^{p} \sum_{i \in [k]} a_{ij} \tau_i = 0$, and $\tau_i = 0$ if $d_i = 0$ (for $(d_i^*)^{\tau_i}$ definable),

$$[g_{0}(\underline{t}^{*})]^{\omega_{0}} = \prod_{k=0}^{l} \prod_{i \in [k]} (c_{i}/d_{i})^{\tau_{i}} \prod_{k=l+1}^{p} \prod_{i \in [k]} (c_{i}/d_{i})^{-\tau_{i}} \prod_{k=1}^{l} (\lambda_{k}^{*})^{\omega_{k}} \prod_{k=l+1}^{p} (\lambda_{k}^{*})^{-\omega_{k}}$$

where $\omega_{0} = \sum_{i \in [k]} \tau_{i}$, $\omega_{k} = \sum_{i \in [k]^{1}} \tau_{i}$, for $k = 1, 2, ..., p$. (4.12)
(iii) $g_{0}(\underline{t}^{*}) = V(\underline{d}^{*})$.

Proof. (i) Note $d_i^* = \lambda_k^* u_i(\underline{t}^*)$, $u_i(\underline{t}^*) > 0$, for $i \in [k]$, k = 1, 2, ..., p. If $\exists i \in [k]$ such that $d_i^* = 0$, then it must be true that $\lambda_k^* = 0$, so for all $i \in [k]$, $d_i^* = 0$.

(ii) We know that $d_i^* g_0(\underline{t}^*) = u_i(\underline{t}^*)$ for $i \in [0]$, $d_i^* = \lambda_k^* u_i(\underline{t}^*)$ for $i \in [k]$, k = 1, 2, ..., p. So,

$$(d_i^* / c_i)^{\tau_i} [g_0(t^*)]^{\tau_i} = [u_i(t^*) / c_i]^{\tau_i}, \text{ for } i \in [0],$$

$$(d_i^*/c_i^{\dagger})^{\tau_i} = [\lambda_k^*]^{\tau_i} [u_i(\underline{t}^*)/c_i^{\dagger}]^{\tau_i}, \text{ for } \in [k], k = 1, 2, ..., l,$$

 $(d_i^*/c_i^{\dagger})^{-\tau_i} = [\lambda_k^*]^{-\tau_i} [u_i(\underline{t}^*)/c_i^{\dagger}]^{-\tau_i}, \text{ for } i \in [k], k = l + 1, l + 2, ..., p.$

Since $A^T \underline{\tau} = \underline{0}$, i.e., $\sum_{k=0}^{p} \sum_{i \in [k]} a_{ij} \tau_i = \eta_j = 0$, for j = 1, 2, ..., m,

$$[g_{0}(t^{*})]^{\omega_{0}}\prod_{k=0}^{l}\prod_{i\in[k]}(d_{i}'c_{i})^{\tau_{i}}\prod_{k=l+1}^{p}\prod_{i\in[k]}(d_{i}'c_{i})^{-\tau_{i}} = \prod_{k=1}^{l}(\lambda_{k}^{*})^{\omega_{k}}\prod_{k=l+1}^{p}(\lambda_{k}^{*})^{-\omega_{k}}[\prod_{j=1}^{m}(t_{j}^{*})^{\tau_{j}}]$$

where

$$\begin{split} & \omega_0 = \sum_{i \in [0]} \tau_i, \\ & \omega_k = \sum_{i \in [k]} \tau_i, \text{ for } k = 1, ..., p, \end{split}$$

Thus,

$$[g_{0}(\mathbf{i}^{*})]^{\omega_{0}} = \prod_{k=0}^{l} \prod_{i \in [k]} (c_{i}^{\prime}/d_{i}^{\prime})^{\tau_{i}} \prod_{k=l+1}^{p} \prod_{i \in [k]} (c_{i}^{\prime}/d_{i}^{\prime})^{-\tau_{i}} \prod_{k=1}^{l} (\lambda_{k}^{*})^{\omega_{k}} \prod_{k=l+1}^{p} (\lambda_{k}^{*})^{-\omega_{k}},$$

where

$$\begin{split} & \omega_0 = \sum_{i \in [0]} \tau_i ,\\ & \omega_k = \sum_{i \in [k]} \tau_i , \text{ for } k = 1, 2, ..., p. \end{split}$$
(iii) Let $\underline{\tau} = \underline{d}^*, \ \lambda_0^* = 1. \text{ Then we get } g_0(\underline{t}^*) = V(\underline{d}^*). \qquad | \\ & \textbf{Definition 4.2.} \text{ (nullity vector) } \underline{\Psi} = (v_1, v_2, ..., v_n)^T \text{ is a nullity vector if } \\ & \sum_{i \in [0]} v_i = 0, \end{split}$

$$A^{\mathsf{T}} \underline{\mathsf{v}} = \underline{\mathsf{0}}.$$

It is clear to see that if take $\underline{\tau}$ of (4.12) as \underline{v} ($\underline{\tau} = \underline{v}$), then we obtain the following theorem that gives a necessary condition on which \underline{d}^* may be a dual equilibrium solution, especially for I = p, \underline{d}^* is a dual optimum solution.

Theorem 4.5. Suppose that \underline{d}^* is a dual equilibrium solution. Then for all \underline{v} such that $\underline{v} = \underline{0}$ if $\underline{d}^* = \underline{0}$,

 $F(\underline{d}^{\star}, \underline{v}) = G(\underline{c}, \underline{v})$ (equilibrium identity),

where

$$F(\underline{d}^{*}, \underline{v}) = \prod_{k=0}^{l} \prod_{i \in [k]} (d_{i}^{*})^{V_{i}} \prod_{k=l+1}^{p} \prod_{i \in [k]} (d_{i}^{*})^{-V_{i}} \prod_{k=1}^{l} [\sum_{i \in [k]} d_{i}^{*}]^{-W_{k}} \prod_{k=l+1}^{p} [\sum_{i \in [k]} d_{i}^{*}]^{W_{k}}$$

$$(w_{0} = \sum_{i \in [0]} v_{i}, w_{k} = \sum_{i \in [k]} v_{i}, \text{ for } k = 1, 2, ..., p),$$

$$G(\underline{c}, \underline{v}) = \prod_{k=0}^{l} \prod_{i \in [k]} (c_{i})^{V_{i}} \prod_{k=l+1}^{p} \prod_{i \in [k]} (c_{i})^{-V_{i}}.$$

Theorem 4.6. Suppose that <u>d</u> is a dual consistent solution, i.e., <u>d</u> \in F_{DR} , <u>v</u> is a nullity vector such that $v_i = 0$ if $d_i = 0$, for i = 1, 2, ..., n. Then <u>d</u> + $\varepsilon \underline{v}$ is a dual consistent solution as $|\varepsilon|$ is sufficiently small ($\varepsilon \rightarrow 0$), and

 $D_v V(\underline{d}) = [\log G (\underline{c}, \underline{v}) - \log F (\underline{d}, \underline{v})] V(\underline{d}),$

where $D_v V(\underline{d})$ is the directional derivative of $V(\underline{d})$ at \underline{d} along the direction \underline{v} .

Especially, as $\underline{d} = \underline{d}^*$ (the dual equilibrium solution),

 $\mathsf{D}_{\mathsf{v}}\left[\mathsf{V}(\underline{d}^{\star}) \right] = 0.$

Proof. Note that $\underline{v} = \underline{0}$ if $\underline{d} = \underline{0}$ and by the definition we know $\underline{d} + \varepsilon \underline{v} \in F_{DR}$, as $\varepsilon \rightarrow 0$.

Since

$$\log V(\underline{d}) = \sum_{k=0}^{l} \sum_{i \in [k]} d_{i} (\log c_{i} - \log d_{i}) + \sum_{k=1}^{l} \lambda_{k} \log \lambda_{k}$$
$$- \sum_{k=l+1}^{p} \sum_{i \in [k]} d_{i} (\log c_{i} - \log d_{i}) - \sum_{k=l+1}^{p} \lambda_{k} \log \lambda_{k},$$
$$D_{v} [\log V(\underline{d})] = D_{v} [V(\underline{d})] / V(\underline{d})$$
$$= \frac{d}{d\epsilon} \{\log [V(\underline{d} + \epsilon \underline{v})]\}|_{\epsilon=0}$$

$$= \sum_{k=0}^{l} \sum_{i \in [k]} (\log c_{i} - \log d_{i} - 1) v_{i} + \sum_{k=1}^{l} \lambda_{k} (\log \lambda_{k} + 1)$$

$$- \sum_{k=i+1}^{p} \sum_{i \in [k]} d_i (\log c_i - \log d_i - 1) v_i - \sum_{k=i+1}^{p} \lambda_k (\log \lambda_k + 1)$$

$$= \log G (\underline{c}, \underline{v}) - \log F (\underline{d}, \underline{v}) - \sum_{i \in [0]} v_i .$$

Note that $\sum_{i \in [0]} v_i = 0$, so we have $D_v [V(\underline{d})] = [\log G(\underline{c}, \underline{v}) - \log F(\underline{d}, \underline{v})] V(\underline{d})$. By Theorem 4.5, as $\underline{d} = \underline{d}^*$ (the dual equilibrium solut

By Theorem 4.5, as
$$\underline{d} = \underline{d}^*$$
 (the dual equilibrium solution), we have $D_v[V(\underline{d}^*)] = 0.$

Lemma 4.1. Suppose $u_i > 0$, $d_i > 0$, for i = 1, 2, ..., n, and $\sum_{i=1}^{n} d_i = 1$. Then $(\sum_{i=1}^{n} u_i)^{-1} \le \prod_{i=1}^{n} (d_i / u_i)^{d_i} \le \sum_{i=1}^{n} d_i^2 / u_i$ and "=" holds if and only if $u_i = d_i \sum_{i=1}^{n} u_i$. *Proof.* From Corollary 1.1 we know that for $u_i > 0$, $d_i > 0$, i = 1, 2, ..., n, $\prod_{i=1}^{n} (u_i / d_i)^{d_i} \lambda^{\lambda} \le (\sum_{i=1}^{n} u_i)^{\lambda}$,

where $\lambda = \sum_{i=1}^{n} d_i$ and "=" holds if and only if $u_i \sum_{i=1}^{n} d_i = d_i \sum_{i=1}^{n} u_i$. That is, $\left(\sum_{i=1}^{n} u_i\right)^{-1} \leq \prod_{i=1}^{n} \left(d_i / u_i\right)^{d_i}$ ($\lambda = 1$).

Now let $u_i = d_i^2 / u_i$. Then

$$\left[\sum_{i=1}^{n} \left(d_{i}^{2} / u_{i}^{2} \right) \right]^{-1} \leq \prod_{i=1}^{n} \left(\frac{d_{i}^{2}}{d_{i}^{2} / u_{i}^{2}} \right)^{d_{i}},$$

i.e.,

$$\prod_{i=1}^{n} (d_{i} / u_{i})^{d_{i}} \leq \sum_{i=1}^{n} d_{i}^{2} / u_{i}.$$

Therefore,

$$\left(\sum_{i=1}^{n} u_{i}^{-1}\right)^{-1} \leq \prod_{i=1}^{n} \left(d_{i}^{-} u_{i}^{-}\right)^{-1} \leq \sum_{i=1}^{n} \left(d_{i}^{-2} u_{i}^{-}\right),$$

I

so, "=" holds if and only if $u_i = d_i \sum_{i=1}^n u_i$ $(\sum_{i=1}^n d_i = 1)$. For $\underline{\omega} > \underline{0}$ (weight vector) such that $\sum_{i=1}^n \omega_i = 1$ and posynomial $g_k(\underline{i}) = \sum_{i \in [k]} u_i(\underline{i}) = \sum_{i \in [k]} c_i \prod_{i=1}^m t_j^{a_{ij}}$, for $c_i \ge 0$, $a_{ij} \in \mathbb{R}$,

define

$$\begin{split} g_{k}'(\underline{t},\,\underline{\omega}) &= \prod_{i\in[k]} (\,\omega_{i} \,/\, u_{i}(\underline{t})\,)^{\,\omega_{i}}\,, \\ g_{k}''(\underline{t},\,\underline{\omega}) &= \sum_{i\in[k]} \omega_{i}^{\,2} \,/\, u_{i}(\underline{t}) \quad (\text{harmonized mean}). \end{split}$$

Then by Lemma 4.1, for all $\underline{t} > \underline{0}$,

 $1/g_k(t) \le g_k'(t, \underline{\omega}) \le g_k''(t, \underline{\omega}).$ For k = l+1, l+2, ..., p in (4.1) and $\underline{\omega} > \underline{0}$ such that $\sum_{i=1}^n \omega_i = 1$, we can form

the harmonized program (4.13) corresponding to (4.1), which is a posynomial geometric programming program. That is,

minimize
$$g_0(t)$$
,
subject to $t \in F_{PW}$, (4.13)

where

$$F_{PW} = \{ t \mid g_k(t) \le 1, \text{ for } k = 1, 2, ..., l, \\g_k''(t, \underline{\omega}) \le 1, \text{ for } k = l+1, l+2, ..., p, \\t_j > 0, \text{ for } j = 1, 2, ..., m \},$$

$$g_k(t) = \sum_{i \in [k]} u_i(t) = \sum_{i \in [k]} c_i \prod_{j=1}^m t_j^{a_{ij}}$$
, for $i \in k$], $k = 0, 1, ..., l, c_i \ge 0$, $a_{ij} \in R$,

$$g_k''(\underline{t}, \underline{\omega}) = \sum_{i \in [k]} \omega_i^2 / u_i(\underline{t}) = \sum_{i \in [k]} (\omega_i^2 / c_i) \prod_{j=1}^m t_j^{a_{ij}}$$
 (harmonized mean).

The associated dual program of (4.13) is:

maximize V'(d),

subject to
$$\underline{d} \in F_{DW}$$
,

where

$$\begin{split} \mathsf{F}_{\mathsf{DR}} &= \{ \underline{d} \mid \lambda_0 = \sum_{i \in [0]} d_i = 1, \\ &\sum_{k=1}^p \sum_{i \in [k]} a_{ij} d_i = 0, \text{ for } j = 1, 2, ..., m, \\ &d_i \geq 0, \ \lambda_k = \sum_{i \in [k]} d_i \geq 0, \text{ for } i = 1, 2, ..., n, \ k = 0, 1, ..., p \}, \end{split}$$

$$V'(\underline{d}) = \prod_{k=0}^{l} \prod_{i \in [k]} (c_i / d_i)^{d_i} \prod_{k=l+1}^{p} \prod_{i \in [k]} (\omega_i^2 / (c_i d_i))^{d_i} \prod_{k=1}^{p} (\lambda_k)^{\lambda_k}.$$

Theorem 4.8. Suppose $\underline{t} \in F_{PW}$ for $\underline{\omega}(\underline{t}) > \underline{0}$ such that $\sum_{i \in [k]} \omega_i(\underline{t}) = 1$, k = l+1, l+2, ..., p, then $\underline{t} \in F_{PR}$. Conversely, if $\underline{t} \in F_{PR}$, then $\underline{t} \in F_{PW}$ for $\underline{\omega}(\underline{t}) > \underline{0}$ such that $\omega_i(\underline{t}) = u_i(\underline{t}) / g_k(\underline{t})$, $i \in [k], k = l+1, l+2, ..., p$.

Proof. Note that

 $1/g_{k}(t) \leq g_{k}''(\underline{t}, \underline{\omega}(\underline{t})) \leq 1$, for k = l+1, l+2, ..., p,

i.e., for all $\underline{t} \in F_{PW}$, it must be true that $\underline{t} \in F_{PR}$, and the converse is not necessarily true. So,

But for $\underline{\omega}(\underline{t}) > \underline{0}$ such that

 $\omega_i(\underline{t}) = u_i(\underline{t}) / g_k(\underline{t})$, for $i \in [k]$, k = l+1, l+2, ..., p,

$$\sum_{i\in[k]}\omega_i(\underline{t})=1,$$

by Lemma 4.1, we know that

$$1/g_{k}(t) = g_{k}''(t, \omega(t))$$

if and only if

$$u_i(\underline{t}) = \omega_i(\underline{t}) \sum_{i \in [k]} u_i(\underline{t}) = \omega_i g_k(t)$$
.

(4.14)

That is, $F_{PW} = F_{PR}$, and $t \in F_{PR}$ implies $t \in F_{PW}$.

Note that $F_{DW} = F_{DR}$. So, we reach the following two theorems.

Theorem 4.9. $\underline{d} \in F_{DR}$ if and only if $\underline{d} \in F_{DW}$. And if $\underline{d} \in F_{DR}$ for $\underline{\omega} > \underline{0}$ such that $\sum_{i \in [k]} \omega_i = 1$, then $\underline{d} \in F_{DW}$.

Lemma 4.2. If (1.1) is consistent and $M_P > 0$, then (1.2) is consistent and $M_D = M_P \in (0, \infty)$.

Theorem 4.10. If (4.1) is consistent, i.e., $F_{PR} \neq \phi$ and $M_{PR} = \inf g_0(\underline{1}) > 0$ for $\underline{1} \in F_{PR}$, then (4.2) must be consistent, i.e., $F_{DR} \neq \phi$.

Proof. Suppose that <u>i</u> ∈ F_{PR}. Then by Theorem 4.8, for <u>ω(t)</u> > <u>0</u> such that $ω_i(t) = u_i(t) / g_k(t)$, for i ∈ [k], k = l+1, l+2, ..., p, t ∈ F_{PW}, i.e., F_{PW} ≠ φ. Since $F_{PW} ⊂ F_{PR}$, so $0 < M_{PR} ≤ M_{PW}$. Note that (4.13) is a posynomial program, so by Lemma 4.2 it must be consistent. Thus, by Theorem 4.9, (4.2) is consistent.

Theorem 4.11. Suppose $\underline{t}^* \in F_{PR}$, $\underline{d}^* \in F_{DR}$ and define $\underline{\omega}^*$ as

 $\omega_i(\underline{t}^*) = u_i(\underline{t}^*) / g_k(\underline{t}^*)$, for $i \in [k], k = i+1, i+2, ..., p$.

Then \underline{t}^* , \underline{d}^* are equilibrium solutions of (4.1) and (4.2), respectively, and $g_0(\underline{t}^*) = V(\underline{d}^*, \underline{\omega}^*)$ if and only if \underline{t}^* and \underline{d}^* are a pair of optimum solutions of (4.13) and (4.14).

Proof. By Theorem 4.8 and 4.9, that $\underline{i}^* \in F_{PR}$, $\underline{d}^* \in F_{DR}$ and $\omega_i(\underline{i}^*) = u_i(\underline{i}^*) / g_k(\underline{i}^*)$ implies $\underline{i}^* \in F_{PW}$, $\underline{d}^* \in F_{DW}$. Then by Theorem 1.4, \underline{i}^* , \underline{d}^* are the optimum solutions of (4.13) and (4.14), respectively, if and only if

$$\begin{split} &d_{i}^{\star} = u_{i}(\underline{t}^{\star}) / g_{0}(\underline{t}^{\star}), \text{ for } i \in [0] , \\ &d_{i}^{\star} = (\lambda_{k}^{\star}) u_{i}(\underline{t}^{\star}), \text{ for } i \in [k], k = 1, 2, ..., l, \\ &d_{i}^{\star} = (\lambda_{k}^{\star}) (\omega_{i}^{\star})^{2} / u_{i}(\underline{t}^{\star}), \text{ for } i \in [k], k = l+1, +2, ..., p. \end{split}$$

Since $\omega_i^* = u_i(t^*) / g_k(t^*)$, so, for $i \in [k], k = l+1, l+2, ..., p$,

$$d_{i}^{*} = (\lambda_{k}^{*}) (\omega_{i}^{*})^{2} / u_{i}(\underline{t}^{*}) = (\lambda_{k}^{*}) [u_{i}(\underline{t}^{*})]^{2} / [g_{k}(\underline{t}^{*})]^{2} u_{i}(\underline{t}^{*}),$$

= $(\lambda_{k}^{*}) u_{i}(\underline{t}^{*}) / [g_{k}(\underline{t}^{*})]^{2}.$ (4.15)

For $\lambda_k^* = 0$, $d_i^* = (\lambda_k^*) u_i(\underline{i}^*) / [g_k(\underline{i}^*)]^2 = 0$, and also $d_i^* = (\lambda_k^*) u_i(\underline{i}^*) = 0$, for $i \in [k]$, k = l+1, l+2, ..., p. For $\lambda_k^* > 0$,

$$\sum_{i \in [k]} d_i^* = (\lambda_k^*) \sum_{i \in [k]} u_i(\underline{t}^*) / [g_k(\underline{t}^*)]^2 \Rightarrow g_k(\underline{t}^*) = 1,$$

and also,

$$\sum_{i \in [k]} d_i^* = (\lambda_k^*) \sum_{i \in [k]} u_i(\underline{t}^*) \implies g_k(\underline{t}^*) = 1,$$

so (4.15) is equivalent to $d_i^* = (\lambda_k^*) u_i(\underline{t}^*)$, for $i \in [k]$, k = i+1, i+2, ..., p. By the definition of equilibrium solution, \underline{t}^* , \underline{d}^* are equilibrium solutions of (4.1) and (4.2), respectively. That means \underline{t}^* , \underline{d}^* are the optimum solutions of (4.13) and (4.14), respectively, if and only if they are a pair of primal and dual equilibrium solutions for (4.1) and (4.2). So, by Theorem 4.4 (iii), we get $g_0(\underline{t}^*) = V(\underline{d}^*)$. Thus, $g_0(\underline{t}^*) = V(\underline{d}^*, \underline{\omega}^*)$.

Suppose F_{DR} is nonempty, i.e. (4.1) and (4.2) are canonical (i.e., (4.2) has $\underline{d} > \underline{0}$, $\underline{d} \in F_{DR}$; otherwise, degenerate). Then for all $\underline{\omega}(\underline{t})$ such that $\sum_{i \in [k]} \omega_i(\underline{t}) = 1$ for k = l+1, l+2, ..., p, $F_{DR} = F_{DW}$, F_{DW} is nonempty, i.e. (4.13) and (4.14) are canonical. Then by Theorem 1.7, (4.13) must have optimum solution.

Based on Theorem 4.11 and 4.8, we have the following algorithm. Suppose $\underline{t}^0 \in F_{PR}$. Let n = 1. Step 1. Let $\omega_i^n = u_i(\underline{t}^{n-1}) / g_k(\underline{t}^{n-1})$, for k = l+1, l+2, ..., p, and form (4.13ⁿ) (i.e., (4.13) for $\underline{\omega} = \underline{\omega}^n$).

Step 2. Suppose
$$t^n$$
 is the optimum solution for (4.13ⁿ). Then
 $g_0(t^n) \le g_0(t^{n-1})$.

(by Theorem 4.8, t^n , $t^{n-1} \in F$, where F is the feasible region for (4.13ⁿ)).

Step 3. If
$$g_0(t^n) = g_0(t^{n-1})$$
, then stop. Otherwise, go to Step 4.

Step 4. Increase n by 1 and turn to Step 1.

By Theorem 4.8, we know $\underline{I}^{n-1} \in F_{PR}$. Conversely,

 $F_{PWn} \subset F_{PR} \implies \underline{t}^n \in F_{PR}, n = 0, 1, ... \quad (F_{PWn} \text{ denotes } F_{PW} \text{ for } \underline{\omega} = \underline{\omega}^n).$

So, { t^n } must be a consistent point sequence for (4.1).

Theorem 4.12.

(i) Suppose (4.13ⁿ) is superconsistent. If there exists n such that $g_0(t^n) = g_0(t^{n-1})$,

then t^{n-1} is the primal equilibrium solution of (4.1).

(ii) Suppose (4.13) is superconsistent. If for n = 1, 2, ..., $g_0(\underline{t}^n) < g_0(\underline{t}^{n-1}),$

 $t^n \rightarrow t^*$ as $n \rightarrow \infty$.

and define that

 $\omega_{i}^{\star} \, = \, u_{i}(\underline{i}^{\star}) \; / \; g_{k}(\underline{i}^{\star}), \ \ \, \text{for} \; i \in [k], \; k = l \! + \! 1, \; l \! + \! 2, \; ..., \; p,$

then t^* is the primal equilibrium solution of (4.1).

Proof. (i) Since t^n is the optimum solution of (4.13^n) and $g_0(t^n) = g_0(t^{n-1})$, t^{n-1} is the optimum solution of (4.13^n) . Note (4.13^n) is superconsistent, so, by Theorem 1.6, (4.14^n) has an optimum solution. By Theorem 4.11, we know that t^{n-1} and any of the optimum solution of (4.14^n) forms a pair of equilibrium solutions of (4.1) and (4.2).

(ii) Suppose $\{\underline{t}^{n'}\}$ is a subsequence of $\{\underline{t}^{n}\}$ and $\underline{t}^{n'} \rightarrow \underline{t}^{*}$ as $n' \rightarrow \infty$.

Use t' to express a superconsistent solution of (4.13^{*}) ((4.13) for $\underline{\omega} = \underline{\omega}^*$),

i.e.,

 $g_k(\underline{t}) < 1$, for k = 1, 2, ..., l,

$$g_k''(\underline{i}', \underline{\omega}^*) < 1$$
, for $k = l+1, l+2, ..., p$,

where

$$g_{k}''(\underline{t}', \underline{\omega}^{\star}) = \sum_{i \in [k]} [(\omega_{i}^{\star})^{2} / c_{i}] \prod_{j=1}^{m} (t_{j}')^{a_{ij}}.$$

Since $g_k(\underline{t})$, $u_i(\underline{t})$ are continuous and $\underline{t}^{n'} \to \underline{t}^*$ as $n' \to \infty$, so, $\underline{\omega}^{n'} \to \underline{\omega}^*$ as $n' \to \infty$, $\exists N$ such that for $n' \ge N$, $g_k''(\underline{t}', \underline{\omega}^{n'}) < 1$, for k = l+1, l+2, ..., p. That means \underline{t}' is also a superconsistent solution for $(4.13^{n'})$ (for n' > N). And by Theorem 1.6, $(4.14^{n'})$ (n' > N) also has an optimum solution $\underline{d}^{n'}$.

For $\varepsilon > 0$ (sufficient small),

$$(1 + \varepsilon) g_{k}(\underline{1}) < 1$$
, for k = 1, 2, ..., i,

 $(1 + \varepsilon) g_k'(\underline{t}', \underline{\omega}^*) < 1$, for k = l+1, l+2, ..., p

That is, for the problem that

minimize
$$(1 + \varepsilon) g_0(\underline{t})$$
,
subject to $\underline{t} \in F_{PeWn'}$, (4.16)

where

$$\begin{split} \mathsf{F}_{\mathsf{PeWn'}} &= \{ \underbrace{t} \mid (1 + \varepsilon) \; g_k(\underbrace{t}) < 1, \quad \text{for } k = 1, 2, ..., l, \\ & (1 + \varepsilon) \; g_k''(\underbrace{t}, \, \underline{\omega}^\star) < 1, \quad \text{for } k = l + 1, \, l + 2, \, ..., p \; \}. \end{split}$$

 $\exists t \in F_{PeWn'}$, that is, (4.16) has a superconsistent solution t',

 $M_{PeWn'} \leq (1 + \varepsilon) g_0(t').$

where

 $M_{PeWn'} = \inf (1 + \varepsilon) g_0(\underline{t}), \text{ for all } \underline{t} \in F_{PeWn'}.$

Note the dual problem of (4.16) has the same constraint set as that which (4.14^{n'}) has, so it has consistent solution $\underline{d}^{n'}$, and its corresponding objective function value is

$$(1 + \varepsilon)^{D} V'(\underline{d}^{n'}, \underline{\omega}^{n'})$$
, where $D = \sum_{k=0}^{p} \sum_{i \in [k]} d_{i}^{n'}$.

By Theorem 1.4, $V'(\underline{d}^{n'}, \underline{\omega}^{n'}) = g_0(\underline{t}^{n'})$ and we know that

 $(1 + \varepsilon)^{\mathsf{D}} g_{\mathsf{O}}(\underline{t}^{\mathsf{n}'}) \leq (1 + \varepsilon) g_{\mathsf{O}}(\underline{t}').$

Since $g_0(\underline{t}^n) \to g_0(\underline{t}^*) > 0$ (bounded), so { \underline{d}^n } must be bounded. Let $n' \to \infty$, we obtain

 $V'(\underline{d}^*, \underline{\omega}^*) = g_0(\underline{t}^*)$.

Note that $\underline{t}^{n'} \in F_{PWn'}$, $\underline{d}^{n'} \in F_{DWn'}$. By taking limit we have

 $\underline{t}^{\star} \in \mathsf{F}_{\mathsf{PW}^{\star}}, \ \underline{d}^{\star} \in \mathsf{F}_{\mathsf{DW}^{\star}}.$

(where F_{PW^*} , F_{DW^*} denote F_{PW} , F_{DW} for $\underline{\omega} = \underline{\omega}^*$, respectively). Therefore, \underline{t}^* and \underline{d}^* are the optimum solutions for (4.13*) and (4.14*), respectively. By Theorem 4.11, \underline{t}^* and any limit point of { $\underline{d}^{n'}$ } are the equilibrium solutions for (4.1) and (4.2).

The following example is an illustration for the properties presented above.

Consider the reversed geometric programming problem that

minimize
$$g_0(\underline{x}) = 3x_1 + 2x_2$$
,
subject to $\underline{x} \in F_{PR1}$, (4.17)

where

$$\begin{split} F_{PR1} &= \{ \underline{x} \mid g_{1}(\underline{x}) = x_{1}^{2} + x_{2}^{2} \ge 1, \\ &\quad x_{1}, x_{2} > 0 \}. \\ \text{Let } x_{1} &= e^{U_{1}}, x_{2}^{2} = e^{U_{2}}. \text{ Then (4.17) can be transformed to} \\ &\quad \text{minimize } f_{0}(\underline{u}) = 3e^{U_{1}} + 2e^{U_{2}}, \\ &\quad \text{subject to } \underline{u} \in F_{PR2}, \end{split}$$

$$(4.18)$$

where

$$F_{PB2} = \{ \underline{u} \mid f_1(\underline{u}) = e^{2u_1} + e^{2u_2} \ge 1 \}.$$

Note that

$$\underline{u}^{1} = (0, -1)^{T}, \ \underline{u}^{2} = (-1, 0)^{T} \in F_{PR2}.$$

But

$$\underline{u}^{3} = (1/2) \, \underline{u}^{1} + (1/2) \, \underline{u}^{2} = (-1/2, \, -1/2)^{\mathsf{T}} \notin \mathsf{F}_{\mathsf{PR2}} \, .$$

So, (4.18) is not a convex program.

It is clear that

 $\inf g_0(\underline{x}) = 2, \quad \text{for } \underline{x} \in F_{PR1}.$

The associated dual program of (4.17) is that

maximize
$$V(\underline{d}) = (3/d_1)^{d_1} (2/d_2)^{d_2} (1/d_3)^{-d_3} (1/d_4)^{-d_4} (d_3+d_4)^{-(d_3+d_4)}$$
,
subject to $\underline{d} \in F_{DR1}$, (4.19)

where

$$F_{DR1} = \{ \underline{d} \mid d_1 + d_2 = 1, \\ d_1 - 2d_3 = 0, \\ d_2 - 2d_4 = 0, \\ d_i \ge 0, \text{ for } i = 1, 2, 3, 4 \}$$

Clearly, $\underline{d}^1 = (1, 0, 1/2, 0)^T \in F_{DR1}$ and $V(\underline{d}^1) = 3$, that is,

$$\inf g_0(\underline{x}) = 2 \text{ for } \underline{x} \in F_{PR1} < V(\underline{d}^1) = 3 \le \sup V(\underline{d}) \text{ for } \underline{d} \in F_{DR1}.$$

Consider that

$$\underline{x}^* = (3/\sqrt{13}, 2/\sqrt{13})^{T},$$

 $\underline{\mu}^* = \sqrt{13}/2.$

Since

$$g_{1}(\underline{x}^{*}) - 1 = (9/13 + 4/13) - 1 = 0,$$

$$\nabla_{\underline{x}} L(\underline{x}^{*}, \underline{\mu}^{*}) = (3, 2)^{T} - (\sqrt{13}/2) (2(3)/\sqrt{13}, 2(2)/\sqrt{13})^{T} = 0.$$

 $(\underline{x}^*, \underline{\mu}^*)$ forms the solution for Lagrange problem of (4.17). And \underline{x}^* is not the local optimum of (4.17) since $g_0(\underline{x}^*) = \sqrt{13} > 2$.

$$A_{11} = \sum_{i=3}^{4} a_{i1} u_i(\underline{x}^*) = (-2) (3 / \sqrt{13})^2 = -18 / 13,$$

$$A_{12} = \sum_{i=3}^{4} a_{i2} u_i(\underline{x}^*) = (-2) (2 / \sqrt{13})^2 = -8 / 13.$$

Then by Theorem 4.1, for

$$\begin{split} \underline{x} \in \mathbf{F}_{t} &= \{ \underline{x} \mid \sum_{j=1}^{2} \mathbf{A}_{1j} (\log x_{j} - \log x_{j}^{*}) \leq 0 \} \\ &= \{ \underline{x} \mid (-18/13) (\log x_{1} - \log (3 / \sqrt{13}) \\ &+ (-8/13) (\log x_{2} - \log (2 / \sqrt{13})) \leq 0 \} \\ &= \{ \underline{x} \mid x_{1}^{9} x_{2}^{4} \geq 2^{4} 3^{9} / 13^{13/2} \}, \end{split}$$

we have

$$g_0(\underline{x}^*) = \sqrt{13} \le g_0(\underline{x}) = 3x_1 + 2x_2$$
.

That is, \underline{x}^* is the solution for the Lagrange problem and is the tangentially optimal solution.

By Theorem 4.2, we know

$$d_{1} = u_{1}(\underline{x}^{*}) / g_{0}(\underline{x}^{*}) = \frac{9 / \sqrt{13}}{\sqrt{13}} = 9 / 13,$$

$$\frac{4 / \sqrt{13}}{\sqrt{13}} = \frac{9 / \sqrt{13$$

$$d_2 = u_2(\underline{x}^*) / g_0(\underline{x}^*) = \frac{47\sqrt{13}}{\sqrt{13}} = 4/13$$
,

$$\begin{aligned} d_3 &= \mu^* \, u_3(\underline{x}^*) \, / \, g_0(\underline{x}^*) \, = (\sqrt{13} \, / \, 2) \, (9 \, / 13) \, / \sqrt{13} \, = 9 \, / \, 26, \\ d_4 &= \mu^* \, u_4(\underline{x}^*) \, / \, g_0(\underline{x}^*) \, = (\sqrt{13} \, / \, 2) \, (4 \, / 13) \, / \sqrt{13} \, = 2 \, / \, 13, \end{aligned}$$

That is, $\underline{d}^* = (9/13, 4/13, 9/26, 2/13)^T$ is the dual equilibrium solution.

By Theorem 4.5, for $\underline{v} = (v_1, v_2, ..., v_n)^T$ such that $\sum_{i \in [0]} v_i = 0$ and $A^T \underline{v} = \underline{0}$, i.e.

$$v_1 + v_2 = 0,$$

 $v_1 - 2v_3 = 0,$
 $v_2 - 2v_4 = 0.$

 $\underline{v} = (2, -2, 1, -1)^{\mathsf{T}} \ t, \ t \in \mathsf{R}.$

For equilibrium identity, we have

$$F (\underline{d}^{*}, \underline{v}) = (d_{1}^{*})^{2t} (d_{2}^{*})^{-2t} (d_{3}^{*})^{-t} (d_{4}^{*})^{t}$$

$$= (9/13)^{2t} (4/13)^{-2t} (9/26)^{-t} (2/13)^{t} = (3/2)^{2t}, \quad t \in \mathbb{R},$$

$$G (\underline{c}, \underline{v}) = (c_{1})^{2t} (c_{2})^{-2t} (c_{3})^{-t} (c_{4})^{t}$$

$$= (3)^{2t} (2)^{-2t} (1)^{-t} (1)^{t} = (3/2)^{2t}, \quad t \in \mathbb{R}.$$

Suppose that $\underline{\omega}^* = u_i(\underline{x}^*) / g_k(\underline{x}^*)$, so,

$$\omega_1^* = 9 / 13,$$

 $\omega_2^* = 4 / 13.$

Then problem (4.13) corresponding to (4.1) is that minimize $g_0(\underline{x}) = 3x_1 + 2x_2$, subject to $g_1(\underline{x}) = (81/169) x_1^{-2} + (16/169) x_2^{-2} \le 1$, $x_1, x_2 > 0$, (4.13')

problem (4.14) corresponding to (4.2) is that

maximize $V'(\underline{d}, \underline{\omega}^*) = (3/d_1)^{d_1} (2/d_2)^{d_2} (81/169d_3)^{-d_3}$

$$(16/169d_4)^{-d_4} (\lambda_1)^{\lambda_1},$$

subject to
$$\lambda_0 = 1$$
,
 $d_1 - 2d_3 = 0$,
 $d_2 - 2d_4 = 0$,
 $d_i \ge 0$, for i =1, 2, 3, 4. (4.14')

By Theorem 4.1, we know that \underline{x}^* , \underline{d}^* are the optimum solutions for (4.13) and (4.14), respectively.

Let

$$\underline{X}^{0} = (3, 2)^{\mathsf{T}} \in \mathsf{F}_{\mathsf{PR}}$$

Then

 $\underline{\omega}^{1} = (9 / 13, 4 / 13)^{T}.$

The optimum solution of (4.13') for $\underline{\omega} = \underline{\omega}^1$ is that

$$\underline{\mathbf{X}}^{*} = (3 / \sqrt{13}, 2 / \sqrt{13})^{\mathrm{T}},$$

and we have

$$\underline{\omega}^2 = \underline{\omega}^1 = (9/13, 4/13)^{\mathrm{T}}.$$

So, (3.13') for $\underline{\omega} = \underline{\omega}^2$ is the same as (3.13') for $\underline{\omega} = \underline{\omega}^1$. Therefore, by Theorem 4.12, \underline{x}^* is the equilibrium solution for the problem.

BIBLIOGRAPHY

.

BIBLIOGRAPHY

- [1] Avriel, M., Dembo, R.S., and Passy, U., *Solution of Generalized Geometric Programs,* International Journal of Numerical Methods in Engineering, Vol. 9, 1975.
- [2] Avriel, M., and Williams, A.C., *Complementary Geometric Programming*, SIAM on Apllied Mathematics, Vol. 19, 1970.
- [3] Duffin, R.J., *Linearizing Geometric Programs*, SIAM Review, Vol.12, 1970.
- [4] Duffin, R.J., and Peterson, E.L., *Geometric Programming with Signomials*, Journal of Optimization Theory and Application (JOTA), Vol. 11, No. 1, 1973.
- [5] Duffin, R.J., and Peterson, E.L., *Reversed Geometric Programming Treated by Harmonic Means*, Indiana University Mathematics Journal, Vol. 22, No. 3, 1980.
- [6] Duffin, R.J., Peterson, E.L., and Zener, C., *Geometric Programming: Theory and Applications*, John Wiley and Sons, New York, New York, 1967.
- [7] Ecker, J.G., *Geometric Programming*, SIAM Review, Vol. 22, No. 3, 1980.
- [8] Morris, A.J., Approximation and Complementary Geometric Programming, SIAM Journal on Applied Mathemetrics, Vol. 23, No. 4, 1972.
- [9] Peterson, E.L., *Geometric Programming*, SIAM Review, Vol. 18, No. 1, Jan. 1976
- [10] Sposito, V.A., *Linear and Nonlinear Programming,* Iowa State University Press, 1975.
- [11] Wu, F., Yuan, Y., *Geometric Programming*, Mathematics in Practice and Theory (Journal in Chinese), No. 1, 2, 3, 4, 1982.
- [12] Xuan, J., and Zheng, L., *Geometric Programming and Its Application* (in Chinese), Hunan University Press, 1988.

APPENDIX

DERIVATION OF DUAL PROBLEM

The associated dual problem can be derived from the primal problem by way of proper rearrangement of Lagrange multiplier [10].

Consider the problem that

minimize $g_0(t)$,

subject to $t \in F_p$, (A 1.1)

where

$$\begin{split} F_{p} &= \{1 \mid s_{k} [1 - g_{k}(\underline{i})] \geq 0, \ s_{k} = 1 \text{ or } -1, \ \text{ for } k = 1, 2, ..., p, \\ t_{j} > 0, \ j = 1, 2, ..., m \}, \\ g_{k}(\underline{i}) &= \sum_{i \in [k]} u_{ki}(\underline{i}) = \sum_{i \in [k]} c_{ki} \prod_{j=1}^{m} t_{j}^{a_{ij}}, \ \text{ for } k = 0, 1, ..., p. \end{split}$$

Suppose that

$$\begin{split} & \omega_{0i} = u_{0i}(\underline{t}) / g_{0}(\underline{t}), & \text{for } i \in [0], \\ & \omega_{ki} = u_{ki}(\underline{t}), & \text{for } i \in [k], \ k = 1, 2, ..., p. \end{split}$$

Then it is clear that

$$\begin{split} &\sum_{i\in[0]}\omega_{0i}(\underline{t})=1,\\ &\sum_{i\in[k]}\omega_{ki}(\underline{t})=g_{k}(\underline{t}), \quad \text{for } k=1,\,2,\,...,\,p. \end{split}$$

Let

$$x_j = \log t_j$$
, for j = 1, 2, ..., m,
 $x_0 = \log g_0(t)$.

Then for k = 0, note that $u_{0i}(\underline{t}) = \omega_{0i} g_0(\underline{t})$ and $u_{0i}(\underline{t}) = c_{0i} \prod_{j=1}^{m} t_j^{a_{ij}}$, we have

$$\log u_{0i}(\underline{t}) = \log \omega_{0i} + \log g_0(\underline{t})$$

$$= \log c_{0i} + \sum_{j=1}^{m} a_{ij} x_{j}$$
.

That is,

$$\begin{split} x_{0} + \log (\omega_{0i} / c_{0i}) - \sum_{j=1}^{m} a_{ij} x_{j} &= 0. \end{split} \tag{A 1.2} \\ \text{And for } k = 1, 2, ..., p, \text{ note that } u_{ki}(\underline{t}) &= \omega_{ki} \text{ and } u_{ki}(\underline{t}) = c_{ki} \prod_{j=1}^{m} t_{j}^{a_{ij}}, \text{ we have} \\ \log \omega_{ki} &= \log c_{ki} + \sum_{j=1}^{m} a_{ij} x_{j}. \end{split}$$

1

That is,

$$s_{k} [\log (\omega_{ki} / c_{ki}) - \sum_{j=1}^{m} a_{ij} x_{j}] = 0.$$
 (A 1.3)

Let

$$h_k = s_k (1 - \sum_{i \in [k]} \omega_{ki}(\underline{t})), \text{ for } k = 1, 2, ..., p.$$

Then (A 1.1) is equivalent to the following problem

minimize x_0 , subject to $\underline{\omega}, \chi \in F_{P1}$, (A 1.4)

where

$$\begin{split} \mathsf{F}_{\mathsf{P1}} &= \{ \ \underline{\omega}, \ \underline{x} \ | \ 1 - \sum_{i \in [0]} \omega_{0i}(\underline{t}) \ = 0, \\ & \mathsf{h}_k \geq 0 \ , \quad \text{for } k = 1, \ 2, \ ..., \ p, \\ & \mathsf{x}_0 + \log \ (\omega_{0i} \ / \ c_{0i}) \ - \ \sum_{j=1}^m a_{ij} \ \mathsf{x}_j = 0 \ , \quad \text{for } i \in [0], \\ & \mathsf{s}_k \ [\ \log \ (\omega_{ki} \ / \ c_{ki}) \ - \ \sum_{j=1}^m a_{ij} \ \mathsf{x}_j \] = 0 \ , \quad \text{for } i \in [k], \ k = 1, \ 2, \ ..., \ p \ \}. \end{split}$$

By defining Lagrange multiplier λ_k ', for k = 1, 2, ..., p, such that

$$\lambda_k' h_k = 0$$
,

that is called Kuhn-Tucker complementary slackness condition, we can construct a proper Lagrangian function for (A 1.4) as follows.

$$L(\underline{\omega}, \underline{x}; \underline{\lambda}, \underline{\lambda}') = x_0 - \lambda_0' (1 - \sum_{i \in [0]} \omega_{0i})$$

$$- \sum_{k=1}^{p} \lambda_{k}' h_{k}$$

$$- \sum_{i \in [0]} \lambda_{0i} [x_{0} + \log (\omega_{0i} / c_{0i}) - \sum_{j=1}^{m} a_{ij} x_{j}]$$

$$- \sum_{k=1}^{p} s_{k} \lambda_{ki} [\log (\omega_{ki} / c_{ki}) - \sum_{j=1}^{m} a_{ij} x_{j}]. \quad (A 1.5)$$

Let

$$\begin{split} &\frac{\partial L}{\partial x_0} = 1 - \sum_{i \in [0]} \lambda_{0i} = 0, \\ &\frac{\partial L}{\partial x_j} = \sum_{k=0}^{p} \sum_{i \in [0]} s_k \lambda_{ki} a_{ij} = 0, \quad \text{for } j = 1, 2, ..., m, \\ &\frac{\partial L}{\partial \omega_{0i}} = \lambda_0' - \lambda_{0i} / \omega_{0i} = 0, \quad \text{for } i \in [0], \\ &\frac{\partial L}{\partial \omega_{ki}} = s_k \left[\lambda_0' - \lambda_{0i} / \omega_{0i} \right] = 0, \quad \text{for } i \in [k], \ k = 1, 2, ..., p. \end{split}$$

Then for the optimal Lagrange multiplier $\underline{\lambda}$,

$$\begin{split} & \sum_{i \in [0]} \lambda_{0i} = 1, \\ & \lambda_{0i} = \lambda_0' \omega_{0i}, \\ & \lambda_{ki} = \lambda_k' \omega_{ki}. \end{split}$$

Note that

$$\begin{split} &\sum_{i\in[0]}\lambda_{0i}=\lambda_{0}'\left(\sum_{i\in[0]}\omega_{0i}\right)=\lambda_{0}'=1\quad\left(\sum_{i\in[0]}\omega_{0i}=1\right),\\ &g_{k}(\underline{t})=\sum_{i\in[k]}\omega_{ki}=1, \ \text{for} \ \lambda_{k}'>0\quad\left(\lambda_{k}'>0\ \Rightarrow h_{k}=0 \ \text{since} \ \lambda_{k}'h_{k}=0\right),\\ &\sum_{i\in[k]}\lambda_{ki}=\lambda_{k}'\left(\sum_{i\in[k]}\omega_{ki}\right)=\lambda_{k}'=1, \ \text{for} \ \lambda_{k}'>0, \end{split}$$

we have

$$\begin{split} & \omega_{0i} = \lambda_{0i} , \\ & \omega_{ki} = \lambda_{ki} / \lambda_{k}' = \lambda_{ki} / \sum_{i \in [k]} \lambda_{ki} , \\ & \sum_{k=0}^{p} \sum_{i \in [0]} s_{k} \lambda_{ki} a_{ij} = 0, \quad \text{for } j = 1, 2, ..., m. \end{split}$$

Let $s_0 = 1$ and note that

$$\begin{split} \sum_{i \in [0]} \omega_{0i} &= 1, \quad \omega_{0i} = \lambda_{0i}, \\ x_0 - \sum_{i \in [0]} \lambda_{0i} x_0 &= x_0 \ (1 - \sum_{i \in [0]} \lambda_{0i}) - 1 + 1 \\ &= x_0 \ (1 - \sum_{i \in [0]} \lambda_{0i}) - (1 - \sum_{i \in [0]} \lambda_{0i}) \\ &= (x_0 - 1) \ (1 - \sum_{i \in [0]} \lambda_{0i}). \end{split}$$

So,

$$\begin{split} L\left(\underline{\lambda},\underline{\lambda}';\underline{x},\underline{h}\right) &= L\left(\underline{\omega},\underline{x};\underline{\lambda},\underline{\lambda}'\right) \\ &= \sum_{k=0}^{p} \sum_{i \in [0]} s_{k} \lambda_{ki} \log\left[c_{ki}\left(\sum_{i \in [k]} \lambda_{ki} / \lambda_{ki}\right)\right] \\ &+ \prod_{j=1}^{m} x_{j}\left(\sum_{k=0}^{p} \sum_{i \in [0]} s_{k} \lambda_{ki} a_{ij}\right) \\ &- \sum_{k=1}^{p} h_{k} \lambda_{k}' + (x_{0} - 1)\left(1 - \sum_{i \in [0]} \lambda_{0i}\right). \end{split}$$

This Lagrangian function can be taken as another constrained maximization problem. That is,

$$\begin{array}{ll} \text{maximize} & f_0(\underline{\lambda}) = \sum_{k=0}^{p} \sum_{i \in [k]} s_k \lambda_{ki} \log \left[c_{ki} \left(\sum_{i \in [k]} \lambda_{ki} / \lambda_{ki} \right) \right], \\ \text{subject to} & \underline{\lambda} \in F_D, \end{array}$$

$$(A 1.6)$$

where

$$\begin{split} \mathsf{F}_{D} &= \{ \underbrace{\lambda} \mid \sum_{i \in [0]}^{\lambda} \lambda_{0i} = 1, \\ & \sum_{i \in [k]}^{\lambda} \lambda_{ki} \ge 0, \quad \text{for } k = 1, 2, ..., p, \\ & \sum_{k=0}^{p} \sum_{i \in [0]}^{N} s_{k} \lambda_{ki} \, a_{ij} = 0, \quad \text{for } j = 1, 2, ..., m \, \}. \end{split}$$

Let

$$d_i = \lambda_{ki}, \text{ for } i \in [k], k = 0, 1, ..., p, \sum_{i \in [k]} d_i = \lambda_k = \sum_{i \in [k]} \lambda_{ki},$$

and

$$\begin{split} \mathsf{V}(\underline{d}) &= \exp\left\{ f_{0}(\underline{\lambda}) \right\} \\ &= \prod_{k=0}^{p} \prod_{i \in [k]} \left[\left(c_{ki} / \lambda_{ki} \right)^{S_{k}\lambda_{ki}} \right] \prod_{k=1}^{p} \left[\left(\lambda_{k} \right)^{S_{k}\lambda_{k}} \right] \quad \left(\lambda_{0} = 1 \right) \\ &= \prod_{k=0}^{p} \prod_{i \in [k]} \left[\left(c_{ki} / d_{i} \right)^{S_{k}d_{i}} \right] \prod_{k=1}^{p} \left[\left(\lambda_{k} \right)^{S_{k}\lambda_{k}} \right]. \end{split}$$

Then for $s_k = 1$, (A 1.6) becomes

maximize
$$V(\underline{d}) = \prod_{i=1}^{n} (c_i / d_i) \stackrel{d_i}{\longrightarrow} \prod_{k=1}^{p} [(\lambda_k)^{\lambda_k}],$$

subject to $\underline{d} \in F_D^{-},$ (A 1.7)

where

$$\begin{split} F_D &= \{ \ \underline{d} \mid \lambda_0 = 1, \\ &\prod_{i=1}^n a_{ij} d_i = 0, \quad \text{for } j = 1, \, 2, \, ..., \, m, \\ &d_i \, , \, \lambda_k \geq 0, \quad \text{for } i = 1, \, 2, \, ..., \, n, \, \, k = 1, \, 2, \, ..., \, p \, \} \, . \end{split}$$

That is, the associated dual problem of (A 1.1) for $s_k = 1, k = 1, 2, ..., p$.

VITA

Yancheng Lin was born in Wendeng, Shandong, China on September 9, 1944. He was educated in Wuhan, Hubei, China and graduated from the Thirteenth Middle School of Wuhan in July 1964. From 1964 to 1969, he studied as a undergraduate majoring in industrial economics at the People's University of China (PUC), Beijing, China, and the study was very badly influenced by the cultural revolution that took place in China in the duration of his study. He reentered the university to study as a graduate majoring in systems engineering and management in September 1979 and received a Master degree in Economics in 1982. Between 1969 and 1979 he worked as an assistant engineer in electronics in the manufacture, test, design and research of the electronic instruments used in communication, medicine and optics successively at the three factories in Shayang, Wuhan, Hubei and Beijing, China. Before he was enrolled as a graduate at the University of Tennessee at Knoxville (UTK) in January 1988 he worked as a lecturer at the People's University of China for about five years. At present he is completing a thesis for a Master degree of Science in Mathematics.